# Algebraic and geometric methods in enumerative combinatorics

Federico Ardila\*

# 0 Introduction

Enumerative combinatorics is about counting. The typical question is to find the number of objects with a given set of properties.

However, enumerative combinatorics is not just about counting. In "real life", when we talk about *counting*, we imagine lining up a set of objects and counting them off:  $1, 2, 3, \ldots$ . However, families of combinatorial objects do not come to us in a natural linear order. To give a very simple example: we do not count the squares in an  $m \times n$  rectangular grid linearly. Instead, we use the rectangular structure to understand that the number of squares is  $m \cdot n$ . Similarly, to count a more complicated combinatorial set, we usually spend most of our efforts understanding the underlying structure of the individual objects, or of the set itself.

Many combinatorial objects of interest have a rich and interesting **algebraic** or **geometric** structure, which often becomes a very powerful tool towards their enumeration. In fact, there are many families of objects which we **only** know how to count using these tools. Our goal in this chapter is to highlight some key aspects of the rich interplay between algebra, discrete geometry, and combinatorics, with an eye towards enumeration.

**About this chapter.** Over the last fifty years, combinatorics has undergone a radical transformation. Not too long ago, combinatorics mostly consisted of ad hoc methods and clever solutions to problems that were fairly isolated from the rest of mathematics. It has since grown to be a central area of mathematics, largely thanks to the discovery of deep connections to other fields. Combinatorics has become an essential tool in many disciplines. Conversely, even though ingenious methods and clever new ideas still abound, there is now a powerful, extensive toolkit of algebraic, geometric, topological, and analytic techniques that can be applied to combinatorial problems.

It is impossible to give a meaningful summary of the many facets of algebraic and geometric combinatorics in a writeup of this length. I found it very difficult but necessary to omit several beautiful, important directions. In the spirit of a *Handbook of Enumerative Combinatorics*, my guiding principle was to focus on algebraic and geometric techniques that are useful towards the solution of enumerative problems. My main goal was to state clearly and concisely some of the most useful tools in algebraic and geometric enumeration, and to give many examples that quickly and concretely illustrate how to put these tools to use.

<sup>\*</sup>San Francisco State University, San Francisco, USA; U. de Los Andes, Bogotá, Colombia. federico@sfsu.edu Partially supported by the US National Science Foundation CAREER Award DMS-0956178 and the SFSU-Colombia Combinatorics Initiative.

# PART 1. ALGEBRAIC METHODS

The first part of this chapter focuses on algebraic methods in enumeration. In Section 1 we discuss the question: what is a good answer to an enumerative problem? Generating functions are the most powerful tool to unify the different kinds of answers that interest us: explicit formulas, recurrences, asymptotic formulas, and generating functions. In Section 2 we develop the algebraic theory of generating functions. Various natural operations on combinatorial families of objects correspond to simple algebraic operations on their generating functions, and this allows us to count many families of interest. In Section 3 we show how many problems in combinatorics can be rephrased in terms of linear algebra, and reduced to the problem of computing determinants. Finally, Section 4 is devoted to the theory of posets. Many combinatorial sets have a natural poset structure, and this general theory is very helpful in enumerating such sets.

# 1 What is a good answer?

The main goal of enumerative combinatorics is to count the elements of a finite set. Most frequently, we encounter a family of sets  $T_0, T_1, T_2, T_3, \ldots$  and we need to find the number  $t_n = |T_n|$  for  $n = 1, 2, \ldots$  What constitutes a good answer?

Some answers are obviously good. For example, the number of subsets of  $\{1, 2, ..., n\}$  is  $2^n$ , and it seems clear that this is the simplest possible answer to this question. Sometimes an answer "is so messy and long, and so full of factorials and sign alternations and whatnot, that we may feel that the disease was preferable to the cure" [Wil06]. Usually, the situation is somewhere in between, and it takes some experience to recognize a good answer.

A combinatorial problem often has several kinds of answers. Which answer is better depends on what one is trying to accomplish. Perhaps this is best illustrated with an example. Let us count the number  $a_n$  of **domino tilings** of a  $2 \times n$  rectangle into  $2 \times 1$  rectangles. There are several different ways of answering this question.



Figure 1.1: The five domino tilings of a  $2 \times 4$  rectangle.

**Explicit formula 1.** We first look for an explicit combinatorial formula for  $a_n$ . To do that, we play with a few examples, and quickly notice that these tilings are structurally very simple: they are just a sequence of  $2 \times 1$  vertical tiles, and  $2 \times 2$  blocks covered by two horizontal tiles. Therefore constructing a tiling is the same as writing n as an ordered sum of 1s and 2s. For example, the tilings of Figure 1.1 correspond, respectively, to 1 + 1 + 1 + 1, 1 + 1 + 2, 1 + 2 + 1, 2 + 1 + 1, 2 + 2. These sums are easy to count. If there are k summands equal to 2 there must be n - 2k summands equal to 1, and there are  $\binom{n-2k+k}{k} = \binom{n-k}{k}$  ways of ordering the summands. Therefore

$$a_n = \sum_{k=0}^{\lfloor n/2 \rfloor} \binom{n-k}{k} = \binom{n}{0} + \binom{n-1}{1} + \binom{n-2}{2} + \cdots$$
 (1)

This is a pretty good answer. It is certainly an explicit formula, and it may be used to compute  $a_n$  directly for small values of n. It does have two drawbacks. Aesthetically, it is certainly not as satisfactory as " $2^n$ ". In practice, it is also not as useful as it seems; after computing a few examples, we will soon notice that computing binomial coefficients is a non-trivial task. In fact there is a more efficient method of computing  $a_n$ .

**Recurrence.** Let  $n \ge 2$ . In a domino tiling, the leftmost column of a  $2 \times n$  can be covered by a vertical domino or by two horizontal dominoes. If the leftmost domino is vertical, the rest of the dominoes tile a  $2 \times (n-1)$  rectangle, so there are  $a_{n-1}$  such tilings. On the other hand, if the two leftmost dominoes are horizontal, the rest of the dominoes tile a  $2 \times (n-2)$  rectangle, so there are  $a_{n-2}$  such tilings. We obtain the **recurrence relation** 

$$a_0 = 1, \quad a_1 = 1, \quad a_n = a_{n-1} + a_{n-2} \text{ for } n \ge 2.$$
 (2)

which allows us to compute each term in the sequence in terms of the previous ones. We see that  $a_n = F_{n+1}$  is the (n+1)th **Fibonacci number**.

This recursive answer is not as nice as " $2^n$ " either; it is not even an explicit formula for  $a_n$ . If we want to use it to compute  $a_n$ , we need to compute all the first n terms of the sequence  $1, 1, 2, 3, 5, 8, 13, 21, 34, 55, 89, 144, \ldots$  However, we can compute those very quickly; we only need to perform n - 1 additions. This is an extremely efficient method for computing  $a_n$ .

**Explicit formula 2.** There is a well established method that turns linear recurrence relations with constant coefficients, such as (2), into explicit formulas. We will review it in Theorem 2.4.1. In this case, the method gives

$$a_n = \frac{1}{\sqrt{5}} \left( \left( \frac{1+\sqrt{5}}{2} \right)^{n+1} - \left( \frac{1-\sqrt{5}}{2} \right)^{n+1} \right).$$
(3)

This is clearly the simplest possible explicit formula for  $a_n$ ; in that sense it is a great formula.

A drawback is that this formula is really not very useful if we want to compute the exact value of, say,  $a_{1000}$ . It is not even clear why (3) produces an integer, and to get it to produce the correct integer would require arithmetic calculations with extremely high precision.

An advantage is that, unlike (1), (3) tells us very precisely how  $a_n$  grows with n.

Asymptotic formula. It follows immediately from (3) that

$$a_n \sim c \cdot \varphi^n,$$
 (4)

where  $c = \frac{1+\sqrt{5}}{2\sqrt{5}}$  and  $\varphi = \frac{1+\sqrt{5}}{2} \approx 1.6179...$  is the golden ratio. This notation means that  $\lim_{n\to\infty} a_n/(c\cdot\alpha^n) = 1$ . In fact, since  $|\frac{1-\sqrt{5}}{2}| < 1$ ,  $a_n$  is the closest integer to  $c\cdot\alpha^n$ .

**Generating function.** The last kind of answer we discuss is the generating function. This is perhaps the strangest kind of answer, but it is often the most powerful one.

Consider the infinite power series  $A(x) = a_0 + a_1x + a_2x^2 + \cdots$ . We call this the **generating** function of the sequence  $a_0, a_1, a_2, \dots^1$  We now compute this power series: From (2) we obtain

<sup>&</sup>lt;sup>1</sup>For the moment, let us not worry about where this series converges. The issue of convergence can be easily avoided (as combinatorialists often do, in a way which will be explained in Section 2.1) or resolved and exploited to our advantage; let us postpone that discussion for the moment.

that  $A(x) = 1 + x + \sum_{n \ge 2} (a_{n-1} + a_{n-2})x^n = 1 + x + x(A(x) - 1) + x^2A(x)$  which implies:

$$A(x) = a_0 + a_1 x + a_2 x^2 + \dots = \frac{1}{1 - x - x^2}.$$
(5)

With a bit of theory and some practice, we will be able to write the equation (5) immediately, with no further computations (Example 18 in Section 2.2). To show this is an **excellent** answer, let us use it to derive all our other answers, and more.

• Generating functions help us obtain explicit formulas. For instance, rewriting

$$A(x) = \frac{1}{1 - (x + x^2)} = \sum_{k \ge 0} (x + x^2)^k$$

we recover (1). If, instead, we use the method of partial fractions, we get

$$A(x) = \left(\frac{1/\sqrt{5}}{1 - \frac{1+\sqrt{5}}{2}x}\right) - \left(\frac{1/\sqrt{5}}{1 - \frac{1-\sqrt{5}}{2}x}\right)$$

which brings us to our second explicit formula (3).

• Generating functions help us obtain recursive formulas. In this example, we simply compare the coefficients of  $x^n$  on both sides of the equation  $A(x)(1-x-x^2) = 1$ , and we get the recurrence relation (2).

• Generating functions help us obtain asymptotic formulas. In this example, (5) leads to (3), which gives (4). In general, almost everything that we know about the rate of growth of combinatorial sequences comes from their generating functions, because analysis tells us that the asymptotic behavior of  $a_n$  is intimately tied to the singularities of the function A(x).

• Generating functions help us enumerate our combinatorial objects in more detail, and understand some of their statistical properties. For instance, say we want to compute the number  $a_{m,n}$ of domino tilings of a  $2 \times n$  rectangle which use exactly *m* vertical tiles. Once we really understand (5) in Section 2.2, we will get the answer immediately:

$$\frac{1}{1 - vx - x^2} = \sum_{m,n \ge 0} a_{m,n} v^m x^n.$$

Now suppose we wish to know what fraction of the tiles is vertical in a large random tiling. Among all the  $a_n$  domino tilings of the  $2 \times n$  rectangle, there are  $\sum_{m\geq 0} ma_{m,n}$  vertical dominoes. We compute

$$\sum_{n\geq 0} \left(\sum_{m\geq 0} ma_{m,n}\right) x^n = \left[\frac{\partial}{\partial v} \left(\frac{1}{1-vx-x^2}\right)\right]_{v=1} = \frac{x}{(1-x-x^2)^2}$$

Partial fractions then tell us that  $\sum_{m\geq 0} ma_{m,n} \sim \frac{n}{5} \left(\frac{1+\sqrt{5}}{2}\right)^{n+1} \sim \frac{1}{\sqrt{5}} na_n$ . Hence the fraction of vertical tiles in a random domino tiling of a  $2 \times n$  rectangle converges to  $1/\sqrt{5}$  as  $n \to \infty$ .

**So what is a good answer to an enumerative problem?** Not surprisingly, there is no definitive answer to this question. When we count a family of combinatorial objects, we look for explicit formulas, recursive formulas, asymptotic formulas, and generating functions. They are all useful. Generating functions are the most powerful framework we have to relate these different kinds of answers and, ideally, find them all.

# 2 Generating functions

In combinatorics, one of the most useful ways of "determining" a sequence of numbers  $a_0, a_1, a_2, \ldots$  is to compute its **ordinary generating function** 

$$A(x) = \sum_{n \ge 0} a_n x^n = a_0 + a_1 x + a_2 x^2 + a_3 x^3 + a_4 x^4 + \cdots$$

or its exponential generating function

$$A_{\exp}(x) = \sum_{n \ge 0} a_n \frac{x^n}{n!} = a_0 + a_1 x + a_2 \frac{x^2}{2} + a_3 \frac{x^3}{6} + a_4 \frac{x^4}{24} + \cdots$$

This simple idea is extremely powerful because some of the most common algebraic operations on ordinary and exponential generating functions correspond to some of the most common operations on combinatorial objects. This allows us to count many interesting families of objects; this is the content of Section 2.2 (for ordinary generating functions) and Section 2.3 (for exponential generating functions). In Section 2.4 we see how nice generating functions can be turned into explicit, recursive, and asymptotic formulas for the corresponding sequences.

Before we get to this interesting theory, we have to understand what we mean by power series. Section 2.1 provides a detailed discussion, which is probably best skipped the first time one encounters power series. In the meantime, let us summarize it in one paragraph:

There are two main attitudes towards power series in combinatorics: the analytic attitude and the algebraic attitude. To harness the full power of power series, one should really understand both. Chapter 2 of this Handbook of Enumerative Combinatorics is devoted to the analytic approach, which treats A(x) as an honest analytic function of x, and uses analytic properties of A(x) to derive combinatorial properties of  $a_n$ . In this chapter we follow the algebraic approach, which treats A(x)as a formal algebraic expression, and manipulates it using the usual laws of algebra, without having to worry about any convergence issues.

## 2.1 The ring of formal power series

Enumerative combinatorics is full of intricate algebraic computations with power series, where justifying convergence is cumbersome, and usually unnecessary. In fact, many natural power series in combinatorics only converge at 0, such as  $\sum_{n\geq 0} n! x^n$ ; so analytic methods are not available to study them. For these reasons we often prefer to carry out our computations algebraically in terms of **formal** power series. We will see that even in this approach, analytic considerations are often useful.

In this section we review the definition and basic properties of the **ring of formal power** series  $\mathbb{C}[[x]]$ . For a more in depth discussion, including the (mostly straightforward) proofs of the statements we make here, see [Niv69].

Formal power series. A formal power series is an expression of the form

$$A(x) = a_0 + a_1 x + a_2 x^2 + \cdots, \qquad a_0, a_1, a_2, \ldots \in \mathbb{C}.$$

Formally, this series is just a sequence of complex numbers  $a_0, a_1, a_2, \ldots$  We will see that it is convenient to denote it A(x), but we do not consider it to be a function of x.

Let  $\mathbb{C}[[x]]$  be the **ring of formal power series**, where the sum and the product of  $A(x) = \sum_{n>0} a_n x^n$  and  $B(x) = \sum_{n>0} b_n x^n$  are:

$$A(x) + B(x) = \sum_{n \ge 0} (a_n + b_n) x^n, \qquad A(x)B(x) = \sum_{n \ge 0} \left( \sum_{k=0}^n a_k b_{n-k} \right) x^n.$$

It is implicit in the definition, but worth mentioning, that  $\sum_{n\geq 0} a_n x^n = \sum_{n\geq 0} b_n x^n$  if and only if  $a_n = b_n$  for all  $n \geq 0$ .

We define the **degree** of  $A(x) = \sum_{n\geq 0} a_n x^n$  to be the **smallest** n such that  $a_n \neq 0$ . We also write

$$[x^n]A(x) := a_n, \qquad A(0) := [x^0]A(x) = a_0.$$

We also define formal power series inspired by series from analysis, such as

$$e^x := \sum_{n \ge 0} \frac{x^n}{n!}, \qquad -\log(1-x) := \sum_{n \ge 1} \frac{x^n}{n}, \qquad (1+x)^r := \sum_{n \ge 0} \binom{r}{n} x^n,$$

for any complex number r, where  $\binom{r}{n} := r(r-1)\cdots(r-n+1)/n!$ .

The ring  $\mathbb{C}[[x]]$  is commutative with  $0 = 0 + 0x + \cdots$  and  $1 = 1 + 0x + \cdots$ . It is an integral domain; that is, A(x)B(x) = 0 implies that A(x) = 0 or B(x) = 0. It is easy to describe the units:

$$\sum_{n \ge 0} a_n x^n \text{ is invertible } \iff a_0 \neq 0.$$

For example  $\frac{1}{1-x} = 1 + x + x^2 + \cdots$  because  $(1-x)(1 + x + x^2 + \cdots) = 1 + 0x + 0x^2 + \cdots$ .

**Convergence.** When working in  $\mathbb{C}[[x]]$ , we will not consider convergence of sequences or series of complex numbers. In particular, we will never substitute a complex number x into a formal power series A(x).

However, we do need a notion of convergence for sequences in  $\mathbb{C}[[x]]$ . We say that a sequence  $A_0(x), A_1(x), A_2(x), \ldots$  of formal power series **converges** to  $A(x) = \sum_{n\geq 0} a_n x^n$  if  $\lim_{n\to\infty} \deg(A_n(x) - A(x)) = \infty$ ; that is, if for any  $n \in \mathbb{N}$ , the coefficient of  $x^n$  in  $A_m(x)$  equals  $a_n$  for all sufficiently large m. This gives us a useful criterion for convergence of infinite sums and products in  $\mathbb{C}[[x]]$ :

$$\sum_{j=0}^{\infty} A_j(x) \text{ converges} \iff \lim_{j \to \infty} \deg A_j(x) = \infty$$
$$\prod_{j=0}^{\infty} (1 + A_j(x)) \text{ converges} \iff \lim_{j \to \infty} \deg A_j(x) = \infty \qquad (A_j(0) = 0)$$

For example, the infinite sum  $\sum_{n\geq 0} (x+1)^n/2^n$  does not converge in this topology. Notice that the coefficient of  $x^0$  in this sum cannot be obtained through a finite computation; it would require interpreting the infinite sum  $\sum_{n>0} 1/2^n$ . On the other hand, the following infinite sum converges:

$$\sum_{n\geq 0} \frac{1}{n!} \left( -\sum_{m\geq 1} \frac{x^m}{m} \right)^n = 1 - x.$$
(6)

It is clear from the criterion above that this series converges; but why does it equal 1 - x?

**Borrowing from analysis.** In  $\mathbb{C}[[x]]$ , (6) is an algebraic identity which says that the coefficients of  $x^k$  in the left hand side – for which we can give an ugly but finite formula – equal 1, -1, 0, 0, 0, .... If we were to follow a purist algebraic attitude, we would give an algebraic or combinatorial proof of this identity. This is probably possible, but intricate and rather dogmatic. A much simpler approach is to shift towards an analytic attitude, at least momentarily, and recognize that (6) is the Taylor series expansion of

$$e^{-\log(1-x)} = 1 - x$$

for |x| < 1. Then we can just invoke the following simple fact from analysis.

**Theorem 2.1.1.** If two analytic functions are equal in an open neighborhood of 0, then their Taylor series at 0 are equal coefficient-by-coefficient; that is, they are equal as formal power series.

**Composition.** The composition of two series  $A(x) = \sum_{n\geq 0} a_n x^n$  and  $B(x) = \sum_{n\geq 0} b_n x^n$  with  $b_0 = 0$  is naturally defined to be:

$$A(B(x)) = \sum_{n \ge 0} a_n \left( \sum_{m \ge 0} b_m x^m \right)^n.$$

Note that this sum converges if and only if  $b_0 = 0$ . Two very important special cases in combinatorics are the series  $\frac{1}{1-B(x)}$  and  $e^{B(x)}$ .

"Calculus". We define the **derivative** of  $A(x) = \sum_{n>0} a_n x^n$  to be

$$A'(x) = \sum_{n \ge 0} (n+1)a_{n+1}x^n.$$

This formal derivative satisfies the usual laws of derivatives, such as:

$$(A+B)' = A' + B',$$
  $(AB)' = A'B + AB',$   $[A(B(x))]' = A'(B(x))B'(x).$ 

We can still solve differential equations formally. For example, if we know that F'(x) = F(x) and F(0) = 1, then  $(\log F(x))' = F'(x)/F(x) = 1$ , which gives  $\log F(x) = x$  and  $F(x) = e^x$ .

This concludes our discussion on the formal properties of power series. Now let us return to combinatorics.

#### 2.2 Ordinary generating functions

Suppose we are interested in enumerating a family  $\mathcal{A} = \mathcal{A}_0 \sqcup \mathcal{A}_1 \sqcup \mathcal{A}_2 \sqcup \cdots$  of combinatorial structures, where  $\mathcal{A}_n$  is a finite set consists of the objects of "size" n. Denote by |a| the size of  $a \in \mathcal{A}$ . The **ordinary generating function** (ogf) of  $\mathcal{A}$  is

$$A(x) = \sum_{a \in \mathcal{A}} x^{|a|} = a_0 + a_1 x + a_2 x^2 + \cdots$$

where  $a_n$  is the number of elements of size n.

We are not interested in the philosophical question of determining what it means for  $\mathcal{A}$  to be "combinatorial"; we are willing to call  $\mathcal{A}$  a combinatorial structure as long as  $a_n$  is finite for all n. We consider two structures  $\mathcal{A}$  and  $\mathcal{B}$  combinatorially equivalent, and write  $\mathcal{A} \cong \mathcal{B}$ , if A(x) = B(x).

More generally, we may consider a family  $\mathcal{A}$  where each element a is given a weight wt(a) – often a constant multiple of  $x^{|a|}$ , or a monomial in one or more variables  $x_1, \ldots, x_n$ . Again, we require that there are finitely many objects of any given weight. Then we define the **weighted** ordinary generating function of  $\mathcal{A}$  to be the formal power series

$$A_{\mathrm{wt}}(x_1,\ldots,x_n) = \sum_{a\in\mathcal{A}} \mathrm{wt}(a)$$

Examples of combinatorial structures (with their respective size functions) are words on the alphabet  $\{0, 1\}$  (length), domino tilings of rectangles of height 2 (width), or Dyck paths (length). We may weight these objects by  $t^k$  where k is, respectively, the number of 1s, the number of vertical tiles, or the number of returns to the x axis.

## 2.2.1 Operations on combinatorial structures and their generating functions.

There are a few simple but very powerful operations on combinatorial structures, all of which have nice counterparts at the level of ordinary generating functions. Many combinatorial objects of interest may be built up from basic structures using these operations.

**Theorem 2.2.1.** Let  $\mathcal{A}$  and  $\mathcal{B}$  be combinatorial structures.

1. (C = A + B: Disjoint union) If a C-structure of size n is obtained by choosing an A-structure of size n or a B-structure of size n, then

$$C(x) = A(x) + B(x)$$

2. ( $C = A \times B$ : Product) If a C-structure of size n is obtained by choosing an A-structure of size k and a B-structure of size n - k for some k, then

$$C(x) = A(x)B(x)$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the respective A and B-structures.

3. (C = Seq(B): Sequence) Assume  $|B_0| = 0$ . If a C-structure of size n is obtained by choosing a sequence of B-structures of total size n, then

$$C(x) = \frac{1}{1 - B(x)}$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the respective  $\mathcal{B}$ -structures.

4. ( $C = A \circ B$ : Composition) Compositional Formula. Assume that  $|B_0| = 0$ . If a C-structure of size n is obtained by choosing a sequence of B-structures of total size n and placing an A-structure on this set of B-structures, then

$$C(x) = A(B(x))$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the A-structure on its blocks and the weights of the B-structures on the individual blocks.

# 5. ( $C = A^{-1}$ : Inversion) Lagrange Inversion Formula.

(a) Algebraic version. If  $A^{<-1>}(x)$  is the compositional inverse of A(x) then

$$n[x^n]A^{<-1>}(x) = [x^{n-1}]\left(\frac{x}{A(x)}\right)^n.$$

(b) Combinatorial version. Assume  $|\mathcal{A}_0| = 0$ ,  $|\mathcal{A}_1| = 1$ , and let

$$A(x) = x - a_2 x^2 - a_3 x^3 - a_4 x^4 - \cdots$$

where  $a_n$  is the number of  $\mathcal{A}$ -structures of size n for  $n \geq 2$ .

Let an  $\mathcal{A}$ -decorated plane rooted tree (or simply  $\mathcal{A}$ -tree) be a rooted tree T where every internal vertex v has an ordered set  $D_v$  of at least two "children", and each set  $D_v$  is given an  $\mathcal{A}$ -structure. The size of T is the number of leaves.

Let C(x) be the generating function for A-decorated plane rooted trees. Then

$$C(x) = A^{<-1>}(x)$$

This result also holds for weighted structures if the weight of a tree is the product of the weights of the A-structures on its vertices.

In Theorem 2.2.1.5(b), to simplify matters, we introduced signs into A(x). Instead we could let A(x) be the ordinary generating function for  $\mathcal{A}$ -structures, but we would need to give each  $\mathcal{A}$ -tree the sign  $(-1)^m$ , where m is the number of internal vertices. Similarly, we could allow  $a_1 \neq 1$  at the cost of some factors of  $a_1$  on the  $\mathcal{A}$ -trees.

*Proof.* 1. is clear. The identity in 2. is equivalent to  $c_n = \sum_k a_k b_{n-k}$ , which corresponds to the given combinatorial description. Iterating 2., the generating function for k-sequences of  $\mathcal{B}$ -structures is  $B(x)^k$ , so in 3. we have  $C(x) = \sum_k B(x)^k = 1/(1-B(x))$  and in 4. we have  $C(x) = \sum_k a_k B(x)^k$ . The weighted statements follow similarly.

5(b). Observe that, by the Compositional Formula, an  $\mathcal{A} \circ \mathcal{C}$  structure is either (i) an  $\mathcal{A}$ -tree, or (ii) a sequence of  $k \geq 2$   $\mathcal{A}$ -trees  $T_1, \ldots, T_k$  with an  $\mathcal{A}$ -structure on  $\{T_1, \ldots, T_k\}$ .

The structure in (ii) is equivalent to an  $\mathcal{A}$ -tree T, obtained by grafting  $T_1, \ldots, T_k$  at a new root and placing the  $\mathcal{A}$ -structure on its offspring (which contributes a negative sign). This T also arises in (i) with a positive sign. These two appearances of T cancel each other out in  $\mathcal{A}(C(x))$ , and the only surviving tree is the trivial tree  $\bullet$  with one vertex, which only arises once and has weight x. 5(a). Let a **sprig** be a rooted plane tree consisting of a path  $r = v_1 v_2 \cdots v_k = l$  starting at the root r and ending at the leaf l, and at least one leaf hanging from each  $v_i$  and to the right of  $v_{i+1}$  for  $1 \le i \le k-1$ . The trivial tree • is an allowable sprig with k = 1.

An  $\mathcal{A}$ -sprig is a sprig where the children of  $v_i$  are given an  $\mathcal{A}$ -structure for  $1 \leq i \leq k-1$ ; its size is the number of leaves other than l minus  $1.^2$  The right panel of Figure 2.1 shows several  $\mathcal{A}$ -sprigs. An  $\mathcal{A}$ -sprig is equivalent to a sequence of  $\mathcal{A}$ -structures, with weights shifted by -1, so Theorem 2.2.1.3 tells us that

$$\frac{1}{A(x)} = \frac{1}{x} \cdot \frac{1}{1 - (a_2 x + a_3 x^2 + \dots)} = \sum_{n \ge -1} (\# \text{ of } \mathcal{A}\text{-sprigs of size } n) x^n$$

Hence  $[x^{n-1}](x/A(x))^n = [x^{-1}](1/A(x))^n$  is the number of sequences of n sprigs of total size -1 by Theorem 2.2.1.2. We need to show that

 $n \cdot (\# \text{ of } \mathcal{A}\text{-trees with } n \text{ vertices}) = (\# \text{ of sequences of } n \mathcal{A}\text{-sprigs of total size } -1)$ 



Figure 2.1: The map from  $\mathcal{A}$ -trees to sequences of  $\mathcal{A}$ -sprigs.

Conversely, suppose we wish to recover the  $\mathcal{A}$ -tree corresponding to a sequence of sprigs  $S_1, \ldots, S_n$ with  $|S_1| + \cdots + |S_n| = -1$  We must reverse the process, adding  $S_1, \ldots, S_n$  to T one at a time; at each step we must graft the new sprig at the leftmost free branch. Note that after grafting  $S_1, \ldots, S_k$  we are left with  $1 + |S_1| + \cdots + |S_k|$  free branches, so a sequence of sprigs corresponds to a tree if and only if the partial sums  $|S_1| + \cdots + |S_k|$  are non-negative for  $k = 1, \ldots, n-1$ . Finally, it remains to observe that any sequence  $a_1, \ldots, a_n$  of integers adding to -1 has a unique cyclic shift  $a_i, \ldots, a_n, a_1, \ldots, a_{i-1}$  whose partial sums are all non-negative. Therefore, out of the *n* cyclic shifts of  $S_1, \ldots, S_n$ , exactly one of them corresponds to an  $\mathcal{A}$ -tree. The desired result follows.  $\Box$ 

The last step of the proof above is a special case of the **Cycle Lemma** of Dvoretsky and Motzkin [Sta99, Lemma 5.3.7], which is worth stating explicitly. Suppose  $a_1, \ldots, a_n$  is a string of 1s and -1s with  $a_1 + \cdots + a_n = k > 0$ . Then there are exactly k cyclic shifts  $a_i, a_{i+1}, \ldots, a_n, a_1, \ldots, a_{i-1}$  whose partial sums are all non-negative.

<sup>&</sup>lt;sup>2</sup>We momentarily allow negative sizes, since the trivial  $\mathcal{A}$ -sprig • has size -1. Thus we need to compute with Laurent series, which are power series with finitely many negative exponents.

## 2.2.2 Examples.

**Classical applications.** With practice, these simple ideas give very easy solutions to many classical enumeration problems.

- 1. (Trivial classes) It is useful to introduce the trivial class  $\circ$  having only one element of size 0, and the trivial class  $\bullet$  having only one element of size 1. Their generating functions are 1 and x, respectively.
- 2. (Sequences) The slightly less trivial class  $\mathsf{Seq} = \{\emptyset, \bullet, \bullet \bullet, \bullet \bullet \bullet, \ldots\} = \mathsf{Seq}(\bullet)$  contains one set of each size. Its generating function is  $\sum_n x^n = 1/(1-x)$ .
- 3. (Subsets and binomial coefficients) Let Subset consist of the pairs ([n], A) where n is a natural number and A is a subset of [n]. Let the size of that pair be n. A Subset-structure is equivalent to a word of length n in the alphabet  $\{0, 1\}$ , so Subset  $\cong$  Seq $(\{0, 1\})$  where |0| = |1| = 1, and

$$\mathsf{Subset}(x) = \frac{1}{1 - (x^1 + x^1)} = \sum_{n \ge 0} 2^n x^n$$

We can use the extra variable y to keep track of the size of the subset A, by giving ([n], A) the weight  $x^n y^{|A|}$ . This corresponds to giving the letters 0 and 1 weights x and xy respectively, so we get the generating function

$$\mathsf{Subset}_{\mathrm{wt}}(x) = \frac{1}{1 - (x + xy)} = \sum_{n \ge k \ge 0} \binom{n}{k} x^n y^k$$

for the **binomial coefficients**  $\binom{n}{k} = \frac{n!}{k!(n-k)!}$ , which count the k-subsets of [n].

From this generating function, we can easily obtain the main results about binomial coefficients. Computing the coefficient of  $x^n y^k$  in  $\left(\sum {n \choose k} x^n y^k\right) (1 - x - xy) = 1$  gives **Pascal's recurrence** 

$$\binom{n}{k} = \binom{n-1}{k} + \binom{n-1}{k-1}, \qquad n \ge k \ge 1$$

with initial values  $\binom{n}{0} = \binom{n}{n} = 1$ . Expanding  $\text{Subset}_{wt}(x) = \frac{1}{1-x(1+y)} = \sum_{n\geq 0} x^n(1+y)^n$  gives the **Binomial Theorem** 

$$(1+y)^n = \sum_{k=0}^n \binom{n}{k} y^k.$$

4. (Multinomial coefficients) Let  $Words^k \cong Seq(\{1, \ldots, k\})$  consist of the words in the alphabet  $\{1, 2, \ldots, k\}$ . The words of length n are in bijection with the ways of putting n numbered balls into k numbered boxes. The placements having  $a_i$  balls in box i, where  $a_1 + \cdots + a_k = n$ , are enumerated by the **multinomial coefficient**  $\binom{n}{a_1, \ldots, a_k} = \frac{n!}{a_1! \cdots a_k!}$ .

Giving the letter i weight  $x_i$ , we obtain the generating function:

Words<sup>k</sup>
$$(x_1, \dots, x_k) = \sum_{a_1, \dots, a_k \ge 0} \binom{a_1 + \dots + a_k}{a_1, \dots, a_k} x_1^{a_1} \cdots x_k^{a_k} = \frac{1}{1 - x_1 - \dots - x_k}$$

from which we obtain the recurrence

$$\binom{n}{a_1, \dots, a_k} = \binom{n-1}{a_1 - 1, a_2, \dots, a_k} + \dots + \binom{n-1}{a_1, \dots, a_{k-1}, a_k - 1}$$

and the multinomial theorem

$$(x_1 + \dots + x_k)^n = \sum_{\substack{a_1, \dots, a_k \ge 0 \\ a_1 + \dots + a_k = n}} \binom{n}{a_1, \dots, a_k} x_1^{a_1} \cdots x_k^{a_k}.$$

5. (Compositions) A composition of n is a way of writing  $n = a_1 + \cdots + a_k$  as an ordered sum of positive integers  $a_1, \ldots, a_k$ . For example, 523212 is a composition of 15. A composition is just a sequence of natural numbers, so Comp  $\cong$  Seq( $\mathbb{N}$ ) where |a| = a. Therefore

$$\mathsf{Comp}(x) = \frac{1}{1 - (x + x^2 + x^3 + \dots)} = \frac{1 - x}{1 - 2x} = \sum_{n \ge 1} 2^{n - 1} x^n.$$

and there are  $2^{n-1}$  compositions of n.

If we give a composition of n with k summands the weight  $x^n y^k$ , the weighted generating function is:

$$\mathsf{Comp}_{\mathrm{wt}}(x) = \frac{1}{1 - (xy + x^2y + x^3y + \dots)} = \frac{1 - x}{1 - x(1 + y)} = \sum_{n \ge 1} \binom{n - 1}{k - 1} x^n y^k$$

so there are  $\binom{n-1}{k-1}$  compositions of *n* with *k* summands.

6. (Compositions into restricted parts) Given a subset  $A \subseteq \mathbb{N}$ , an A-composition of n is a way of writing n as an ordered sum  $n = a_1 + \cdots + a_k$  where  $a_1, \ldots, a_k \in A$ . The corresponding combinatorial structure is A-Comp  $\cong$  Seq(A) where |a| = a, so

$$A\text{-}\mathsf{Comp}(x) = \frac{1}{1 - (\sum_{a \in A} x^a)}$$

For example, the number of compositions of n into odd parts is the Fibonacci number  $F_{n-1}$ , because the corresponding generating functions is

$$\mathsf{OddComp}(x) = \frac{1}{1 - (x + x^3 + x^5 + \cdots)} = \frac{1 - x^2}{1 - x - x^2} = 1 + \sum_{n \ge 1} F_{n-1} x^n.$$

7. (Multisubsets) Let  $\mathsf{Multiset}^m$  be the collection of multisets consisting of possibly repeated elements of [m]. The size of a multiset is the number of elements, counted with repetition. For example,  $\{1, 2, 2, 2, 3, 5\}$  is a multisubset of [7] of size 6. Then  $\mathsf{Multiset}^m \cong \mathsf{Seq}(\{1\}) \times \cdots \times \mathsf{Seq}(\{m\})$ , where |i| = 1 for  $i = 1, \ldots, m$ , so the corresponding ogf is

$$\mathsf{Multiset}^{m}(x) = \left(\frac{1}{1-x}\right)^{m} = \sum_{n \ge 0} \binom{-m}{n} (-x)^{n},$$

and the number of multisubsets of [m] of size n is  $\binom{m}{n} := (-1)^n \binom{-m}{n} = \binom{m+n-1}{n}$ .

8. (Partitions) A **partition** of n is a way of writing  $n = a_1 + \cdots + a_k$  as an **unordered** sum of positive integers  $a_1, \ldots, a_k$ . We usually write the parts in weakly decreasing order. For example, 532221 is a partition of 15 into 6 parts. Let **Partition** be the family of partitions weighted by  $x^n y^k$  where n is the sum of the parts and k is the number of parts. Then **Partition**  $\cong$  Seq({1})  $\times$  Seq({2})  $\times \cdots$ , where wt(i) =  $x^i y$  for  $i = 1, 2, \ldots$ , so the corresponding ogf is

Partition
$$(x, y) = \left(\frac{1}{1 - xy}\right) \left(\frac{1}{1 - x^2y}\right) \left(\frac{1}{1 - x^3y}\right) \cdots$$

There is no simple explicit formula for the number p(n) of partitions of n, although there is a very elegant and efficient recursive formula. Setting y = -1 in the previous identity, and invoking Euler's pentagonal theorem [Aig07]

$$\prod_{n \ge 0} (1 - x^n) = 1 + \sum_{j \ge 1} (-1)^j \left( x^{j(3j-1)/2} + x^{j(3j+1)/2} \right)$$
(7)

we obtain:

$$p(n) = p(n-1) + p(n-2) - p(n-5) - p(n-7) + p(n-12) + p(n-15) - \cdots$$

where  $1, 2, 5, 7, 12, 15, 22, 26, \ldots$  are the pentagonal numbers.

9. (Partitions into distinct parts) Let DistPartition be the family of partitions into distinct parts, weighted by x<sup>n</sup>y<sup>k</sup> where n is the sum of the parts and k is the number of parts. Then DistPartition ≈ {1, 1} × {2, 2} × · · ·, where wt(i) = x<sup>i</sup>y and wt(i) = 1 for i = 1, 2, . . ., so the corresponding off is

$$\mathsf{DistPartition}(x,y) = (1+xy)(1+x^2y)(1+x^3y)\cdots$$

10. (Partitions into restricted parts) It is clear how to adapt the previous generating functions to partitions where the parts are restricted. For example, the identity

$$\frac{1}{1-x} = (1+x)(1+x^2)(1+x^4)(1+x^8)(1+x^{16})\cdots$$

expresses the fact that every positive integer can be written uniquely in binary notation, as a sum of distinct powers of 2. The identity

$$(1+x)(1+x^2)(1+x^3)(1+x^4)\cdots = \frac{1}{1-x} \cdot \frac{1}{1-x^3} \cdot \frac{1}{1-x^5} \cdot \frac{1}{1-x^7}\cdots$$

which may be proved by writing  $1 + x^k = (1 - x^{2k})/(1 - x^k)$ , expresses that the number of partitions of n into distinct parts equals the number of partitions of n into odd parts.

11. (Partitions with restrictions on the size and the number of parts) Let  $p_{\leq k}(n)$  be the number of partitions of n into at most k parts. This is also the number of partitions of n into parts of size at most k. To see this, represent a partition  $n = a_1 + \cdots + a_j$  as a left-justified array of squares, where the *i*th row has  $a_i$  squares. Each partition  $\lambda$  has a conjugate partition  $\lambda'$ obtained by exchanging the rows and the columns of the Ferrers diagram. Figure 2.2 shows the Ferrers diagram of 431 and its conjugate partition 3221. It is clear that  $\lambda$  has at most k parts if and only if  $\lambda'$  has parts of size at most k.

From the previous discussion it is clear that



Figure 2.2: The Ferrers diagrams of the conjugate partitions 431 and 3221.

Now let  $p_{\leq j,\leq k}(n)$  be the number of partitions of n into at most j parts of size at most k. Then

$$\sum_{n\geq 0} p_{\leq j,\leq k}(n)x^n = \frac{(1-x)(1-x^2)\cdots(1-x^{j+k})}{(1-x)(1-x^2)\cdots(1-x^j)\cdot(1-x)(1-x^2)\cdots(1-x^k)}$$

This is easily proved by induction, using that  $p_{\leq j, \leq k}(n) = p_{\leq j, \leq k-1}(n) + p_{\leq j, \leq k}(n-k)$ .

12. (Even and odd partitions) Setting y = -1 into the generating function for partitions into distinct parts of Example 9, we get

$$\sum_{n \ge 0} (\operatorname{epd}(n) - \operatorname{opd}(n)) x^n = \prod_{j \ge 1} (1 - x^j) = 1 - x - x^2 + x^5 + x^7 - x^{12} - x^{15} + \cdots$$

where epd(n) (resp. opd(n)) counts the partitions of n into an even (resp. odd) number of distinct parts. Euler's pentagonal formula (7) says that epd(n) - opd(n) equals 0 for all n except for the pentagonal numbers, for which it equals 1 or -1.

There are similar results for partitions into distinct parts coming from a given set S.

• When S is the set of Fibonacci numbers, the coefficients of the generating function

$$\prod_{n \ge 1} (1 - x^{F_n}) = 1 - x - x^2 + x^4 + x^7 - x^8 + x^{11} - x^{12} - x^{13} + x^{14} + \cdots$$

are also equal to 0, 1, or -1. [Rob96, Ard04]

• This is also true for any "k-Fibonacci sequence"  $S = \{a_1, a_2, \ldots\}$  given by  $a_n = a_{n-1} + \cdots + a_{n-k}$  for n > k and  $a_j > a_{j-1} + \cdots + a_1$  for  $1 \le j \le k$ . [Dia09]

• The result also holds trivially for  $S = \{2^j : j \in \mathbb{N}\}$  since there is a unique partition of any n into distinct powers of 2.

These three results seem qualitatively different from (and increasingly less surprising than) Euler's result, as these sequences S grow much faster than  $\{1, 2, 3, ...\}$ , and S-partitions are sparser. Can more be said about the sets S of positive integers for which the coefficients of  $\prod_{n \in S} (1 - x^n)$  are all 1,0 or -1? 13. (Set partitions) A set partition of a set S is an unordered collection of pairwise disjoint sets S<sub>1</sub>,..., S<sub>k</sub> whose union is S. The family of set partitions with k parts is SetPartition<sup>k</sup> ≅ •×Seq({1})ו×Seq({1,2})×···×•×Seq({1,2,...,k}), where the singleton • and all numbers i have size 1. To see this, we regard a word such as w = •11 • 1221 • 31 as an instruction manual to build a set partition S<sub>1</sub>,..., S<sub>k</sub>. The jth symbol w<sub>j</sub> tells us where to put the number j: if w<sub>j</sub> is a number h, we add j to the part S<sub>h</sub>; if w<sub>j</sub> is the ith •, then we add j to a new part S<sub>i</sub>. The sample word above leads to the partition {1, 2, 3, 5, 8, 11}, {4, 6, 7}, {9, 10}. This process is easily reversible. It follows that

$$\sum_{n\geq 0} S(n,k)x^n = \frac{x}{1-x} \cdot \frac{x}{1-2x} \cdot \dots \cdot \frac{x}{1-kx},$$

where S(n, k) is the number of set partitions of [n] into k parts. These numbers are called the **Stirling numbers of the second kind**.

The equation  $(1 - kx) \sum_{n \ge 0} S(n, k) x^n = x \sum_{n \ge 0} S(n, k - 1) x^n$  gives the recurrence

$$S(n,k) = kS(n-1,k) + S(n-1,k-1), \qquad 1 \le k \le n,$$

with initial values S(n,0) = S(n,n) = 1. Note the great similarity with Pascal's recurrence.

- 14. (Catalan structures) It is often said that if you encounter a new family of mathematical objects, and you have to guess how many objects of size n there are, you should guess "the Catalan number  $C_n = \frac{1}{n+1} {\binom{2n}{n}}$ ." The Catalan family has more than 200 incarnations in combinatorics and other fields [Sta99, Sta]; let us see three important ones.
  - (a) (Plane binary trees) A plane binary tree is a rooted tree where every internal vertex has a left and a right child. Let PBTree be the family of plane binary trees, where a tree with n internal vertices (and n + 1 leaves) has size n. A tree is either the trivial tree  $\circ$  of size 0, or the grafting of a left subtree and a right subtree at the root  $\bullet$ , so PBTree  $\cong \circ +$  (PBTree  $\times \bullet \times$  PBTree). It follows that the generating function for plane binary trees satisfies

$$T(x) = 1 + T(x) x T(x).$$

We may use the quadratic formula and the binomial theorem<sup>3</sup> to get

$$T(x) = \frac{1 - \sqrt{1 - 4x}}{2x} = \sum_{n \ge 0} \frac{1}{n + 1} \binom{2n}{n} x^n.$$

It follows that the number of plane binary trees with n internal vertices (and n + 1 leaves) is the **Catalan number**  $C_n = \frac{1}{n+1} {\binom{2n}{n}}$ .

(b) (Triangulations) A **triangulation** of a convex polygon is a subdivision into triangles using only the diagonals of P. A triangulation of an (n + 2)-gon has n triangles; we say it has size n. If we fix an edge e of P, then a triangulation of P is obtained by choosing the triangle T that will cover e, and then choosing a triangulation of the two polygons to the left and to the right of T. Therefore Triang  $\cong \circ + (\text{Triang} \times \bullet \times \text{Triang})$  and the number of triangulations of an (n + 2)-gon is also the Catalan number  $C_n$ .

<sup>&</sup>lt;sup>3</sup>Since this is the first time we are using the quadratic formula, let us do it carefully. Rewrite the equation as  $(1-2xT(x))^2 = 1-4x$ , or  $(1-2xT(x) - \sqrt{1-4x})(1-2xT(x) + \sqrt{1-4x}) = 0$ . Since  $\mathbb{C}[[x]]$  is an integral domain, one of the factors must be 0. From the constant coefficients we see that it must be the first factor.

(c) (Dyck paths) A **Dyck path** P of length n is a path from (0,0) to (2n,0) which uses the steps (1,1) and (1,-1) and never goes below the x-axis. Say P is **irreducible** if it touches the x axis exactly twice, at the beginning and at the end. Let D(x) and I(x)be the generating functions for Dyck paths and irreducible Dyck paths.

A Dyck path is equivalent to a sequence of **irreducible** Dyck paths. Also, an irreducible path of length n is the same as a Dyck path of length n - 1 with an additional initial and final step. Therefore

$$D(x) = \frac{1}{1 - I(x)}, \qquad I(x) = xD(x)$$

from which it follows that  $D(x) = \frac{1-\sqrt{1-4x}}{2x}$  as well, and the number of Dyck paths of length n is also the Catalan number.

Generatingfunctionology gives us fairly easy algebraic proofs that these three families are enumerated by the Catalan numbers. Once we have discovered this fact, the temptation to search for nice bijections is hard to resist.

Our algebraic analysis suggests a bijection  $\phi$  from (b) to (a). The families of plane binary trees and triangulations grow under the same recursive recipe, and so we can let the bijection grow with them, mapping a triangulation  $T \times \bullet \times T'$  to the tree  $\phi(T) \times \bullet \times \phi(T')$ . A non-recursive description of the bijection is the following. Consider a triangulation T of the polygon P, and fix an edge e. Put a vertex inside each triangle of T, and a vertex outside P next to each edge other than e. Then connect each pair of vertices separated by an edge. Finally, root the resulting tree at the vertex adjacent to e. This bijection is illustrated in Figure 2.3.



Figure 2.3: The bijection from triangulations to plane binary trees.

A bijection from (a) to (c) is less obvious from our algebraic computations, but is still not difficult to obtain. Given a plane binary tree T of size n, prune all the leaves to get a tree T' with n vertices. Now walk around the periphery of the tree, starting on the left side from the root, and continuing until we traverse the whole tree. Record the walk in a Dyck path D(T): every time we walk up (resp. down) a branch we take a step up (resp. down) in D(T). One easily checks that this is a bijection.

Even if it may be familiar, it is striking that two different (and straightforward) algebraic computations show us that two families of objects that look quite different are in fact equivalent combinatorially. Although a simple, elegant bijection can often explain the connection



Figure 2.4: The bijection from plane binary trees to Dyck paths.

between two families more transparently, the algebraic approach is sometimes simpler, and better at discovering such connections.

15. (k-Catalan structures) Let  $\mathsf{PTree}_k$  be the class of **plane** k-ary trees, where every vertex that is not a leaf has k ordered children; let the size of such a tree be its number of leaves. In the sense of Theorem 2.2.1.5(a), this is precisely an  $\mathcal{A}$ -tree, where  $\mathcal{A} = \{\bullet, \bullet^k\}$  consisting of one structure of size 1 and one of size k. Therefore  $\mathsf{PTree}_k = (x - x^k)^{\leq -1 \geq}$ . Lagrange inversion then gives

$$m[x^m]A^{<-1>}(x) = [x^{m-1}] \left(\frac{1}{1-x^{k-1}}\right)^m = [x^{m-1}] \sum_{n \ge 0} \binom{m+n-1}{n} x^{(k-1)n}.$$

It follows that the a plane k-ary tree must have m = (k - 1)n + 1 leaves for some integer n, and the number of such trees is the k-Catalan number

$$C_n^k = \frac{1}{(k-1)n+1} \binom{kn}{n}.$$

This is an alternative way to compute the ordinary Catalan numbers  $C_n = C_n^2$ .

The k-Catalan number  $C_n^k$  also has many different interpretations [HLM08]; we mention two more. It counts the subdivisions of an (n(k-1)+2)-gon P into (necessarily n) (k+1)-gons using diagonals of P, and the paths from (0,0) to (n, (k-1)n) with steps (0,1) and (1,0)that never rise above the line y = (k-1)x.

**Other applications.** Let us now discuss a few other interesting applications which illustrate the power of Theorem 2.2.1.

16. (Motzkin paths) The **Motzkin number**  $M_n$  is the number of paths from (0,0) to (n,0) using the steps (1,1), (1,-1), and (1,0) which never go below the x-axis. Imitating our argument for Dyck paths, we obtain a formula for the generating function:

$$M(x) = \frac{1}{1 - (x + xM(x)x)} \implies M(x) = \frac{1 - x - \sqrt{1 - 2x - 3x^2}}{2x^2}.$$

The quadratic equation  $x^2M^2 + (x-1)M + 1 = 0$  gives rise to the quadratic recurrence  $M_n = M_{n-1} + \sum_i M_i M_{n-2-i}$ . The fact that M(x) satisfies a polynomial equation leads to a more efficient recurrence:

$$(n+2)M_n = (2n+1)M_{n-1} + (3n-3)M_{n-2}$$

We will see this in Section 2.4.2.

17. (Schröder paths) The (large) Schröder number  $r_n$  is the number of paths from (0,0) to (2n,0) using steps NE = (1,1), SE = (1,-1), and E = (2,0) which stays above the x axis. Their generating function satisfies R(x) = 1/(1-x-xR(x)), and therefore

$$R(x) = \frac{1 - x - \sqrt{1 - 6x + x^2}}{2x}$$

Theorem 2.2.1.3 is useful when we are counting combinatorial objects which "factor" uniquely into an ordered "product" of "irreducible" objects. It tells us that we can count all objects if and only if we can count the irreducible ones. We have already used that idea several times; let us see it in action in some other examples.

18. (Domino tilings of rectangles) In Section 1 we let  $a_n$  be the number of domino tilings of a  $2 \times n$  rectangle. Such a tiling is uniquely a sequence of blocks, where each block is either a vertical domino (of width 1) or two horizontal dominoes (of width 2). This truly explains the formula:

$$A(x) = \frac{1}{1 - (x + x^2)}$$

Similarly, if  $a_{m,n}$  is the number of domino tilings of a  $2 \times n$  rectangle using v vertical tiles, we immediately obtain:

$$\sum_{m,n\geq 0} a_{m,n} v^m x^n = \frac{1}{1 - (vx + x^2)}.$$

Sometimes the enumeration of irreducible structures is not immediate, but still tractable.

19. (Monomer-dimer tilings of rectangles) Let T(2, n) be the number of tilings of a  $2 \times n$  rectangles with dominoes and unit squares. Say a tiling is irreducible if it does not contain an internal vertical line from top to bottom. Then Tilings  $\cong$  Seq(IrredTilings). It now takes some thought to recognize the irreducible tilings:



Figure 2.5: The irreducible tilings of  $2 \times n$  rectangles into dominoes and unit squares.

There are 3 irreducible tilings of length 2, and 2 of every other length greater than or equal to 1. Therefore

$$\sum_{n\geq 0} T(2,n)x^n = \frac{1}{1 - (2x + 3x^2 + 2x^3 + 2x^4 + \dots)} = \frac{1 - x}{1 - 3x - x^2 + x^3}$$

We will see in Theorem 2.4.1.2 that this gives  $T(2,n) \sim c \cdot \alpha^n$  where  $\alpha \approx 3.214...$  is the inverse of the smallest positive root of the denominator.

Sometimes the enumeration of **all** objects is easier than the enumeration of the irreducible ones. In that case we can use Theorem 2.2.1.3 in the opposite direction.

20. (Irreducible permutations) A permutation  $\pi$  of [n] is **irreducible** if it does not factor as a permutation of  $\{1, \ldots, m\}$  and a permutation of  $\{m + 1, \ldots, n\}$  for  $1 \le m < n$ ; that is, if  $\pi([m]) \ne [m]$  for all  $1 \le m < n$ . Clearly every permutation factors uniquely into irreducibles, so

$$\sum_{n\geq 0} n! x^n = \frac{1}{1-\operatorname{IrredPerm}(x)}.$$

This gives the series for IrredPerm.

There are many interesting situations where it is possible, but not at all trivial, to decompose the objects that interest us into simpler structures. To a combinatorialist this is good news – the techniques of this section are useful tools, but are not enough; there is no shortage of interesting work to do. Here is a great example.

21. (Domino towers) [GBV88, BP93, Zei] A **domino tower** is a stack of horizontal  $2 \times 1$  bricks in a brickwork pattern, so that no brick is directly above another brick, such that the bricks on the bottom level are contiguous, and every higher brick is (half) supported on at least one brick in the row below it. Let the size of a domino tower be the number of bricks.



Figure 2.6: A domino tower of 19 bricks.

Remarkably, there are  $3^{n-1}$  domino towers consisting of *n* bricks. Equally remarkably, no simple bijection is known. The nicest argument known is as follows.



Figure 2.7: The decomposition of a domino tower into a pyramid and three half-pyramids.

We decompose a domino tower x into smaller pieces, as illustrated in Figure 2.7. Each new piece is obtained by pushing up the leftmost remaining brick in the bottom row, dragging with it all the bricks encountered along the way. The first piece p will be a **pyramid**, which we define to be a domino tower with only one brick in the bottom row. All subsequent pieces  $h_1, \ldots, h_k$  are **half-pyramids**, which are pyramids containing no bricks to the left of the bottom brick. This decomposition is reversible. To recover x, we drop  $h_k, h_{k-1}, \ldots, h_1, p$ from the top in that order; each piece is dropped in its correct horizontal position, and some of its bricks may get stuck on the previous pieces. This shows that the corresponding combinatorial classes satisfy  $X \cong P \times \text{Seq}(H)$ .



Figure 2.8: A pyramid and its decomposition into half-pyramids.

Similarly, we may decompose a pyramid p into half-pyramids, as shown in Figure 2.8. Each new half-pyramid is obtained by pushing up the leftmost remaining brick (which is not necessarily in the bottom row), dragging with it all the bricks that it encounters along the way. This shows that  $P \cong Seq_{>1}(H) := H + (H \times H) + \cdots$ .



Figure 2.9: A (non-half-pyramid) pyramid and its decomposition into two half-pyramids and a bottom brick.

Finally consider a half-pyramid h; there are two cases. If there are other bricks on the same horizontal position as the bottom brick, consider the lowest such brick, and push it up, dragging with it all the bricks it encounters along the way, obtaining a half-pyramid  $h_1$ . Now remove the bottom brick; what remains is a half-pyramid  $h_2$ . This is shown in Figure 2.9. As before, we can recover h from  $h_1$  and  $h_2$ . On the other hand, if there are no bricks above the bottom brick, removing the bottom brick leaves either a half-pyramid or the empty set. Therefore  $H \cong (H \times \bullet \times H) + (\bullet \times H) + \bullet$ .

The above relations correspond to the following identities for the corresponding ogfs:

$$X = \frac{P}{1-H}, \qquad P = \frac{H}{1-H}, \qquad H = xH^2 + xH + x$$

Surprisingly cleanly, we obtain  $X(x) = x/(1-3x) = \sum_{n\geq 1} 3^{n-1}x^n$ . This proves that there are  $3^{n-1}$  domino towers of size n.

Although we do not need this here, it is worth noting that half-pyramids are enumerated by Motzkin numbers; their generating functions are related by H(x) = xM(x).

## 2.3 Exponential generating functions

Ordinary generating functions are usually not well suited for counting combinatorial objects with a labelled ground set. In such situations, exponential generating functions are a more effective tool.

Consider a family  $\mathcal{A} = \mathcal{A}_0 \sqcup \mathcal{A}_1 \sqcup \mathcal{A}_2 \sqcup \cdots$  of labelled combinatorial structures, where  $\mathcal{A}_n$  consists of the structures that we can place on the ground set  $[n] = \{1, \ldots, n\}$  (or, equivalently, on any other labeled ground set of size n). The **exponential generating function** (egf) of  $\mathcal{A}$  is

$$A(x) = \sum_{a \in \mathcal{A}} \frac{x^{|a|}}{|a|!} = a_0 \frac{x^0}{0!} + a_1 \frac{x^1}{1!} + a_2 \frac{x^2}{2!} + a_3 \frac{x^3}{3!} + \cdots$$

where  $a_n$  is the number of elements of size n.

We may again assign a weight wt(a) to each object a, usually a monomial in variables  $x_1, \ldots, x_n$ , and consider the **weighted egf** of  $\mathcal{A}$  to be the formal power series

$$A_{\mathrm{wt}}(x_1,\ldots,x_n,x) = \sum_{a \in \mathcal{A}} \mathrm{wt}(a) \frac{x^{|a|}}{|a|!}$$

Examples of combinatorial structures (with their respective size functions) are permutations (number of elements), graphs (number of vertices), or set partitions (size of the set). We may weight these objects by  $t^k$  where k is, respectively, the number or cycles, the number of edges, or the number of parts.

#### **2.3.1** Operations on labeled structures and their exponential generating functions.

Again, there are some simple operations on *labelled* combinatorial structures, which correspond to simple algebraic operations on the exponential generating functions. Starting with a few simple structures, these operations are sufficient to generate many interesting combinatorial structures. This will allow us to compute the exponential generating functions for those structures.

**Theorem 2.3.1.** Let  $\mathcal{A}$  and  $\mathcal{B}$  be labeled combinatorial structures.

1. (C = A + B: Disjoint union) If a C-structure on a finite set S is obtained by choosing an A-structure on S or a B-structure on S, then

$$C(x) = A(x) + B(x)$$

2. (C = A \* B: Labelled Product) If a C-structure on a finite set S is obtained by partitioning S into disjoint sets  $S_1$  and  $S_2$  and putting an A-structure on  $S_1$  and a B-structure on  $S_2$ , then

$$C(x) = A(x)B(x)$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the respective A and B-structures.

3.  $(C = Seq_*(B): Labeled Sequence)$  If a C-structure on a finite set S is obtained by choosing an ordered partition of S into a sequence of blocks and putting a B-structure on each block, then

$$C(x) = \frac{1}{1 - B(x)}$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the respective  $\mathcal{B}$  structures.

4. (C = Set(B): Set) Exponential Formula. If a C-structure on a finite set S is obtained by choosing an unordered partition of S into a set of blocks and putting a B-structure on each block, then

$$C(x) = e^{B(x)}$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the respective  $\mathcal{B}$ -structures.

In particular, if  $c_k(n)$  is the number of C-structures of an n-set which decompose into k components (B-structures), we have

$$\sum_{n,k,\ge 0} c_k(n) \frac{x^n}{n!} y^k = e^{yB(x)} = C(x)^y$$

5. ( $C = A \circ B$ : Composition) Compositional Formula. If a C-structure on a finite set S is obtained by choosing an unordered partition of S into a set of blocks, putting a B-structure on each block, and putting an A-structure on the set of blocks, then

$$C(x) = A(B(x))$$

This result also holds for weighted structures if the weight of a C-structure is the product of the weights of the A structure on its set of blocks and the weights of the B structures on the individual blocks.

Again, Theorem 2.3.2.3 is useful when we are counting combinatorial objects which "decompose" uniquely as a set of "indecomposable" objects. It tells us that we can count all objects if and only if we can count the indecomposable ones (or vice versa). Amazingly, in Theorem 2.3.2.4 we also obtain for free the finer enumeration of the objects by their number of components.

Proof. 1. is clear. The identity in 2. is equivalent to  $c_n = \sum_k {\binom{n}{k}} a_k b_{n-k}$ , which corresponds to the given combinatorial description. Iterating 2., we see that the exponential generating functions for k-sequences of  $\mathcal{B}$ -structures is  $B(x)^k$ , and hence the one for k-sets of  $\mathcal{B}$ -structures is  $B(x)^k/k!$ . This readily implies 3, 4, and 5. The weighted statements follow similarly.

The following statements are perhaps less fundamental, but also useful.

**Theorem 2.3.2.** Let  $\mathcal{A}$  be a labeled combinatorial structure.

1. ( $C = A_+$ : Shifting) If a C-structure on S is obtained by adding a new element t to S and choosing an A-structure on  $S \cup \{t\}$ , then

$$C(x) = A'(x).$$

2.  $(\mathcal{C} = \mathcal{A}_{\bullet}: Rooting)$  If a C-structure on S is a rooted  $\mathcal{A}$ -structure, obtained by choosing an  $\mathcal{A}$ -structure on S and an element of S called the **root**, then

$$C(x) = xA(x).$$

3. (Sieving by parity of size) If the C-structures are precisely the A-structures of even size,

$$C(x) = \frac{A(x) + A(-x)}{2}.$$

4. (Sieving by parity of components) Suppose A-structures decompose uniquely into components, so  $\mathcal{A} = \mathsf{Set}(\mathcal{B})$  for some  $\mathcal{B}$ . If the C-structures are the A-structures having only components of even size,

$$C(x) = \sqrt{A(x)A(-x)}.$$

5. (Sieving by parity of number of components) Suppose  $\mathcal{A}$ -structures decompose uniquely into components, so  $\mathcal{A} = \mathsf{Set}(\mathcal{C})$  for some  $\mathcal{C}$ . If the  $\mathcal{C}$ -structures are precisely the  $\mathcal{A}$ -structures having an even number of components,

$$C(x) = \frac{1}{2} \left( A(x) + \frac{1}{A(x)} \right).$$

Similar sieving formulas hold modulo k for any  $k \in \mathbb{N}$ .

*Proof.* We have  $c_n = a_{n+1}$  in 1.,  $c_n = na_n$  in 2., and  $c_n = \frac{1}{2}(a_n + (-1)^n a_n)$  in 3., from which the generating function formulas follow. Combining 3. with the Exponential Formula we obtain 4. and 5.

Similarly we see that the generating function for  $\mathcal{A}$ -structures whose size is a multiple of k is  $\frac{1}{k} (A(x) + A(\omega x) + \cdots + A(\omega^{k-1}x))$  where  $\omega$  is a primitive kth root of unity. If we wish to count elements of size  $i \mod k$ , we use 1. to shift this generating function i times.

# 2.3.2 Examples.

**Classical applications.** Once again, these simple ideas give very easy solutions to many classical enumeration problems.

(Trivial classes) Again we consider the trivial classes ∘ with only one element of size 0, and
 with only one element of size 1. Their exponential generating functions are 1 and x, respectively.

2. (Sets) A slightly less trivial class of Sets contains one set of each size. We also let  $Sets_{\geq 1}$  denote the class of non-empty sets, with generating function  $e^x - 1$ . The exponential generating functions are

$$Sets(x) = e^x$$
,  $Sets_{>1}(x) = e^x - 1$ .

3. (Set Partitions) In Section 2.3.2 we found the ordinary generating function for Stirling numbers S(n,k) for a given k; but in fact it is easier to use exponential generating functions. Simply notice that SetPartition  $\cong$  Set(Set<sub> $\geq 1$ </sub>), and the Weighted Exponential Formula then gives

SetPartition
$$(x, y) = \sum_{n,k \ge 0} S(n,k) \frac{x^n}{n!} y^k = e^{y(e^x - 1)}.$$

4. (Permutations) Let  $\operatorname{Perm}_n$  consist of the n! permutations of [n]. A permutation is a labeled sequence of singletons, so  $\operatorname{Perm} = \operatorname{Seq}_*(\bullet)$ , and the generating function for permutations is

$$\operatorname{Perm}(x) = \sum_{n \ge 0} n! \frac{x^n}{n!} = \frac{1}{1-x}.$$

5. (Cycles) Let  $\mathsf{Cycle}_n$  consist of the cyclic orders of [n]. These are the ways of arranging  $1, \ldots, n$  around a circle, where two orders are the same if they differ by a rotation of the circle. There is an *n*-to-1 mapping from permutations to cyclic orders obtained by wrapping a permutation around a circle, so

$$Cycle(x) = \sum_{n} (n-1)! x^n / n! = -\log(1-x).$$

There is a more indirect argument which will be useful to us later. Recall that a permutation  $\pi$  can be written uniquely as a (commutative) product of disjoint cycles of the form  $(i, \pi(i), \pi^2(i), \ldots, \pi^{k-1}(i))$  where k is the smallest index such that  $\pi^k(i) = i$ . For instance, the permutation 835629741 can be written in cycle notation as (18469)(235)(7). Then  $\mathsf{Perm} = \mathsf{Set}(\mathsf{Cycle})$  so  $1/(1-x) = e^{\mathsf{Cycle}(x)}$ .

6. (Permutations by number of cycles) The (signless) **Stirling number of the first kind** c(n,k) is the number of permutations of n having k cycles. The Weighted Exponential Formula gives

$$\sum_{n,k\geq 0} c(n,k) \frac{x^n}{n!} y^k = e^{y\mathsf{Cycle}(x)} = \left(\frac{1}{1-x}\right)^y = \sum_{n\geq 0} y(y+1)\cdots(y+n-1)\frac{x^n}{n!}$$

so the Stirling numbers c(n,k) of the first kind are the coefficients of the polynomial  $y(y + 1) \cdots (y + n - 1)$ .

**Other applications.** The applications of these techniques are countless; let us consider a few more applications, old and recent.

7. (Permutations by cycle type) The **type** of a permutation  $\pi \in S_n$  is type $(w) = (c_1, \ldots, c_n)$ where  $c_i$  is the number of cycles of length *i*. For indeterminates  $\mathbf{t} = (t_1, \ldots, t_n)$ , let  $\mathbf{t}^{\text{type}(w)} =$   $t_1^{c_1} \cdots t_n^{c_n}$ . The **cycle indicator** of the symmetric group  $S_n$  is  $Z_n = \frac{1}{n!} \sum_{w \in S_n} \mathbf{t}^{\operatorname{type}(w)}$ . The Weighted Exponential Formula immediately gives

$$\sum_{n>0} Z_n x^n = e^{t_1 x + t_2 x^2/2 + t_3 x^3/3 + \dots}$$

Let us discuss two special cases of interest.

8. (Derangements) A derangement of [n] is a permutation such that  $\pi(i) \neq i$  for all  $i \in [n]$ . Equivalently, a derangement is a permutation with no cycles of length 1. It follows that Derangement = Set(Cycle\_{>2}), so the number  $d_n$  of derangements of [n] is given by

Derangement
$$(x) = \sum_{n \ge 0} d_n \frac{x^n}{n!} = e^{-\log(1-x)-x} = e^{-x} + xe^{-x} + x^2e^{-x} + \cdots$$

which leads to the explicit formula:

$$d_n = n! \left( 1 - \frac{1}{1!} + \frac{1}{2!} - \frac{1}{3!} + \dots \pm \frac{1}{n!} \right) \sim \frac{n!}{e}.$$

9. (Involutions) An **involution** of [n] is a permutation w such that  $w^2$  is the identity. Equivalently, an involution is a permutation with cycles of length 1 and 2, so the number  $i_n$  of involutions of [n] is given by

$$\operatorname{Inv}(x) = \sum_{n \ge 0} i_n \frac{x^n}{n!} = e^{x + \frac{x^2}{2}}$$

Note that  $\ln v'(x) = (x+1)\ln v(x)$ , which gives  $i_n = i_{n-1} + (n-1)i_{n-2}$ . In Section 2.4.2 we will explain the theory of D-finite power series, which turns differential equations for power series into recurrences for the corresponding sequences.

- 10. (Trees.) A **tree** is a connected graph with no cycles. Consider a "birooted" tree (T, a, b) on [n] with two (possibly equal) root vertices a and b. Regard the unique path  $a = v_0, v_1, \ldots, v_k = b$  as a "spine" for T; the rest of the tree consists of rooted trees hanging from the  $v_i$ s; direct their edges towards the spine. Now regard  $v_1 \ldots v_k$  as a permutation in one-line notation, and rewrite it in cycle notation. This transforms (T, a, b) into a directed graph consisting of a disjoint collection of cycles with trees directed towards them. Every vertex has outdegree 1, so this defines a function  $f : [n] \rightarrow [n]$ . A moment's thought will convince us that this is a bijection. Therefore there are  $n^n$  birooted trees on [n], and hence there are  $n^{n-2}$  trees on [n].
- 11. (Trees, revisited.) Let us count trees in a different way. Let a **rooted tree** be a tree with a special vertex called the root, and a **planted forest** be a graph with no cycles where each connected component has a root. Let  $t_n, r_n, f_n$  and T(x), R(x) = xT(x), F(x) be the sequences and exponential generating functions enumerating trees, rooted trees, and planted forests, respectively.

Planted forests are vertex-disjoint unions of rooted trees, so  $F(x) = e^{R(x)}$ . Also, as illustrated in Figure 2.11, a rooted tree T consists of a root attached to the roots of a planted forest, so R(x) = xF(x). It follows that  $x = R(x)e^{-R(x)}$ , so

$$R(x) = (xe^{-x})^{<-1>}.$$



Figure 2.10: A tree on [8] birooted at 3 and 5, and the corresponding function  $f: [8] \to [8]$ .



Figure 2.11: A rooted tree seen as a root attached to the roots of a planted forest.

Lagrange inversion gives  $n \cdot \frac{r_n}{n!} = [x^{n-1}]e^{nx} = \frac{n^{n-1}}{(n-1)!}$ , so  $r_n = n^{n-1}, \qquad f_n = (n+1)^{n-1}, \qquad t_n = n^{n-2}.$ 

We state a finer enumeration; see [Sta99, Theorem 5.3.4] for a proof. The **degree sequence** of a rooted forest on [n] is  $(\deg 1, \ldots, \deg n)$  where  $\deg i$  is the number of children of *i*. For example the degree sequence of the rooted tree in Figure 2.11 is (3, 1, 0, 0, 0, 0, 2, 0, 2). Then the number of planted forests with a given degree sequence  $(d_1, \ldots, d_n)$  and (necessarily)  $k = n - (d_1 + \cdots + d_n)$  components is

$$\binom{n-1}{k-1}\binom{n-k}{d_1,\ldots,d_n}.$$

The number of forests on [n] is given by a more complicated alternating sum; see [Tak90].

- 12. (Permutations, revisited) Here is an unnecessarily complicated way of proving there are n! permutations of [n]. A permutation  $\pi$  of [n + 1] decomposes uniquely as a concatenation  $\pi = L(n + 1)R$  for permutations L and R of two complementary subsets of [n]. Therefore Shift(Perm) = (Perm) \* (Perm), and the generating function P(x) for permutations satisfies  $P'(x) = P(x)^2$  with P(0) = 1. Solving this differential equation gives  $P(x) = \frac{1}{1-x} = \sum_{n \ge 0} n! \frac{x^n}{n!}$
- 13. (Alternating permutations) The previous argument was gratuitous for permutations, but it will now help us to enumerate the class Alt of **alternating** permutations w, which satisfy  $w_1 < w_2 > w_3 < w_4 > \cdots$ . The **Euler numbers** are  $E_n = |\mathsf{Alt}_n|$ ; let E(x) be their exponential generating function. We will need the class RevAlt of permutations w with  $w_1 > w_2 < w_3 > w_4 < \cdots$ . The map  $w = w_1 \dots w_n \mapsto w' = (n+1-w_1) \dots (n+1-w_n)$ , on permutations of [n] shows that Alt  $\cong$  RevAlt.

Now consider alternating permutations L and R of two complementary subsets of [n]. For  $n \ge 1$ , exactly one of the permutations L(n+1)R and L'(n+1)R is alternating or reverse alternating, and every such permutation arises uniquely in that way. For n = 0 both are alternating. Therefore Shift(Alt + RevAlt) = Alt \* Alt, so  $2E'(x) = E(x)^2 + 1$  with E(0) = 1. Solving this differential equation we get

$$E(x) = \sum_{n \ge 0} E_n \frac{x^n}{n!} = \sec x + \tan x$$

Therefore  $\sec x$  and  $\tan x$  enumerate the alternating permutations of even and odd length, respectively. The Euler numbers are also called secant and tangent numbers. This surprising connection allows us to give combinatorial interpretations of various trigonometric identities, such as  $1 + \tan^2 x = \sec^2 x$ .

14. (Graphs) Let g(v) and  $g_{\text{conn}}(v)$  be the number of simple<sup>4</sup> graphs and connected graphs on [v], respectively. The Exponential Formula tells us that their exponential generating functions are related by  $G(x) = e^{G_{\text{conn}}(x)}$ . In this case it is hard to count the connected graphs directly, but it is easy to count all graphs: to choose a graph we just have to decide whether each edge is present or not, so  $g(v) = 2^{\binom{v}{2}}$ . This gives us

$$\sum_{v \ge 0} g_{\text{conn}}(v) \frac{x^v}{v!} = \log\left(\sum_{v \ge 0} 2^{\binom{v}{2}} \frac{x^v}{v!}\right)$$

We may easily adjust this computation to account for edges and components. There are  $\binom{v(v-1)/2}{e}$  graphs on [v] with e edges; say g(v, c, e) of them have c components, and give them weight  $y^c z^e$ . Then

$$\sum_{v,c,e\geq 0} g(v,c,e) \frac{x^v}{v!} y^c z^e = \left(\sum_{v,e\geq 0} \binom{\binom{v}{2}}{e} \frac{x^v}{v!} z^e\right)^y = F(x,1+z)^y$$

where

$$F(\alpha,\beta) = \sum_{n \ge 0} \frac{\alpha^n \beta^{\binom{n}{2}}}{n!}$$

is the **deformed exponential function** of [Sok12].

15. (Signed Graphs) A simple **signed graph** G is a set of vertices, with at most one "positive" edge and one "negative" edge connecting each pair of vertices. We say G is connected if and only if its underlying graph  $\overline{G}$  (ignoring signs) is connected. A cycle in G corresponds to a cycle of  $\overline{G}$ ; we call it balanced if it contains an even number of negative edges, and unbalanced otherwise. We say that G is balanced if all its cycles are balanced. Let  $s(v, c_+, c_-, e)$  be the number of signed graphs with v vertices, e edges,  $c_+$  balanced components, and  $c_-$  unbalanced components; we wish to compute

$$S(x, y_+, y_-, z) = \sum_{G \text{ signed graph}} s(v, c_+, c_-, e) \frac{x^v}{v!} y_+^{c_+} y_-^{c_-} z^e.$$

<sup>&</sup>lt;sup>4</sup>containing no multiple edges or loops

#### We follow [ACH14].

Let  $S(x, y_+, y_-, z)$ ,  $B(x, y_+, z)$ ,  $C_+(x, z)$ , and  $C_-(x, z)$  be the generating functions for signed, balanced, connected balanced, and connected unbalanced graphs, respectively. The Weighted Exponential Formula gives:

$$B = e^{y_+ C_+}, \qquad S = e^{y_+ C_+ + y_- C_-}$$

so if we can compute  $C_+$  and  $C_-$  we will obtain B and S. In turn, these equations give

$$C_{+}(x,z) = \frac{1}{2}\log B(x,2,z), \qquad C_{+}(x,z) + C_{-}(x,z) = \log S(x,1,1,z)$$

and we now compute the right hand side of these two equations. (In the first equation, we set  $t_{+} = 2$  because, surprisingly, B(x, 2, z) is easier to compute than B(x, 1, z).) One is easy:



Figure 2.12: The two marked graphs that give rise to one balanced signed graph.

For the other one, we count balanced signed graphs by relating them with **marked graphs**, which are simple graphs with a sign + or – on each **vertex**. [HK81] A marked graph M gives rise to a balanced signed graph G by assigning to each edge the product of its vertex labels. Furthermore, if G has c components, then it arises from precisely  $2^c$  different marked graphs, obtained from M by choosing some connected components and changing their signs. This correspondence is illustrated in Figure 2.12. It follows that  $B(x, 2y, z) = \sum_{B \text{ balanced }} 2^c b(v, c, e) \frac{x^v}{v!} y^c z^e = \sum_{M \text{ marked }} m(v, c, e) \frac{x^v}{v!} y^c z^e$  is the generating function for marked graphs, and hence B(x, 2, z) may be computed easily:

$$B(x, 2, z) = \sum_{e, v} {\binom{v}{2}}_{e} 2^{v} \frac{x^{v}}{v!} z^{e} = F(2x, 1+z)$$

Putting these equations together yields

$$S(x, y_+, y_-, z) = F(2x, 1+z)^{(y_+-y_-)/2} F(x, (1+z)^2)^{y_-}.$$

## 2.4 Nice families of generating functions

#### 2.4.1 Rational generating functions

Many sequences in combinatorics and other fields satisfy three equivalent properties: they satisfy a recursive formula with constant coefficients, they are given by an explicit formula in terms of polynomials and exponentials, and their generating functions are rational. We understand these sequences very well. The following theorem tells us how to translate any one of these formulas into the others.

**Theorem 2.4.1.** [Sta12, Theorem 4.1.1] Let  $a_0, a_1, a_2, \ldots$  be a sequence of complex numbers and let  $A(x) = \sum_{n\geq 0} a_n x^n$  be its ordinary generating function. Let  $q(x) = 1 + c_1 x + \cdots + c_d x^d = (1 - r_1 x)^{d_1} \cdots (1 - r_k x)^{d_k}$  be a complex polynomial of degree d. The following are equivalent:

1. The sequence satisfies the linear recurrence with constant coefficients

$$a_n + c_1 a_{n-1} + \dots + c_d a_{n-d} = 0$$
  $(n \ge d)$ 

2. There exist polynomials  $f_1(x), \ldots, f_k(x)$  with deg  $f_i(x) < d_i$  for  $1 \le i \le n$  such that

$$a_n = f_1(n) r_1^n + \dots + f_k(n) r_k^n.$$

3. There exists a polynomial p(x) with deg p(x) < d such that A(x) = p(x)/q(x).

Notice that Theorem 2.4.1.2 gives us the asymptotic growth of  $a_n$  immediately. Let us provide more explicit recipes.

 $(1 \Rightarrow 2)$  Extract the inverses  $r_i$  of the roots of  $q(x) = 1 + c_1 x + \dots + c_d x^d$  and their multiplicities  $d_i$ . The  $d_1 + \dots + d_k = d$  coefficients of the  $f_i$ s are the unknowns in the system of d linear equations  $a_n = f_1(n) r_1^n + \dots + f_k(n) r_k^n$   $(n = 0, 1, \dots, d - 1)$ , which has a unique solution.

 $(1 \Rightarrow 3)$  Read off  $q(x) = 1 + c_1 x + \dots + c_d x^d$  from the recurrence; the coefficients of p(x) are  $[x^k]p(x) = a_k + c_1 a_{k-1} + \dots + c_d a_{k-d}$  for  $0 \le k < d$ , where  $a_i = 0$  for i < 0.

 $(2 \Rightarrow 1)$  Compute the  $c_i$ s using  $q(x) = \prod_i^k (1 - r_i x)^{\deg f_i + 1}$ .

 $(2 \Rightarrow 3)$  Let  $q(x) = \prod_i (1 - r_i x)^{\deg f_i + 1}$ , and compute the first k terms of p(x) = A(x)q(x); the others are 0.

 $(3 \Rightarrow 1)$  Extract the  $c_i$ s from the denominator q(x).

 $(3 \Rightarrow 2)$  Compute the partial fraction decomposition  $p(x)/q(x) = \sum_{i=1}^{k} p_i(x)/(1 - r_i x)^{d_i}$  where  $\deg p_i(x) < d_i$  and use  $(1 - r_i x)^{-d_i} = \sum_n {\binom{d_i + n - 1}{d_i - 1}} r_i^n x^n$  to extract  $a_n = [x^n]p(x)/q(x)$ .

**Characterizing polynomials.** As a special case of Theorem 2.4.1, we obtain a useful characterization of sequences given by a polynomial. The **difference operator**  $\Delta$  acts on sequences, sending the sequence  $\{a_n : n \in \mathbb{N}\}$  to the sequence  $\{\Delta a_n : n \in \mathbb{N}\}$  where  $\Delta a_n = a_{n+1} - a_n$ .

**Theorem 2.4.2.** [Sta12, Theorem 4.1.1] Let  $a_0, a_1, a_2, \ldots$  be a sequence of complex numbers and let  $A(x) = \sum_{n\geq 0} a_n x^n$  be its ordinary generating function. Let d be a positive integer. The following are equivalent:

- 1. We have  $\Delta^{d+1}a_n = 0$  for all  $n \in \mathbb{N}$ .
- 2. There exists a polynomial f(x) with deg  $f \leq d$  such that  $a_n = f(n)$  for all  $n \in \mathbb{N}$ .
- 3. There exists a polynomial p(x) with deg  $p(x) \le d$  such that  $A(x) = p(x)/(1-x)^{d+1}$ .

We have already seen some combinatorial polynomials and generating functions whose denominator is a power of 1 - x; we will see many more examples in the following sections.

## 2.4.2 Algebraic and D-finite generating functions

When the generating function  $A(x) = \sum_{n} a_n x^n$  we are studying is not rational, the next natural question to ask is whether A(x) is algebraic. If it is, then just as in the rational case, the sequence  $a_n$  still satisfies a linear recurrence, although now the coefficients are polynomial in n. This general phenomenon is best explained by introducing the wider family of "D-finite" (also known as "differentially finite" or "holonomic") power series. Let us discuss a quick example before we proceed to the general theory.

We saw that the ordinary generating function for the Motzkin numbers satisfies the quadratic equation

$$x^2M^2 + (x-1)M + 1 = 0 (8)$$

which gives rise to the quadratic recurrence  $M_n = M_{n-1} + \sum_i M_i M_{n-2-i}$  with  $M_0 = 1$ . This is not a bad recurrence, but we can find a better one. Differentiating (8) we can express M' in terms of M. Our likely first attempt leads us to  $M' = -(2xM^2 + M)/(2x^2M + x - 1)$ , which is not terribly enlightening. However, using (8) and a bit of purposeful algebraic manipulation, we can rewrite this as a linear equation with polynomial coefficients:

$$(x - 2x^{2} - 3x^{3})M' + (2 - 3x - 3x^{2})M - 2 = 0.$$

Extracting the coefficient of  $x^n$  we obtain the much more efficient recurrence relation

$$(n+2)M_n - (2n+1)M_{n-1} - (3n-3)M_{n-2} = 0. \qquad (n \ge 2)$$

We now explain the theoretical framework behind this example.

**Rational, algebraic, and D-finite series**. Consider a formal power series A(x) over the complex numbers. We make the following definitions.

A(x) is <b>rational</b>	There exist polynomials $p(x)$ and $q(x) \neq 0$ such that						
	q(x)A(x) = p(x).						
A(x) is algebraic	There exist polynomials $p_0(x), \ldots, p_d(x)$ such that						
	$p_0(x) + p_1(x)A(x) + p_2(x)A(x)^2 + \dots + p_d(x)A(x)^d = 0.$						
A(x) is <b>D-finite</b>	There exist polynomials $q_0(x), \ldots, q_d(x), q(x)$ such that						
	$q_0(x)A(x) + q_1(x)A'(x) + q_2(x)A''(x) + \dots + q_d(x)A^{(d)}(x) = q(x).$						

$\{a_0, a_1, \ldots\}$ is <b>c-recursive</b>	There are constants $c_0, \ldots, c_d \in \mathbb{C}$ such that				
	$c_0 a_n + c_1 a_{n-1} + \dots + c_d a_{n-d} = 0$ $(n \ge d)$				
$\{a_0, a_1, \ldots\}$ is <b>P-recursive</b>	There are complex polynomials $c_0(x), \ldots, c_d(x)$ such that				
	$c_0(n)a_n + c_1(n)a_{n-1} + \dots + c_d(n)a_{n-d} = 0$ $(n \ge d)$				

Now consider the corresponding sequence  $a_0, a_1, a_2 \dots$  and make the following definitions.

These families contain most (but certainly not all) series and sequences that we encounter in combinatorics. They are related as follows.

**Theorem 2.4.3.** Let  $A(x) = a_0 + a_1x + a_2x^2 + \cdots$  be a formal power series. The following implications hold.

$$\begin{array}{cccc} A(x) \text{ is rational} & \implies & A(x) \text{ is algebraic} & \implies & A(x) \text{ is D-finite} \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & &$$

*Proof.* We already discussed the correspondence between rational series and c-recursive functions, and rational series are trivially algebraic. Let us prove the remaining statements.

 $\{a_0$ 

 $(Algebraic \Rightarrow D\text{-finite})$  Suppose A(x) satisfies an algebraic equation of degree d. Then A is algebraic over the field  $\mathbb{C}(x)$ , and the field extension  $\mathbb{C}(x, A)$  is a vector space over  $\mathbb{C}(x)$  having dimension at most d.

Taking the derivative of the polynomial equation satisfied by A, we get an expression for A' as a rational function of A and x. Taking derivatives repeatedly, we find that all derivatives of A are in  $\mathbb{C}(x, A)$ . It follows that  $1, A, A', A'', \ldots, A^{(d)}$  are linearly dependent over  $\mathbb{C}(x)$ , and a linear relation between them is a certificate for the D-finiteness of A.

(*P*-recursive  $\Leftrightarrow$  *D*-finite): If  $q_0(x)A(x) + q_1(x)A'(x) + \cdots + q_d(x)A^{(d)}(x) = q(x)$ , comparing the coefficients of  $x^n$  gives a *P*-recursion for the  $a_i$ s. In the other direction, given a *P*-recursion for the  $a_i$ s of the form  $c_0(n)a_n + \cdots + c_d(n)a_{n-d} = 0$ , it is easy to obtain the corresponding differential equation after writing  $c_i(x)$  in terms of the basis  $\{(x+i)_k : k \in \mathbb{N}\}$  of  $\mathbb{C}[x]$ , where  $(y)_k = y(y-1)\cdots(y-k+1)$ .

The converses are not true. For instance,  $\sqrt{1+x}$  is algebraic but not rational, and  $e^x$  and  $\log(1-x)$  are D-finite but not algebraic.

**Corollary 2.4.4.** The ordinary generating function  $\sum_{n} a_n x^n$  is D-finite if and only if the exponential generating function  $\sum_{n} a_n \frac{x^n}{n!}$  is D-finite.

*Proof.* This follows from the observation that  $\{a_n : n \in \mathbb{N}\}$  is P-recursive if and only if  $\{a_n/n! : n \in \mathbb{N}\}$  is P-recursive

**A** few examples. Before we discuss general tools, we collect some examples. We will prove all of the following statements later in this section.

The power series for subsets, Fibonacci numbers, and Stirling numbers are rational:

$$\sum_{n \ge 0} 2^n x^n = \frac{1}{1 - 2x}, \quad \sum_{n \ge 0} F_n x^n = \frac{x}{1 - x - x^2}, \quad \sum_{n \ge k} S(n, k) x^n = \frac{x}{1 - x} \cdot \frac{x}{1 - 2x} \cdots \frac{x}{1 - kx}$$

The "diagonal binomial", k-Catalan, and Motzkin series are algebraic but not rational:

$$\sum_{n\geq 0} \binom{2n}{n} x^n = \frac{1}{\sqrt{1-4x}}, \quad \sum_{n\geq 0} M_n x^n = \frac{1-x-\sqrt{1-2x-3x^2}}{2x^2}, \quad \sum_{n\geq 0} \frac{1}{(k-1)n+1} \binom{kn}{n} x^n.$$

The following series are D-finite but not algebraic:

$$e^x$$
,  $\log(1+x)$ ,  $\sin x$ ,  $\cos x$ ,  $\arctan x$ ,  $\sum_{n\geq 0} {\binom{2n}{n}}^2 x^n$ ,  $\sum_{n\geq 0} {\binom{3n}{n,n}} x^n$ 

The following series are not D-finite:

$$\sqrt{1 + \log(1 + x^2)}$$
, sec  $x$ ,  $\tan x$ ,  $\sum_{n \ge 0} p(n)x^n = \prod_{k \ge 0} \frac{1}{1 - x^k}$ .

**Recognizing algebraic and D-finite series**. It is not always obvious whether a given power series is algebraic or D-finite, but there are some tools available. Fortunately, algebraic functions behave well under a few operations, and D-finite functions behave even better. This explains why these families contain most examples arising in combinatorics.

The following table summarizes the properties of formal power series that are preserved under various key operations. For example, the fifth entry on the bottom row says that if A(x) and B(x)are D-finite, then the composition A(B(x)) is not necessarily D-finite.

	cA	A+B	AB	1/A	$A \circ B$	$A \star B$	A'	$\int A$	$A^{\langle -1 \rangle}$
rational	Y	Y	Y	Y	Y	Y	Y	N	N
algebraic	Y	Y	Y	Y	Y	N	Y	N	Y
D-finite	Y	Y	Y	N	Ν	Y	Y	Y	N

Here  $A \star B(x) := \sum_{n \ge 0} a_n b_n x^n$  denotes the **Hadamard product** of A(x) and B(x),  $\int A(x) := \sum_{n \ge 0} a_n b_n x^n$  $\sum_{n\geq 1} \frac{a_{n-1}}{n} x^n$  is the **formal integral** of A(x), and  $A^{\langle -1 \rangle}(x)$  is the **compositional inverse** of A(x).

In the fourth column we are assuming that  $A(0) \neq 0$  so that 1/A(x) is well-defined, in the fifth column we are assuming that B(0) = 0 so that A(B(x)) is well-defined, and in the last column we are assuming that A(0) = 0 and  $A'(0) \neq 0$ , so that  $A^{\langle -1 \rangle}(x)$  is well-defined.

For proofs of the "Yes" entries, see [Sta80a], [Sta99], [FS09]. For the "No" entries, we momentarily assume the statements of the previous subsection. Then we have the following counterexamples:

- $\cos x$  is D-finite but  $1/\cos x = \sec x$  is not.
- $\sqrt{1+x}$  and  $\log(1+x^2)$  are D-finite but their composition  $\sqrt{1+\log(1+x^2)}$  is not.  $A(x) = \sum_{n\geq 0} {\binom{2n}{n}} x^n$  is algebraic but  $A \star A(x) = \sum_{n\geq 0} {\binom{2n}{n}}^2 x^n$  is not.

- 1 + x is rational and algebraic but its integral  $\log(1 + x)$  is neither.
- $x + x^2$  is rational but its compositional inverse  $(-1 + \sqrt{1 4x})/2$  is not.
- $\arctan x$  is D-finite but its compositional inverse  $\tan x$  is not.

Some of these negative results have weaker positive counterparts:

- If A(x) is algebraic and B(x) is rational, then  $A(x) \star B(x)$  is algebraic.
- If A(x) is D-finite and  $A(0) \neq 0$ , 1/A(x) is D-finite if and only if A'(x)/A(x) is algebraic.
- If A(x) is D-finite and B(x) is algebraic with B(0) = 0, then A(B(x)) is D-finite.

See [Sta99, Proposition 6.1.11], [HS85], and [Sta99, Theorem 6.4.10] for the respective proofs.

The following result is also useful:

**Theorem 2.4.5.** Consider a multivariate formal power series and its diagonal:

$$F(x_1, \dots, x_d) = \sum_{n_1, \dots, n_d \ge 0} a_{n_1, \dots, n_d} x_1^{n_1} \cdots x_d^{n_d}, \qquad diag \ F(x) = \sum_{n \ge 0} a_{n, \dots, n} x^n.$$

- 1. If  $F(x_1, \ldots, x_d)$  is a rational function of  $x_1, \ldots, x_d$ , then diag F(x) is D-finite.
- 2. If d = 2 and  $F(x_1, \ldots, x_d)$  is rational, then diag F(x) is algebraic, but not necessarily D-finite.

Now we are ready to prove our positive claims about the series at the beginning of this section. The first three expressions are visibly rational, and the diagonal binomial and Motzkin series are visibly algebraic. We proved that the k-Catalan series is algebraic in Section 2.3.2. The functions  $e^x$ ,  $\log(1 + x)$ ,  $\sin x$ ,  $\cos x$ ,  $\arctan x$  satisfy the differential equations y' = y, (1 + x)y' = 1, y'' = -y, y'' = -y,  $(1 + x^2)y' = 1$ , respectively. The series  $\sum_{n\geq 0} {\binom{2n}{n}}^2 x^n$  is the Hadamard product of  $(1 - 4x)^{-1/2}$  with itself, and hence D-finite. Finally,  $\sum_{n\geq 0} {\binom{3n}{n,n}} x^n$  is the diagonal of the rational function  $\frac{1}{1-x-y-z} = \sum_{a,b,c\geq 0} {\binom{a+b+c}{a,b,c}} x^a y^b z^c$ , and hence D-finite.

Proving the negative claims requires more effort and, often, a bit of analytic machinery. We briefly outline some key results.

**Recognizing series that are not algebraic**. There are a few methods available to prove that a series is **not** algebraic. The simplest algebraic and analytic criteria are the following.

**Theorem 2.4.6** (Eisenstein's Theorem). [PS98] If a series  $A(x) = \sum_{n\geq 0} a_n x^n$  with rational coefficients is algebraic, then there exists a positive integer m such that  $a_n m^n$  is an integer for all n.

This shows that  $e^x$ ,  $\log(1+x)$ ,  $\sin x$ ,  $\cos x$ , and  $\arctan x$  are not algebraic.

**Theorem 2.4.7.** [Jun31] If the coefficients of an algebraic power series  $A(x) = \sum_{n\geq 0} a_n x^n$  satisfy  $a_n \sim cn^r \alpha^n$  for nonzero  $c, \alpha \in \mathbb{C}$  and r < 0, then  $r + \frac{1}{2}$  is an integer.

Stirling's approximation  $n! \sim \sqrt{2\pi n} (n/e)^n$  gives  $\binom{2n}{n}^2 \sim c \cdot 16^n/n$  and  $\binom{3n}{n,n,n} \sim c \cdot 27^n/n$ , so the corresponding series are not algebraic.

Another useful analytic criterion is that an algebraic series A(x) must have a Newton-Puiseux expansion at any of its singularities. See [FS09, Theorem VII.7] and [FGS06] for details.

**Recognizing series that are not D-finite**. The most effective methods to show that a function is **not** D-finite are analytic.

**Theorem 2.4.8.** [Hen91, Theorem 9.1] Suppose that A(x) is analytic at x = 0, and it is D-finite, satisfying the equation  $q_0(x)A(x) + q_1(x)A'(x) + \cdots + q_d(x)A^{(d)}(x) = q(x)$  with  $q_d(x) \neq 0$ . Then A(x) can be extended to an analytic function in any simply connected region of the complex plane not containing the (finitely many) zeroes of  $q_d(x)$ .

Since sec x and  $\tan x$  have a pole at every odd multiple of  $\pi$ , they are not D-finite. Similarly,  $\sum_{n} p(n)x^{n} = \prod_{k=1}^{\infty} \frac{1}{1-x^{k}}$  is not D-finite because it has the circle |x| = 1 as a natural boundary of analyticity.

There are other powerful analytic criteria to prove a series is not D-finite. See [FGS06, Theorem VII.7] for details and further examples.

Sometimes it is possible to give *ad hoc* proofs that series are D-finite. For instance, consider  $y = \sqrt{1 + \log(1 + x^2)}$ . By induction, for any  $k \in \mathbb{N}$  there exist polynomials  $r_1(x), \ldots, r_k(x)$  such that  $y^{(k)} = r_1/y + r_2/y^3 + \cdots + r_k/y^{2k-1}$ . An equation of the form  $\sum_{i=0}^d q_i(x)y^{(i)} = q(x)$  would then give rise to a polynomial equation satisfied by y. This would also make  $y^2 - 1 = \log(1 + x^2)$  algebraic; but this contradicts Theorem 2.4.6.

# 3 Linear algebra methods

# 3.1 Determinants in combinatorics

There are several important theorems in enumerative combinatorics which express a combinatorial quantity in terms of a determinant. Of course, evaluating a determinant is not always straightforward, but there is a wide array of tools at our disposal.

The goal of Section 3.1 is to reduce many combinatorial problems to "just computing a determinant"; examples include walks in a graph, spanning trees, Eulerian cycles, matchings, and routings. In particular, we discuss the Transfer Matrix Method, which allows us to encode many combinatorial objects as walks in graphs, so that these linear algebraic tools apply. These problems lead us to many beautiful, mysterious, and highly non-trivial determinantal evaluations. We will postpone the proofs of the evaluations until Section 3.2, which is an exposition of some of the main techniques in the subtle science of computing combinatorial determinants.

#### 3.1.1 Preliminaries: Graph matrices.

An undirected graph, or simply a graph G = (V, E) consists of a set V of vertices and a set E of edges  $\{u, v\}$  where  $u, v \in V$  and  $u \neq v$ . In an undirected graph, we write uv for the edge  $\{u, v\}$ . A walk is a set of edges of the form  $v_1v_2, v_2v_3, \ldots, v_{k-1}v_k$ . This walk is closed if  $v_k = v_1$ .

A directed graph or digraph G = (V, E) consists of a set V of vertices and a set E of oriented edges (u, v) where  $u, v \in V$  and  $u \neq v$ . In an undirected graph, we write uv for the directed edge (u, v). A walk is a set of directed edges of the form  $v_1v_2, v_2v_3, \ldots, v_{k-1}v_k$ . This walk is closed if  $v_k = v_1$ .

We will see in this section that many graph theory problems can be solved using tools from linear algebra. There are several matrices associated to graphs which play a crucial role; we review them here.

**Directed graphs.** Let G = (V, E) be a directed graph.

• The adjacency matrix A = A(G) is the  $V \times V$  matrix whose entries are

 $a_{uv}$  = number of edges from u to v.

• The incidence matrix M = M(G) is the  $V \times E$  matrix with

$$m_{ve} = \begin{cases} 1 & \text{if } v \text{ is the final vertex of edge } e, \\ -1 & \text{if } v \text{ is the initial vertex of edge } e, \\ 0 & \text{otherwise.} \end{cases}$$

• The directed Laplacian matrix  $\vec{L} = \vec{L}(G)$  is the  $V \times V$  matrix whose entries are

$$\vec{l}_{uv} = \begin{cases} -(\text{number of edges from } u \text{ to } v) & \text{if } u \neq v, \\ \text{outdeg}(u) & \text{if } u = v. \end{cases}$$

**Undirected graphs.** Let G = (V, E) be an undirected graph.

• The (undirected) adjacency matrix A = A(G) is the  $V \times V$  matrix whose entries are

 $a_{uv}$  = number of edges connecting u and v.

This is the directed adjacency matrix of the directed graph on V containing edges  $u \to v$  and  $v \to u$  for every edge uv of G.

• The (undirected) Laplacian matrix L = L(G) is the  $V \times V$  matrix with entries

$$l_{uv} = \begin{cases} -(\text{number of edges connecting } u \text{ and } v) & \text{if } u \neq v \\ \deg u & \text{if } u = v \end{cases}$$

If M is the incidence matrix of any orientation of the edges of G, then  $L = MM^T$ .

#### 3.1.2 Counting walks: the Transfer Matrix Method.

Counting walks in a graph is a fundamental problem, which (often in disguise) includes many important enumerative problems. The Transfer Matrix Method addresses this problem by expressing the number of walks in a graph G in terms of its adjacency matrix A(G), and then uses linear algebra to count those walks.

**Directed or undirected graphs.** The Transfer Matrix Method is based on the following simple, powerful observation, which applies to directed and undirected graphs:

**Theorem 3.1.1.** Let G = (V, E) be a graph and let A = A(G) be the  $V \times V$  adjacency matrix of G, where  $a_{uv}$  is the number of edges from u to v. Then

$$(A^n)_{uv} =$$
 number of walks of length n in G from u to v.

*Proof.* Observe that

$$(A^{n})_{uv} = \sum_{w_1, \dots, w_{n-1} \in V} a_{uw_1} a_{w_1 w_2} \cdots a_{w_{n-1} v}$$

and there are  $a_{uw_1}a_{w_1w_2}\cdots a_{w_{n-1}v}$  walks visiting vertices  $u, w_1, \ldots, w_{n-1}, v$  in that order.

**Corollary 3.1.2.** The generating function  $\sum_{n\geq 0} (A^n)_{uv} x^n$  for walks of length n from u to v in G is a rational function.

*Proof.* Using Cramer's formula, we have

$$\sum_{n \ge 0} (A^n)_{uv} x^n = ((I - xA)^{-1})_{uv} = (-1)^{u+v} \frac{\det(I - xA : v, u)}{\det(I - xA)}$$

where (M: v, u) denotes the cofactor of M obtained by removing row v and column u.

**Corollary 3.1.3.** If  $C_G(n)$  is the number of closed walks of length n in G, then

$$C_G(n) = \lambda_1^n + \dots + \lambda_k^n, \qquad \sum_{n \ge 1} C_G(n) x^n = \frac{-x Q'(x)}{Q(x)}$$

where  $\lambda_1, \ldots, \lambda_k$  are the eigenvalues of the adjacency matrix A and  $Q(x) = \det(I - xA)$ .
*Proof.* Theorem 3.1.1 implies that  $C_G(n) = \operatorname{tr}(A^n) = \lambda_1^n + \cdots + \lambda_k^n$ . The second equation then follows from  $Q(x) = (1 - \lambda_1 x) \cdots (1 - \lambda_k x)$ .

In view of Theorem 3.1.1, we want to be able to compute powers of the adjacency matrix A. As we learn in linear algebra, this is very easy to do if we are able to diagonalize A. This is not always possible, but we can do it when A is undirected.

**Undirected graphs.** When our graph G is undirected, the adjacency matrix A(G) is symmetric, and hence diagonalizable.

**Theorem 3.1.4.** Let G = (V, E) be an **undirected** graph and let  $\lambda_1, \ldots, \lambda_k$  be the eigenvalues of the adjacency matrix A = A(G). Then for any vertices u and v there exist constants  $c_1, \ldots, c_k$  such that

number of walks of length n from u to  $v = c_1 \lambda_1^n + \dots + c_k \lambda_k^n$ .

*Proof.* The key fact is that a real symmetric  $k \times k$  matrix A has k real orthonormal eigenvectors  $q_1, \ldots, q_k$  with real eigenvalues  $\lambda_1, \ldots, \lambda_k$ . Equivalently, the  $k \times k$  matrix Q with columns  $q_1, \ldots, q_k$  is orthogonal (so  $Q^T = Q^{-1}$ ) and diagonalizes A:

$$Q^{-1}AQ = D = \operatorname{diag}(\lambda_1, \dots, \lambda_k)$$

where  $D = \operatorname{diag}(\lambda_1, \ldots, \lambda_k)$  is the diagonal matrix with diagonal entries  $\lambda_1, \ldots, \lambda_k$ . The result then follows from  $A^n = QD^nQ^{-1} = Q\operatorname{diag}(\lambda_1^n, \ldots, \lambda_k^n)Q^T$ , with  $c_t = q_{it}q_{jt}$ .

**Applications.** Many families of combinatorial objects can be enumerated by first recasting the objects as walks in a "transfer graph", and then applying the transfer matrix method. We illustrate this technique with a few examples.

1. (Colored necklaces) Let f(n, k) be the number of ways of coloring the beads of a necklace of length n with k colors so that no two adjacent beads have the same color. (Different rotations and reflections of a coloring are considered different.) There are several ways to compute this number, but a very efficient one is to notice that such a coloring is a graph walk in disguise. If we label the beads  $1, \ldots, n$  in clockwise order and let  $a_i$  be the color of the *i*th bead, then the coloring corresponds to the closed walk  $a_1, a_2, \ldots, a_n, a_1$  in the complete graph  $K_n$ . The adjacency graph of  $K_n$  is A = J - I where J is the matrix all of whose entries equal 1, and I is the identity. Since J has rank 1, it has n - 1 eigenvalues equal to 0. Since the trace is n, the last eigenvalue is n. It follows that the eigenvalues of A = J - I are  $-1, -1, \ldots, -1, n - 1$ . Then Corollary 3.1.3 tells us that

$$f(n,k) = (n-1)^k + (n-1)(-1)^k.$$

It is possible to give a bijective proof of this formula, but this algebraic proof is much simpler.

2. (Words with forbidden subwords, 1.) Let  $h_n$  be the number of words of length n in the alphabet  $\{a, b\}$  which do not contain aa as a consecutive subword. This is the same as a walk of length n-1 in the transfer graph with vertices a and b and edges  $a \to b$ ,  $b \to a$  and  $b \to b$ .

The absence of the edge  $a \rightarrow a$  guarantees that these walks produce only the valid words we wish to count. The adjacency matrix and its powers are

$$A = \begin{pmatrix} 0 & 1 \\ 1 & 1 \end{pmatrix}, \qquad A^n = \begin{pmatrix} F_{n-1} & F_n \\ F_n & F_{n+1} \end{pmatrix}$$

where the Fibonacci numbers  $F_0, F_1, \ldots$  are defined recursively by  $F_0 = 0, F_1 = 1$ , and  $F_k = F_{k-1} + F_{k-2}$  for  $k \ge 2$ .

Since  $h_n$  is the sum of the entries of  $A^{n-1}$ , we get that  $h_n = F_{n+2}$ , and  $g_n \sim c \cdot \alpha^n$  where  $\alpha = \frac{1}{2}(1 + \sqrt{5}) \approx 1.6179...$  is the golden ratio. Of course there are easier proofs of this fact, but this approach works for any problem of enumerating words in a given alphabet with given forbidden consecutive subwords. Let us study a slightly more intricate example, which should make it clear how to proceed in general.

3. (Words with forbidden subwords, 2.) Let  $g_n$  be the number of cyclic words of length n in the alphabet  $\{a, b\}$  which do not contain aa or abba as a consecutive subword. We wish to model these words as walks in a directed graph. At first this may seem impossible because, as we construct the word sequentially, the validity of a new letter depends on more than just the previous letter. However, a simple trick resolves this difficulty: we can introduce more memory into the vertices of the transfer graph. In this case, since the validity of a new letter depends on the previous three letters, we let the vertices of the transfer graph be aba, abb, bab, bba, bbb (the allowable "windows" of length 3) and put an edge  $wxy \rightarrow xyz$  in the graph if the window wxy is allowed to precede the window xyz; that is, if wxyz is an allowed subword. The result is the graph of Figure 3.1, whose adjacency matrix A satisfies

$$\det(I - xA) = \det \begin{pmatrix} 1 & 0 & -x & 0 & 0\\ 0 & 1 & 0 & 0 & -x\\ -x & -x & 1 & 0 & 0\\ 0 & 0 & -x & 1 & 0\\ 0 & 0 & 0 & -x & 1 - x \end{pmatrix} = -x^4 + x^3 - x^2 - x + 1.$$



Figure 3.1: The transfer graph for words on the alphabet  $\{a, b\}$  avoiding aa and abba as consecutive subwords.

The valid cyclic words of length n correspond to the closed walks of length n in the transfer graph, so Corollary 3.1.3 tells us that the generating function for  $g_n$  is

$$\sum_{n\geq 0} g_n x^n = \frac{x+2x^2-3x^3+4x^4}{1-x-x^2+x^3-x^4} = x+3x^2+x^3+7x^4+6x^5+15x^6+15x^7+31x^8+37x^9+\cdots$$

Theorem 2.4.1.2 then tells us that  $g_n \approx c \cdot \alpha^n$  where  $\alpha \approx 1.5129$  is the inverse of the smallest positive root of  $1 - x - x^2 + x^3 - x^4 = 0$ . The values of  $g_1, g_2, g_3$  may surprise us. Note that the generating function does something counterintuitive: it does not count the words *a* (because *aa* is forbidden), *aba* (because *aa* is forbidden), or *abb* (because *abba* is forbidden).

This example serves as a word of caution: when we use the transfer-matrix method to enumerate "cyclic" objects using Corollary 3.1.3, the initial values of the generating function may not be the ones we expect. In a particular problem of interest, it will be straightforward to adjust those values accordingly.

To illustrate the wide applicability of this method, we conclude this section with a problem where the transfer graph is less apparent.

4. (Monomer-dimer problem) An important open problem in statistical mechanics is the **monomerdimer problem** of computing the number of tilings T(m,n) of an  $m \times n$  rectangle into dominoes (2 × 1 rectangles) and unit squares. Equivalently, T(m,n) is the number of partial matchings of an  $m \times n$  grid, where each node is matched to at most one of its neighbors.

There is experimental evidence, but no proof, that  $T(n,n) \sim c \cdot \alpha^{1/n^2}$  where  $\alpha \approx 1.9402...$  is a constant for which no exact expression is known. The transfer-matrix method is able to solve this problem for any fixed value of m, proving that the generating function  $\sum_{n\geq 0} T(m,n)x^n$  is rational. We carry this out for m = 3.

Let t(n) be the number of tilings of a  $3 \times n$  rectangle into dominoes and unit squares. As with words, we can build our tilings sequentially from left to right by covering the first column, then the second column, and so on. The tiles that we can place on a new column depend only on the tiles from the previous column that are sticking out, and this can be modeled by a transfer graph.



Figure 3.2: A tiling of a  $3 \times 5$  rectangle into dominoes and unit squares.

More specifically, let T be a tiling of a  $3 \times n$  rectangle. We define n + 1 triples  $v_0, \ldots, v_n$  which record how T interacts with the n + 1 vertical grid lines of the rectangle. The *i*th grid

line consists of three unit segments, and each coordinates of  $v_i$  is 0 or 1 depending on whether these three segments are edges of the tiling or not. For example, Figure 3.2 corresponds to the triples 111, 110, 011, 101, 010, 111.

The choice of  $v_i$  is restricted only by  $v_{i-1}$ . The only restriction is that  $v_{i-1}$  and  $v_i$  cannot both have a 0 in the same position, because this would force us to put two overlapping horizontal dominoes in T. These compatibility conditions are recorded in the transfer graph of Figure 3.3. When  $v_{i-1} = v_i = 111$ , there are 3 ways of covering column *i*. If  $v_{i-1}$  and  $v_i$  share two 1s in consecutive positions, there are 2 ways. In all other cases, there is a unique way. It follows that the tilings of a  $3 \times n$  rectangle are in bijection with the walks of length *n* from 111 to 111 in the transfer graph.



Figure 3.3: The transfer graph for tilings of  $3 \times n$  rectangles into dominoes and unit squares.

Since the adjacency matrix is

		000	001	110	010	101	001	110	111
A =	000	/ 0	0	0	0	0	0	0	1
	001	0	0	1	0	0	0	0	1
	110	0	1	0	0	1	0	1	2
	010	0	0	0	0	1	0	0	1
	101	0	0	1	1	0	0	1	1
	001	0	0	0	0	0	0	1	1
	110	0	0	1	0	1	1	0	2
	111	$\setminus 1$	1	2	1	1	1	2	3 /

Theorem 3.1.1 tells us that

$$\sum_{n\geq 0} t(n)x^n = \frac{\det(I - xA : 111, 111)}{\det(I - xA)} = \frac{(1 + x - x^2)(1 - 2x - x^2)}{(1 + x)(1 - 5x - 9x^2 + 9x^3 + x^4 - x^5)}$$
$$= 1 + 3x + 22x^2 + 131x^3 + 823x^4 + 5096x^5 + 31687x^6 + 196785x^7 + \cdots$$

By Theorem 2.4.1.2,  $t_n \sim c \cdot \alpha^n$  where  $\alpha \approx 6.21207...$  is the inverse of the smallest positive root of the denominator  $(1+x)(1-5x-9x^2+9x^3+x^4-x^5)$ .

#### 3.1.3 Counting spanning trees: the Matrix-Tree theorem.

In this section we discuss two results: Kirkhoff's determinantal formula for the number of spanning trees of a graph, and Tutte's generalization to oriented spanning trees of directed graphs.

**Undirected Matrix-Tree theorem.** Let G = (V, E) be a connected graph with no loops. A **spanning tree** T of G is a collection of edges such that for any two vertices u and v, T contains a unique path between u and v. If G has n vertices, then:

- T contains no cycles,
- T spans G; that is, there is a path from u to v in T for any vertices  $u \neq v$ , and
- T has n-1 edges.

Furthermore, any two of these properties imply that T is a spanning tree. Our goal in this section is to compute the number c(G) of spanning trees of G.

Orient the edges of G arbitrarily. Recall that the incidence matrix M of G is the  $V \times E$  matrix whose eth column is  $\mathbf{e}_u - \mathbf{e}_v$  if  $e = u \rightarrow v$ , and the Laplacian  $L = MM^T$  has entries

$$l_{uv} = \begin{cases} -(\text{number of edges connecting } u \text{ and } v) & \text{if } u \neq v \\ \deg u & \text{if } u = v \end{cases}$$

Note that L(G) is singular because all its row sums are 0. A **principal cofactor**  $L_v(G)$  is obtained from L(G) by removing the vth row and vth column for some vertex v.

**Theorem 3.1.5.** (Kirkhoff's Matrix-Tree Theorem) The number c(G) of spanning trees of a connected graph G is

$$c(G) = \det L_v(G) = \frac{1}{n}\lambda_1 \cdots \lambda_{n-1}.$$

where  $L_v(G)$  is any principal cofactor of the Laplacian L(G), and  $\lambda_1, \ldots, \lambda_{n-1}, \lambda_n = 0$  are the eigenvalues of L(G).

*Proof.* We use the Binet-Cauchy formula, which states that if A and B are  $m \times n$  and  $n \times m$  matrices, respectively, with m < n, then

$$\det AB = \sum_{S \subseteq [n] : |S| = m} \det A[S] \det B[S]$$

where A[S] (resp. B[S]) is the  $n \times n$  matrix obtained by considering only the columns of A (resp. the rows of B) indexed by S.

We also use the following observation: if  $M_v$  is the "reduced" adjacency matrix M with the vth row removed, and S is a set of n-1 edges of E, then

$$\det M_v[S] = \begin{cases} \pm 1 & \text{if } S \text{ is a spanning tree,} \\ 0 & \text{otherwise.} \end{cases}$$

This observation is easily proved: if S is not a spanning tree, then it contains a cycle C, which gives a linear dependence among the columns indexed by the edges of C. Otherwise, if S is a spanning tree, think of v as its root, and "prune" it by repeatedly removing a leaf  $v_i \neq v$  and its only incident edge  $e_i$  for  $1 \leq i \leq n-1$ . Then if we list the rows and columns of M[S] in the orders  $v_1, \ldots, v_{n-1}$ and  $e_1, \ldots, e_{n-1}$ , respectively, the matrix will be lower triangular with 1s and -1s in the diagonal. Combining these two equations, we obtain the first statement:

$$\det L_v(G) = \sum_{S \subseteq [n] : |S| = m} \det M[S] \det M^T[S] = \sum_{S \subseteq [n] : |S| = m} \det M[S]^2 = c(G).$$

To prove the second one, observe that the coefficient of  $-x^1$  in the characteristic polynomial  $\det(L - xI) = (\lambda_1 - x) \cdots (\lambda_{n-1} - x)(0 - x)$  is the sum of the *n* principal cofactors, which are all equal to c(G).

The matrix-tree theorem is a very powerful tool for computing the number of spanning trees of a graph. Let us state a few examples.

The complete graph  $K_n$  has n vertices and an edge joining each pair of vertices. The complete bipartite graph  $K_{m,n}$  has m "top" vertices and n "bottom" vertices, and mn edges joining each top vertex to each bottom vertex. The hyperoctahedral graph  $\Diamond_n$  has vertices  $\{1, 1', 2, 2', \ldots, n, n'\}$  and its only missing edges are ii' for  $1 \le i \le n$ . The *n*-cube graph  $C_n$  has vertices  $(\epsilon_1, \ldots, \epsilon_n)$  where  $\epsilon_i \in \{0, 1\}$ , and an edge connecting any two vertices that differ in exactly one coordinate. The *n*-dimensional grid of size m, denoted  $mC_n$ , has vertices  $(\epsilon_1, \ldots, \epsilon_n)$  where  $\epsilon_i \in \{1, \ldots, m\}$ , and an edge connecting any two vertices that differ in exactly one coordinate i, where they differ by 1.

**Theorem 3.1.6.** The number of spanning trees of some interesting graphs are as follows.

- 1. (Complete graph):  $c(K_n) = n^{n-2}$
- 2. (Complete bipartite graph):  $c(K_{m,n}) = m^{n-1}n^{m-1}$
- 3. (Hyperoctahedral graph):  $c(\Diamond_n) = 2^{2n-2}(n-1)^n n^{n-2}$ .
- 4. (n-cube):  $c(C_n) = 2^{2^n n 1} \prod_{k=1}^n k^{\binom{n}{k}}$
- 5. (*n*-dimensional grid of size m):  $c(mC_n) = m^{m^n n 1} \prod_{k=1}^n k^{\binom{n}{k}(m-1)^k}$

We will see proofs of the first and third example in Section 3.2. For the others, and many additional examples, see [CDS95].

**Directed Matrix-Tree theorem.** Now let G = (V, E) be a **directed** graph containing no loops. An **oriented spanning tree rooted at** v is a collection of edges T such that for any vertex u there is a unique path from u to v. The underlying unoriented graph  $\underline{T}$  is a spanning tree of the unoriented graph  $\underline{G}$ . Let c(G, v) be the number of spanning trees rooted at G.

Recall that the **directed Laplacian matrix**  $\vec{L}$  has entries

$$\vec{l}_{uv} = \begin{cases} -(\text{number of edges from } u \text{ to } v) & \text{if } u \neq v \\ \text{outdeg } u & \text{if } u = v \end{cases}$$

Now the matrix  $\overrightarrow{L}(G)$  is not necessarily symmetric, but it is still singular.

**Theorem 3.1.7.** (Tutte's Directed Matrix-Tree Theorem) Let G be a directed graph and v be a vertex. The number c(G, v) of oriented spanning trees rooted at v

$$c(G, v) = \det \overrightarrow{L}_v(G)$$

where  $\overrightarrow{L}_{v}(G)$  is obtained from L(G) by removing the vth row and column. Furthermore, if G is **balanced**, so indeg v =outdeg v for all vertices v, then

$$c(G, v) = \frac{1}{n}\lambda_1 \cdots \lambda_{n-1}$$

where  $\lambda_1, \ldots, \lambda_{n-1}, \lambda_n = 0$  are the eigenvalues of L(G).

*Proof.* Proceed by induction. Consider a vertex  $w \neq v$  and an edge e starting at w. Let G' = G - e be obtained from G by removing e. If e is the only edge starting at w, then every spanning tree must use it, and we have

$$c(G, v) = c(G', v) = \det \overrightarrow{L}_v(G') = \det \overrightarrow{L}_v(G).$$

Otherwise, let G'' be obtained from G by removing all edges starting at w other than e. There are c(G', v) oriented spanning trees rooted at v which do not contain e, and c(G'', v) which do contain e, so we have

$$c(G,v) = c(G',v) + c(G'',v) = \det \overrightarrow{L}_v(G') + \det \overrightarrow{L}_v(G'') = \det \overrightarrow{L}_v(G)$$

where the last equality holds since determinants are multilinear.

We postpone the proof of the second statement to the next section, where it will be an immediate consequence of Theorem 3.1.9.  $\hfill \Box$ 

#### 3.1.4 Counting Eulerian cycles: the BEST theorem.

One of the earliest combinatorial questions is the problem of the Seven Bridges of Königsberg. In the early 1700s, the Prussian city of Königsberg was separated by the Pregel river into four regions, connected to each other by seven bridges. In the map of Figure 3.4 we have labeled the regions N, S, E, and I; there are two bridges between N and I, two between S and I, and three bridges connecting E to each of N, I, and S. The problem was to find a walk through the city that crossed each bridge exactly once. Euler proved in 1735 that it was impossible to find such a walk; this is considered to be the first paper in graph theory.

Euler's argument is simple, and relies on the fact that every region of Königsberg is adjacent to an odd number of bridges. Suppose there existed such a walk, starting at region A and ending at region B. Now consider a region C other than A and B. Then our path would enter and leave Cthe same number of times; but then it would not use all the bridges adjacent to C, because there is an odd number of such bridges.

In modern terminology, each region of the city is represented by a vertex, and each bridge is represented by an edge connecting two vertices. We will be more interested in the directed case, where every edge has an assigned direction. An **Eulerian path** is a path in the graph which visits every edge exactly once. If the path starts and ends at the same vertex, then it is called a **Eulerian cycle**. We say G is an **Eulerian graph** if it has an Eulerian cycle.



Figure 3.4: The seven bridges of Königsberg.

**Theorem 3.1.8.** A directed graph is Eulerian if and only if it is connected and every vertex v satisfies indeg(v) = outdeg(v).

*Proof.* If a graph has an Eulerian cycle C, then C enters and leaves each vertex v the same number of times. Therefore indeg(v) = outdeg(v).

To prove the converse, let us start by arbitrarily "walking around G until we get stuck". More specifically, we start at any vertex  $v_0$ , and at each step, we exit the current vertex by walking along any outgoing edge which we have not used yet. If there is no available outgoing edge, we stop.

Whenever we enter a vertex  $v \neq v_0$ , we will also be able to exit it since  $\operatorname{indeg}(v) = \operatorname{outdeg}(v)$ ; so the walk can only get stuck at  $v_0$ . Hence the resulting walk C is a cycle. If C uses all edges of the graph, we are done. If not, then since G is connected we can find a vertex v' of C with an unused outgoing edge, and we use this edge to start walking around the graph G - C until we get stuck, necessarily at v'. The result will be a cycle C'. Starting at v' we can traverse C and then C', thus obtaining a cycle  $C \cup C'$  which is longer than C. Repeating this procedure, we will eventually construct an Eulerian cycle.

There is a remarkable formula for the number of Eulerian cycles, due to de Bruijn, van Ardenne-Ehrenfest, Smith, and Tutte:

**Theorem 3.1.9.** (BEST Theorem) If G is an Eulerian directed graph, then the number of Eulerian cycles of G is

$$c(G,v) \cdot \prod_{w \in V} (\mathit{outdeg}(w) - 1)!$$

for any vertex v, where c(G, v) is the number of oriented spanning trees rooted at v.

*Proof.* We fix an edge e starting at v, and let each Eulerian cycle start at e. For each vertex w let  $E_w$  be the set of outgoing edges from w.

Consider an Eulerian cycle C. For each vertex  $w \neq v$ , let  $e_w$  be the last outgoing edge from w that C visits, and let  $\pi_w$  (resp.  $\pi_v$ ) be the ordered set  $E_w - e_w$  (resp.  $E_v - e$ ) of the other outgoing edges from w (resp. v), listed in the order that C traverses them. It is easy to see that  $T = \{e_w : w \neq v\}$  is an oriented spanning tree rooted at v.

Conversely, an oriented tree T and permutations  $\{\pi_w : w \in V\}$  serve as directions to tour G. We start with edge e. Each time we arrive at vertex w, we exit it by using the first unused edge according to  $\pi_w$ . If we have used all the edges  $E_w - e_w$  of  $\pi_w$ , then we use  $e_w \in T$ . It is not hard to check that this is a bijection. This completes the proof.

**Corollary 3.1.10.** In an Eulerian directed graph, the number of oriented spanning trees rooted at v is the same for all vertices v; it equals

$$c(G,v) = \frac{1}{n}\lambda_1\cdots\lambda_{n-1}$$

where  $\lambda_1, \ldots, \lambda_{n-1}, \lambda_n = 0$  are the eigenvalues of  $\overrightarrow{L}(G)$ .

*Proof.* The BEST theorem implies that c(G, v) is independent of v, and then the argument in the proof of Theorem 3.1.5 applies to give the desired formula.

The BEST theorem can be used beautifully to enumerate a very classical, and highly nontrivial, family of objects. A k-ary **de Bruijn sequence** of order n is a cyclic word W of length  $k^n$  in the alphabet  $\{1, \ldots, k\}$  such that the  $k^n$  consecutive subwords of W of length n are the  $k^n$  distinct words of length n. For example, the 2-ary deBruijn sequences of order 3 are 11121222 and 22212111; these "memory wheels" were described in Sanskrit poetry several centuries ago [Kak00]. Their existence and enumeration was proved by Flye Saint-Marie in 1894 for k = 2 and by van Aardenne-Ehrenfest and de Bruijn in 1951 in general.

**Theorem 3.1.11.** [FSM94, dB46] The number of k-ary de Bruijn sequences of order n is  $(k!)^{k^{n-1}}/k^n$ .

*Proof.* Consider the **de Bruijn graph** whose vertices are the  $k^{n-1}$  sequences of length n-1 in the alphabet  $\{1, \ldots, k\}$ , and where there is an edge from  $a_1a_2 \ldots a_{n-1}$  to the word  $a_2a_3 \ldots a_n$  for all  $a_1, \ldots, a_n$ . It is natural to label this edge  $a_1a_2 \ldots a_n$ . It then becomes apparent that k-ary de Bruijn sequence are in bijection with the Eulerian cycles of the de Bruijn graph. Since indeg(v) = outdeg(v) = k for all vertices v, this graph is indeed Eulerian, and we proceed to count its Eulerian cycles.

Notice that for any vertices u and v there is a unique path of length n from u to v. Therefore the  $k^{n-1} \times k^{n-1}$  adjacency matrix A satisfies  $A^n = J$ , where J is the matrix whose entries are all equal to 1. We already saw that the eigenvalues of J are  $0, \ldots, 0, k^n$ . Since the trace of A is k, the eigenvalues of A must be  $0, \ldots, 0, k$ . Therefore the Laplacian L = kI - A has eigenvalues  $k, \ldots, k, 0$ . It follows from Corollary 3.1.10 that the de Bruijn graph has  $c(G, v) = \frac{1}{k^{n-1}}k^{(k^{n-1}-1)} = k^{k^{n-1}-n}$ oriented spanning trees rooted at any vertex v, and

$$c(G, v) \cdot \prod_{w \in V} (\text{outdeg}(w) - 1)! = k^{k^{n-1} - n} \cdot (k - 1)!^{k^{n-1}}$$

Eulerian cycles, as desired.

#### 3.1.5 Counting perfect matchings: the Pfaffian method

A **perfect matching** of a graph G = (V, E) is a set M of edges such that every vertex of G is on exactly one edge from M. We are interested in computing the number m(G) of perfect matchings of a graph G. We cannot expect to be able to do this in general; in fact, even for bipartite graphs G, the problem of computing m(G) is #P-complete. However, for many graphs of interest, including all planar graphs, there is a beautiful technique which produces a determinantal formula for m(G).

**Determinants and Pfaffians.** Let A be a skew-symmetric matrix of size  $2m \times 2m$ , so  $A^T = -A$ . The Pfaffian is a polynomial encoding the matchings of the complete graph  $K_{2m}$ . A perfect matching M of the complete graph  $K_{2m}$  is a partition M of [2m] into disjoint pairs  $\{i_1, j_1\}, \ldots, \{i_m, j_m\}$ , where  $i_k < j_k$  for  $1 \le k \le m$ . Draw the points  $1, \ldots, 2m$  in order on a line and connect each  $i_k$  to  $j_k$  by a semicircle above the line. Let  $\operatorname{cr}(M)$  be the number of crossings in this drawing, and let  $\operatorname{sign}(M) = (-1)^{\operatorname{cr}(M)}$ . Let  $a_M = a_{i_1j_1} \cdots a_{i_nj_n}$ . The **Pfaffian** of A is

$$Pf(A) = \sum_{M} sign(M) a_M$$

summing over all perfect matchings M of the complete graph  $K_n$ .

**Theorem 3.1.12.** If A is a skew-symmetric matrix, so  $A^T = -A$ , then

$$\det(A) = \Pr(A)^2. \tag{9}$$

Sketch of Proof. The first step is to show that the skew symmetry of A causes many cancellations in the determinant, and

$$\det A = \sum_{\pi \in ECS_n} \operatorname{sign}(\pi) a_{\pi}$$

where  $ECS_n \subset S_n$  is the set of permutations of [n] having only even cycles. Then, to prove that this equals  $(\sum_M \operatorname{sign}(M)a_M)^2$ , we need a bijection between ordered pairs  $(M_1, M_2)$  of matchings and permutations  $\pi$  in  $ECS_n$  such that  $a_{M_1}a_{M_2} = a_{\pi}$  and  $\operatorname{sign}(M_1)\operatorname{sign}(M_2) = \operatorname{sign}(\pi)$ . We now describe such a bijection.



Figure 3.5: The matchings  $\{1,3\}, \{2,6\}, \{4,8\}, \{5,7\}$  and  $\{1,5\}, \{2,4\}, \{3,7\}, \{6,8\}$  give the permutation (1375)(2684) with  $(-a_{13}a_{26}a_{48}a_{57})(-a_{15}a_{24}a_{37}a_{68}) = a_{13}a_{37}a_{75}a_{51}a_{26}a_{68}a_{84}a_{42}$ .

Draw the matchings  $M_1$  and  $M_2$  above and below the points  $1, \ldots, n$  on a line, respectively. Let  $\pi$  be the permutation given by the cycles of the resulting graph, where each cycle is oriented following the direction of  $M_1$  at its smallest element. This is illustrated in Figure 3.5. It is clear that  $a_{M_1}a_{M_2} = a_{\pi}$ , while some care is required to show that  $\operatorname{sign}(M_1)\operatorname{sign}(M_2) = \operatorname{sign}(\pi)$ . For details, see for example [Aig07].

**Counting perfect matchings via Pfaffians.** Suppose we wish to compute the number m(G) of perfect matchings of a graph G = (V, E) with no loops. After choosing an orientation of the edges, we define the  $V \times V$  signed adjacency matrix S(G) whose entries are

$$s_{ij} = \begin{cases} 1 & \text{if } i \to j \text{ is an edge of } G \\ -1 & \text{if } j \to i \text{ is an edge of } G \\ 0 & \text{otherwise.} \end{cases}$$

Then  $s_M = s_{i_1 j_1} \cdots s_{i_m j_m}$  is nonzero if and only if  $\{i_1, j_1\}, \ldots, \{i_m, j_m\}$  is a perfect matching of G.

We say that our edge orientation is **Pfaffian** if all the perfect matchings of G have the same sign. At the moment there is no efficient test to determine whether a graph admits a Pfaffian orientation. There is a simple combinatorial restatement: an orientation is Pfaffian if every even cycle C such that  $G \setminus V(C)$  has a perfect matching has an odd number of edges in each direction.

Fortunately, we have the following result of Kasteleyn [Kas63, LP09]:

Every planar graph has a Pfaffian orientation.

This is very desirable, because Theorem 3.1.12 implies:

For a Pfaffian orientation of G,  $m(G) = \sqrt{\det S(G)}$ .

Therefore the number of matchings of a planar graph is reduced to the evaluation of a combinatorial determinant. We will see in Section 3.2 that there are many techniques at our disposal to carry out this evaluation.

Let us illustrate this method with an important example, due to Kasteleyn [Kas63] and Temperley– Fisher [TF61].

**Theorem 3.1.13.** The number  $m(R_{a,b})$  of matchings of the  $a \times b$  rectangular grid  $R_{a,b}$  (where we assume b is even) is

$$m(R_{a,b}) = 4^{\lfloor a/2 \rfloor (b/2)} \prod_{j=1}^{\lfloor a/2 \rfloor} \prod_{k=1}^{b/2} \left( \cos^2 \frac{\pi j}{a+1} + \cos^2 \frac{\pi k}{b+1} \right) \approx e^{\frac{G}{\pi}ab} \approx 1.3385^{ab},$$

where  $G = 1 - \frac{1}{9} + \frac{1}{25} - \frac{1}{49} + \cdots$  is Catalan's constant.

Clearly this is also the number of domino tilings of an  $a \times b$  rectangle.

Sketch of Proof. Orient all columns of  $R_{a,b}$  going up, and let the rows alternate between going right or left, assigning the same direction to all edges of the same row. The resulting orientation is Pfaffian because every square has an odd number of edges in each direction. The adjacency matrix

S satisfies  $m(R_{a,b}) = \sqrt{\det S}$ . To compute this determinant, it is slightly easier<sup>5</sup> to consider the following  $mn \times mn$  matrix B:

$$b_{ij} = \begin{cases} 1 & \text{if } i \text{ and } j \text{ are horizontal neighbors} \\ i & \text{if } i \text{ and } j \text{ are vertical neighbors} \\ 0 & \text{otherwise.} \end{cases}$$

We can obtain B from the S by scaling the rows and columns by suitable powers of i, so we still have  $m(R_{a,b}) = \sqrt{|\det B|}$ . We will compute this determinant in Section 3.2.

We then use this product formula to give an asymptotic formula for  $m(R_{a,b})$ . Note that  $\log m(R_{a,b})/ab$  may be regarded as a Riemann sum; as  $m, n \to \infty$  it converges to

$$c = \frac{1}{\pi^2} \int_0^{\pi/2} \int_0^{\pi/2} \log(4\cos^2 x + 4\cos^2 y) \, dx \, dy = \frac{G}{\pi}$$

where G is Catalan's constant. Therefore

$$m(R_{a,b}) \approx e^{\frac{G}{\pi}ab} \approx 1.3385^{ab}$$

Loosely speaking, this means that in a matching of the rectangular grid there are about 1.3385 degrees of freedom per vertex.

Obviously, this beautiful formula is not an efficient method of computing the exact value of  $m(R_{a,b})$  for particular values of a and b; it is not even clear why it gives an integer! There are alternative determinantal formulas for this quantity that are more tractable; see for example [Aig07, Section 10.1].

#### 3.1.6 Counting routings: the Lindström–Gessel–Viennot lemma

Let G be a directed graph with no directed cycles, which has a weight wt(e) on each edge e. We are most often interested in the unweighted case, where all weights are 1. Let  $S = \{s_1, \ldots, s_n\}$  and  $T = \{t_1, \ldots, t_n\}$  be two (not necessarily disjoint) sets of vertices, which we call sources and sinks, respectively. A **routing** from S to T is a set of paths  $P_1, \ldots, P_n$  from the n sources  $s_1, \ldots, s_n$  to the n sinks  $t_1, \ldots, t_n$  such that no two paths share a vertex. Assume that  $P_i$  starts at source  $s_i$  and ends at sink  $t_{\pi(i)}$  where  $\pi$  is a permutation of [n]. Define sign $(R) = \text{sign}(\pi)$ .

Let the weight of a path or a routing be the product of the weights of the edges it contains. Consider the  $n \times n$  path matrix Q whose (i, j) entry is

$$q_{ij} = \sum_{P \text{ path from } s_i \text{ to } t_j} \operatorname{wt}(P).$$

**Theorem 3.1.14.** (Lindström-Gessel-Viennot Lemma) Let G be a directed acyclic graph with edge weights, and let  $S = \{s_1, \ldots, s_n\}$  and  $T = \{t_1, \ldots, t_n\}$  be sets of vertices in G. Then the determinant of the  $n \times n$  path matrix Q is

$$\det Q = \sum_{R \text{ routing from } S \text{ to } T} sign(R)wt(R).$$

<sup>&</sup>lt;sup>5</sup>In fact, this is the matrix that Kasteleyn uses in his computation.

In particular, if all edge weights are 1 and if every routing takes  $s_i$  to  $t_i$  for all i, then

 $\det Q = number of routings from S to T.$ 

Proof. We have det  $A = \sum_{P} \operatorname{sign}(P)\operatorname{wt}(P)$  summing over all path systems  $P = \{P_1, \ldots, P_n\}$  from S to T; we need to cancel out the path systems which are not routings. For each such P, consider the lexicographically first pair of paths  $P_i$  and  $P_j$  which intersect, and let v be their first vertex of intersection. Now exchange the subpath of  $P_i$  from  $s_i$  to v and the subpath of  $P_j$  from  $s_j$  to v, to obtain new paths  $P'_i$  and  $P'_j$ . Replacing  $\{P_i, P_j\}$  with  $\{P'_i, P'_j\}$ , we obtain a new path system  $\varphi(P)$  from S to T. Notice that  $\varphi(\varphi(P)) = P$ , and  $\operatorname{sign}(\varphi(P))\operatorname{wt}(\varphi(P)) + \operatorname{sign}(P)\operatorname{wt}(P) = 0$ ; so for all non-routings P, the path systems P and  $\varphi(P)$  cancel each other out.

This theorem was also anticipated by Karlin and McGregor [KM59] in the context of birth-anddeath Markov processes.

**Determinants via routings.** The Lindström-Gessel-Viennot Lemma is also a useful combinatorial tool for computing determinants of interest, usually by enumerating routings in a lattice. We illustrate this with several examples.

1. (Binomial determinants) Consider the **binomial determinant** 

$$\binom{a_1,\ldots,a_n}{b_1,\ldots,b_n} = \det \left[ \binom{a_i}{b_j} \right]_{1 \le i,j \le n}$$

where  $0 \leq a_1 < \cdots < a_n$  and  $0 \leq b_1 < \cdots < b_n$  are integers. These determinants arise as coefficients of the Chern class of the tensor product of two vector bundles. [Las78] This algebro-geometric interpretation implies these numbers are positive integers; as combinatorialists, we would like to know what they count.

A SE path is a lattice path in the square lattice  $\mathbb{N}^2$  consisting of unit steps south and east. Consider the sets of points  $A = \{A_1, \ldots, A_n\}$  and  $B = \{B_1, \ldots, B_n\}$  where  $A_i = (0, a_i)$  and  $B_i = (b_i, b_i)$  for  $1 \le i \le n$ . Since there are  $\binom{a_i}{b_j}$  SE paths from  $A_i$  to  $B_j$ , and since every SE routing from A to B takes  $A_i$  to  $B_i$  for all i, we have

$$\binom{a_1, \dots, a_n}{b_1, \dots, b_n} = \text{ number of SE routings from } A \text{ to } B$$

This is the setting in which Gessel and Viennot discovered Theorem 3.1.14; they also evaluated these determinants in several special cases. [GV85] We now discuss one particularly interesting special case.

2. (Counting permutations by descent set) The **descent set** of a permutation  $\pi$  is the set of indices *i* such that  $\pi_i > \pi_{i+1}$ . We now prove that

$$\binom{c_1, \ldots, c_k, n}{0, c_1, \ldots, c_k}$$
 = number of permutations of  $[n]$  with descent set  $\{c_1, \ldots, c_k\}$ ,

for any  $0 < c_1 < \cdots < c_k < n$ . It is useful to define  $c_0 = 0, c_{k+1} = n$ .

Encode such a permutation  $\pi$  by a routing as follows. For each *i* let  $f_i$  be the number of indices  $j \leq i$  such that  $\pi_j \leq \pi_i$ . Note that the descents  $c_1, \ldots, c_k$  of  $\pi$  are the positions where

f does not increase. Splitting f at these positions, we are left with k + 1 increasing subwords  $f^1, \ldots, f^{k+1}$ . Now, for  $1 \le i \le n+1$  let  $P_i$  be the NW path from  $(c_{i-1}, c_{i-1})$  to  $(0, c_i)$  taking steps north precisely at the steps listed in  $f^i$ . These paths give one of the routings enumerated by the binomial determinant in question, and this is a bijection. [GV85]



Figure 3.6: The routing corresponding to  $\pi = 28351674$  and  $f(\pi) = 12.23.157.4$ .

3. (Rhombus tilings and plane partitions) Let  $R_n$  be the number of tilings of a regular hexagon of side length n using unit rhombi with angles 60° and 120°. There are several equivalent combinatorial models for this problem, illustrated in Figure 3.7, which we now discuss.

Firstly, it is almost inevitable to view these tilings as three-dimensional pictures. This shows that  $R_n$  is also the number of ways of stacking unit cubes into the corner of a cubical box of side length n. Incidentally, this three-dimensional view makes it apparent that there are exactly  $n^2$  rhombi of each one of the 3 possible orientations.

Secondly, we may consider the triangular grid inside our hexagon, and place a dot on the center of each triangle. These dots form a hexagonal grid, where two dots are neighbors if they are at distance 1 from each other. Finally, join two neighboring dots when the corresponding triangles are covered by a tile. The result is a perfect matching of the hexagonal grid.

Next, on each one of the  $n^2$  squares of the floor of the box, write down the number of cubes above it. The result is a **plane partition**: an array of non-negative integers (finitely many of which are non-zero) which is weakly decreasing in each row and column. We conclude that  $R_n$  is also the number of plane partitions whose non-zero entries are at most n, and fit inside an  $n \times n$  square.

Finally, given such a rhombus tiling, construct n paths as follows. Each path starts at the center of one of the vertical edges on the western border of the hexagon, and successively crosses each tile splitting it into equal halves. It eventually comes out at the southeast side of the diamond, at the same height where it started (as is apparent from the 3-D picture). The final result is a routing from the n sources  $S_1, \ldots, S_n$  on the left to the sinks  $T_1, \ldots, T_n$  on the right in the "rhombus" graph shown below. It is clear how to recover the tiling from the routing. Since there are  $\binom{2n}{n+i-j}$  paths from  $S_i$  to  $T_j$ , the Lindström–Gessel-Viennot lemma



Figure 3.7: Four models for the rhombus tilings of a hexagon.

tells us that  $R_n$  is given by the determinant

$$R_n = \det\left[\binom{2n}{n+i-j}\right]_{1 \le i,j \le n} = \prod_{i,j,k=1}^n \frac{i+j+k-1}{i+j+k-2}.$$

We will prove this product formula in Section 3.2.

4. (Catalan determinants, multitriangulations, and Pfaffian rings) The **Hankel matrices** of a sequence  $A = (a_0, a_1, a_2, ...)$  are

$$H_n(A) = \begin{pmatrix} a_0 & a_1 & \cdots & a_n \\ a_1 & a_2 & \cdots & a_{n+1} \\ \vdots & \vdots & \ddots & \vdots \\ a_n & a_{n+1} & \cdots & a_{2n} \end{pmatrix}, \qquad H'_n(A) = \begin{pmatrix} a_1 & a_2 & \cdots & a_{n+1} \\ a_2 & a_3 & \cdots & a_{n+2} \\ \vdots & \vdots & \ddots & \vdots \\ a_{n+1} & a_{n+2} & \cdots & a_{2n+1} \end{pmatrix}.$$

Note that if we know the Hankel determinants det  $H_n(A)$  and det  $H'_n(A)$  and they are nonzero for all n, then we can use them as a recurrence relation to recover each  $a_k$  from  $a_0, \ldots, a_{k-1}$ . There is a natural interpretation of the Hankel matrices of the Catalan sequence  $C = (C_0, C_1, C_2, \ldots)$ . Consider the "diagonal" grid on the upper half plane with steps (1, 1) and (1,-1). Let  $A_i = (-2i,0)$  and  $B_i = (2i,0)$ . Then there are  $C_{i+j}$  paths from  $A_i$  to  $B_j$ , and there is clearly a unique routing from  $(A_0,\ldots,A_n)$  to  $(B_0,\ldots,B_n)$ . This proves that det  $H_n(C) = 1$ , and an analogous argument proves that det  $H'_n(C) = 1$ . Therefore

 $\det H_n(A) = \det H'_n(A) = 1$  for all  $n \ge 0 \iff A$  is the Catalan sequence.



Figure 3.8: Routing interpretation of the Hankel determinant  $H_n(C)$ .

The Hankel determinants of the shifted Catalan sequences also arise naturally in several contexts; they are given by:

$$\det \begin{pmatrix} C_{n-2k} & C_{n-2k+1} & \cdots & C_{n-k-1} \\ C_{n-2k+1} & C_{n-2k+2} & \cdots & C_{n-k} \\ \vdots & \vdots & \ddots & \vdots \\ C_{n-k-1} & C_{n-k} & \cdots & C_{n-2} \end{pmatrix} = \prod_{i+j \le n-2k-1} \frac{i+j+2(k-1)}{i+j}.$$
 (10)

There are several ways of proving this equality; for instance, it is a consequence of [Kra99, Theorem 26]. We describe three appearances of this determinant.

(a) A k-fan of Dyck paths of length 2n is a collection of k Dyck paths from (-n, 0) to (n, 0) which do not cross (although they necessarily share some edges). Shifting the (i + 1)st path i units up and adding i upsteps at the beginning and i downsteps at the end, we obtain a routing of k Dyck paths starting at the points  $A = \{-(n + k - 1), \ldots, -(n + 1), -n\}$  and ending at the points  $B = \{n, n + 1, \ldots, n + k - 1\}$  on the x axis. It follows that the number of k-fans of Dyck paths of length 2(n - 2k) is given by (10).



Figure 3.9: A k-fan of Dyck paths.

(b) There is also an extension of the classical one-to-one correspondence between Dyck paths and triangulations of a polygon. Define a k-crossing in an n-gon to be a set of k diagonals

which cross pairwise. A k-triangulation is a maximal set of diagonals with no (k + 1)crossings. The main enumerative result, due to Jonsson [Jon05], is that the number of ktriangulations of an n-gon is also given by (10). [Jon05]. A subtle bijection with fans of Dyck paths is given in [SS12].

Several properties of triangulations extend non-trivially to this context. For example, every k-triangulation has exactly k(2n - 2k - 1) diagonals. [Nak00, DKM02]. The k-triangulations are naturally the facets of a simplicial complex called the **multiassociahedron**, which is topologically a sphere [Jon03]; it is not currently known whether it is polytopal. There is a further generalization in the context of Coxeter groups, with connections to cluster algebras [CLS14].

(c) These determinants also arise naturally in the commutative algebraic properties of Pfaffians, defined earlier in this section. Let A be a skew-symmetric  $n \times n$  matrix whose entries above the diagonal are indeterminates  $\{a_{ij} : 1 \leq i < j \leq n\}$  over a field k. Consider the **Pfaffian ideal**  $I_k(A)$  generated by the  $\binom{n}{2k}$  Pfaffian minors of A of size  $2k \times 2k$ , and the **Pfaffian ring**  $R_k(A) = k[a_{ij}]/I_k(A)$ . Then the multiplicity of the Pfaffian ring  $R_k(X)$  is also given by (10). [HT92, GK04]

5. (Schröder determinants and Aztec diamonds.) Recall from Section 2.2.2 that a Schröder path of length n is a path from (0,0) to (2n,0) using steps NE = (1,1), SE = (1,-1), and E = (2,0) which stays above the x axis. The Hankel determinant det  $H_n(R)$  counts the routings of Schröder paths from the points  $A = \{0, -2, \ldots, -(2n)\}$  to the points  $B = \{0, 2, \ldots, 2n\}$  on the x-axis.

These Hankel determinants have a natural interpretation in terms of tilings. Consider the **Aztec diamond**<sup>6</sup>  $AD_n$  consisting of 2n rows centered horizontally, consisting successively of  $2, 4, \ldots, 2n, 2n, \ldots, 4, 2$  squares. We are interested in counting the tilings of the Aztec diamond into dominoes.



Figure 3.10: A tiling of the Aztec diamond and the corresponding routing.

Given a domino tiling of  $AD_n$ , construct n paths as follows. Each path starts at the center

<sup>&</sup>lt;sup>6</sup>This shape is called the **Aztec diamond** because it is reminiscent of designs of several Native American groups. Perhaps the closest similarity is with Mayan pyramids, such as the Temple of Kukulcán in Chichén Itz '; the name **Mayan diamond** would have beenAztec more appropriate.

of one of the vertical unit edges on the southwest border of the diamond, and successively crosses each tile that it encounters following a straight line through the center of the tile. It eventually comes out at the southeast side of the diamond, at the same height where it started. If we add *i* initial NE steps NE steps and *i* final SE steps to the (i + 1)st path for each *i*, the result will be a routing of Schröder paths from  $A = \{-(2n), \ldots, -2, 0\}$  to  $B = \{0, 2, \ldots, 2n\}$ . In fact this correspondence is a bijection [EF05].

We will prove in Section 3.2 that

$$\begin{cases} \det H_n(A) = 2^{n(n-1)/2} \\ \det H'_n(A) = 2^{n(n+1)/2} \text{ for all } n \ge 0 \quad \iff \quad A \text{ is the Schröder sequence} \end{cases}$$

It will follow that

number of domino tilings of the Aztec diamond  $AD_n = 2^{n(n+1)/2}$ .

# 3.2 Computing determinants

In light of Section 3.1, it is no surprise that combinatorialists have become talented at computing determinants. Fortunately, this is a very classical problem with connections to many branches of mathematics and physics, and by now there are numerous general techniques and guiding examples available to us. Krattenthaler's surveys [Kra99] and [Kra05] are excellent references which have clearly influenced the exposition in this section. We now highlight some of the key tools and examples.

# 3.2.1 Is it known?

Of course, when we wish to evaluate a new determinant, one first step is to check whether it is a special case of some known determinantal evaluation. Starting with classical evaluations such as the Vandermonde determinant

$$\det(x_i^{j-1})_{1 \le i, j \le n} = \prod_{1 \le i < j \le n} (x_j - x_i),$$
(11)

there is now a wide collection of powerful results at our disposal. A particularly useful one [Kra99, Lemma 3] states that for any  $x_1, \ldots, x_n, a_2, \ldots, a_n, b_2, \ldots, b_n$  we have:

$$\det \left[ (x_i + b_2) \cdots (x_i + b_j) (x_i + a_{j+1}) \cdots (x_i + a_n) \right]_{1 \le i, j \le n} = \prod_{1 \le i < j \le n} (x_j - x_i) \prod_{1 \le i < j \le n} (b_i - a_j).$$
(12)

For instance, as pointed out in [GK04] and [Kra99], the Catalan determinant (10) is a special case of this formula. Recognizing it as such is not immediate, but the product formula for Catalan numbers gives an indication of why this is feasible.

In fact, here is a counterintuitive principle: often the easiest way to prove a determinantal identity is to generalize it. It is very useful to introduce as many parameters as possible into a determinant, while making sure that the more general determinant still evaluates nicely. We will see this principle in action several times in what follows.

#### 3.2.2 Row and column operations.

A second step is to check whether the standard methods of computing determinants are useful: Laplace expansion by minors, or performing row and column operations until we get a matrix whose determinant we can compute easily. For example, recall the determinant  $L_0(K_n)$  of the  $(n-1) \times (n-1)$  reduced Laplacian of the complete graph  $K_n$ , discussed in Section 3.1.3. We can compute it by first adding all rows to the first row, and then adding the first row to all rows:

$$\det L_0(K_n) = \begin{vmatrix} n-1 & -1 & \cdots & -1 \\ -1 & n-1 & \cdots & -1 \\ \vdots & \vdots & \ddots & \vdots \\ -1 & -1 & \cdots & n-1 \end{vmatrix} = \begin{vmatrix} 1 & 1 & \cdots & 1 \\ -1 & n-1 & \cdots & -1 \\ \vdots & \vdots & \ddots & \vdots \\ -1 & -1 & \cdots & n-1 \end{vmatrix} = \begin{vmatrix} 1 & 1 & \cdots & 1 \\ 0 & n & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & n \end{vmatrix} = n^{n-2},$$

reproving Theorem 3.1.6.1.

### 3.2.3 Identifying linear factors.

Many  $n \times n$  determinants of interest have formulas of the form det  $M(\mathbf{x}) = cL_1(\mathbf{x}) \cdots L_n(\mathbf{x})$  where c is a constant and the  $L_i(\mathbf{x})$  are linear functions in the variables  $\mathbf{x} = (x_1, \ldots, x_k)$ . We may prove such a formula by first checking that each  $L_i(\mathbf{x})$  is indeed a factor of M, and then computing the constant c.

The best known application of this technique is the proof of the formula (11) for Vandermonde's determinant  $V(x_1, \ldots, x_n)$ . If  $x_i = x_j$  for  $i \neq j$ , then rows *i* and *j* are equal, and the determinant is 0. It follows that  $x_i - x_j$  must be a factor of the polynomial det  $V(x_1, \ldots, x_n)$ . Since this polynomial is homogeneous of degree  $\binom{n}{2}$ , it must equal a constant times  $\prod_{i < j} (x_i - x_j)$ . Comparing the coefficients of  $x_1^0 x_2^1 \cdots x_n^{n-1}$  we see that the constant equals 1.

A similar argument may be used to prove the more general formula (12).

To use this technique, it is sometimes necessary to introduce new variables into our determinant. For example, the formula  $\det(i^{j-1})_{1 \le i,j \le n} = 1^{n-1}2^{n-2}\cdots(n-1)^1$  cannot immediately be treated with this technique. However, the factorization of the answer suggests that this may be a special case of a more general result where this method does apply; in this case, Vandermonde's determinant.

#### 3.2.4 Computing the eigenvalues.

Sometimes we can compute explicitly the eigenvalues of our matrix, and multiply them to get the determinant. One common technique is to produce a complete set of eigenvectors.

- 1. (The Laplacian of the complete graph  $K_n$ ) Revisiting the example above, the Laplacian of the complete graph is  $L(K_n) = nI - J$  where I is the identity matrix and J is the matrix all of whose entries equal 1. We first find the eigenvalues of J: 0 is an eigenvalue of multiplicity n-1, as evidenced by the linearly independent eigenvectors  $e_1 - e_2, \ldots, e_{n-1} - e_n$ . Since the sum of the eigenvalues is tr(J) = n, the last eigenvalue is n; an eigenvector is  $e_1 + \cdots + e_{n-1}$ . Now, if v is an eigenvector for J with eigenvalue  $\lambda$ , then it is an eigenvector for nI - J with eigenvalue  $n - \lambda$ . Therefore the eigenvalues of nI - J are  $n, n, \ldots, n, 0$ . Using Theorem 3.1.5, we have reproved yet again that det  $L_0(K_n) = \frac{1}{n}(n^{n-1}) = n^{n-2}$ .
- 2. (The Laplacian of the *n*-cube  $C_n$ ) A more interesting example from Section 3.1.3 is the reduced Laplacian  $L_0(C_n)$  of the graph of the *n*-dimensional cube. By producing explicit eigenvectors,

one may prove that if the Laplacians L(G) and L(H) have eigenvalues  $\{\lambda_i : 1 \leq i \leq a\}$  and  $\{\mu_j : i \leq j \leq b\}$  then the Laplacian of the product graph  $L(G \times H)$  has eigenvalues  $\{\lambda_i + \mu_j : 1 \leq i \leq a, 1 \leq j \leq b\}$ . Since  $C_1$  has eigenvalues 0 and 2, this implies that  $C_n = C_1 \times \cdots \times C_1$  has eigenvalues  $0, 2, 4, \ldots, 2n$  with multiplicities  $\binom{n}{0}, \binom{n}{1}, \ldots, \binom{n}{n}$ , respectively. Therefore the number of spanning trees of the cube  $C_n$  is

$$\det L_0(C_n) = \frac{1}{2^n} 2^{\binom{n}{1}} 4^{\binom{n}{2}} \cdots (2n)^{\binom{n}{n}} = 2^{2^n - n - 1} 1^{\binom{n}{1}} 2^{\binom{n}{2}} \cdots n^{\binom{n}{n}}.$$

3. (The perfect matchings of a rectangle) An even more interesting example comes from the perfect matchings of the  $a \times b$  rectangle, which we discussed in Section 3.1.5. Let V be the 4mn-dimensional vector space of functions  $f : [2m] \times [2n] \to \mathbb{C}$ , and consider the linear transformation  $L: V \to V$  given by

$$(Lf)(x,y) = f(x-1,y) + f(x+1,y) + if(x,y-1) + if(x,y+1),$$

where f(x, y) = 0 when  $x \in \{0, a+1\}$  or  $y \in \{0, b+1\}$ . The matrix of this linear transformation is precisely the one we are interested in. A straightforward computation shows that the following are eigenfunctions and eigenvalues of L:

$$g_{k,l}(x,y) = \sin \frac{k\pi x}{a+1} \sin \frac{l\pi y}{b+1}, \qquad \lambda_{k,l} = 2\cos \frac{k\pi}{a+1} + 2i\cos \frac{l\pi}{b+1}$$

for  $1 \le k \le a$  and  $1 \le l \le b$ . (Note that  $g_{k,l}(x, y) = 0$  for  $x \in \{0, a\}$  or  $y \in \{0, b\}$ .) This is then the complete list of eigenvalues for L, so

det 
$$L = 2^{ab} \prod_{k=1}^{a} \prod_{l=1}^{b} \left( \cos \frac{k\pi}{a+1} + i \cos \frac{l\pi}{b+1} \right)$$

## 3.2.5 LU factorizations.

A classic result in linear algebra states that, under mild hypotheses, a square matrix M has a unique factorization

$$M = LU$$

where L is a lower triangular matrix and U is an upper triangular matrix with all diagonal entries equal to 1. Computer algebra systems can compute the LU-factorization of a matrix, and if we can guess and prove such a factorization it will follow immediately that det M equals the product of the diagonal entries of L.

An interesting application of this technique is the determinant

$$\det(\gcd(i,j))_{1\le i,j\le n} = \prod_{i=1}^{n} \varphi(i), \tag{13}$$

where  $\varphi(k) = \{i \in \mathbb{N} : (\gcd(i, k) = 1 \text{ and } 1 \leq i \leq k\}$  is Euler's totient function. This is a special case of a more general formula for semilattices which is easier to prove. For this brief computation, we assume familiarity with the Möbius function  $\mu$  and the zeta function  $\zeta$  of a poset; these will be treated in detail in Section 4.6.

Let P be a finite meet semilattice and consider any function  $F : P \times P \to k$ . We will prove the **Lindström–Wilf determinantal formula**:

$$\det F(p \lor q, p)_{p,q \in P} = \prod_{p \in P} \left( \sum_{r \ge p} \mu(p, r) F(r, p) \right)$$
(14)

Computing some examples will suggest that the LU factorization of F is F = MZ where

$$M_{pq} = \begin{cases} \sum_{r \ge q} \mu(q, r) F(r, p) & \text{if } p \le q, \\ 0 & \text{otherwise,} \end{cases} \qquad Z_{pq} = \begin{cases} 1 & \text{if } p \ge q, \\ 0 & \text{otherwise} \end{cases}$$

This guess is easy to prove, and it immediately implies (14). In turn, applying the Lindström–Wilf to the poset of integers  $\{1, \ldots, n\}$  ordered by reverse divisibility and the function F(x, y) = x, we obtain (13).

Another interesting special case is the determinant:

$$\det(x^{\operatorname{rank}(p\vee q)})_{p,q\in P} = \prod_{p\in P} \left(x^{\operatorname{rank}(p)} \chi_{[p,\widehat{1}]}(1/x).\right)$$

where  $\chi_{[p,\hat{1}]}(x)$  is the characteristic polynomial of the interval  $[x,\hat{1}]$ . When P is the partition lattice  $\Pi_n$ , this determinant arises in Tutte's work of the Birkhoff-Lewis equations [Tut91].

#### 3.2.6 Hankel determinants and continued fractions.

For Hankel determinants, the following connection with continued fractions [Wal48] is extremely useful. If the expansion of the generating function for a sequence  $f_0, f_1, \ldots$  as a **J-fraction** is

$$\sum_{n=0}^{\infty} f_n x^n = \frac{\mu_0}{1 + a_0 x - \frac{b_1 x^2}{1 + a_1 x - \frac{b_2 x^2}{1 + a_2 x - \dots}}},$$

then the Hankel determinants of  $a_0, a_1, \ldots$  equal

$$\det H_n(A) = \mu_0^n b_1^{n-1} b_2^{n-2} \cdots b_{n-2}^2 b_{n-1}$$

For instance, using the generating function for the Schrörder numbers  $r_n$ , it is easy to prove that

$$\sum_{n=0}^{\infty} r_n x^n = \frac{1}{1 - 2x - \frac{2x^2}{1 - 3x - \frac{2x^2}{1 - 3x - \dots}}}, \qquad \sum_{n=0}^{\infty} r_{n+1} x^n = \frac{2}{1 - 3x - \frac{2x^2}{1 - 3x - \frac{2x^2}{1 - 3x - \dots}}}.$$

Therefore

det 
$$H_n(R) = 2^{n(n-1)/2}$$
, det  $H'_n(R) = 2^{n(n+1)/2}$ ,

as stated in Section 3.1.6.

By computer calculation, it is often easy to guess J-fractions experimentally. With a good guess in place, there is an established procedure for proving their correctness, rooted in the theory of orthogonal polynomials; see [Kra99, Section 2.7]. **Dodgson condensation.** It is often repeated that Lewis Carroll, author of *Alice in Wonderland*, was also an Anglican deacon and a mathematician, publishing under his real name, Rev. Charles L. Dodgson. His contributions to mathematics are discussed less often, and one of them is an elegant method for computing determinants.

To compute an  $n \times n$  determinant A, one creates a square pyramid of numbers, consisting of n+1 levels of size  $n+1, n \dots, 1$ , respectively. On the bottom level we place an  $(n+1) \times (n+1)$  array of 1s, and on the next level we place the  $n \times n$  matrix A. Each subsequent floor is obtained from the previous two by the following rule: each new entry is given by f = (ad - bc)/e where f is directly above the entries  $\begin{pmatrix} a & b \\ c & d \end{pmatrix}$  and two floors above the entry e.<sup>7</sup> The top entry of the pyramid is the determinant. For example, the computation

shows that the determinant of the  $4 \times 4$  determinant is 21.

Dodgson's condensation method relies on the following fact, due to Jacobi. If A is an  $n \times n$  matrix and  $A_{i_1,\ldots,i_k}$ ;  $j_1,\ldots,j_k$  denotes the matrix A with rows  $i_1,\ldots,i_k$  and columns  $j_1,\ldots,j_k$  removed, then

$$\det A \cdot \det A_{1,n;1,n} = \det A_{1;1} \cdot \det A_{n;n} - \det A_{1;n} \cdot \det A_{n;1}.$$
(15)

This proves that the numbers appearing in the pyramid are precisely the determinants of the "contiguous" submatrices of A, consisting of consecutive rows and columns.

If we have a guess for the determinant of A, as well as the determinants of its contiguous submatrices, Dodgson condensation is an extremely efficient method to prove it. All we need to do is to verify that our guess satisfies (15).

To see how this works in an example, let us use Dodgson condensation to prove the formula in Section 3.1.6 for  $R_n = \det {\binom{2n}{n+i-j}}_{1 \le i,j \le n}$ , the number of stacks of unit cubes in the corner of an  $n \times n \times n$  box. The first step is to guess the determinant of the matrix in question, as well as all its contiguous submatrices; they are all of the form  $R(a, b, c) = \det {\binom{a+b}{a+i-j}}_{1 \le i,j \le c}$ , where a + b = 2n. This more general determinant is equally interesting combinatorially: it counts the stacks of unit cubes in the corner of an  $a \times b \times c$  box. By computer experimentation, it is not too difficult to arrive at the following guess:

$$R(a,b,c) = \det\left[\binom{a+b}{a+i-j}\right]_{1 \le i,j \le c} = \prod_{i=1}^{a} \prod_{j=1}^{b} \prod_{k=1}^{c} \frac{i+j+k-1}{i+j+k-2}$$

Proving this formula by Dodgson condensation is straightforward; we just need to check that our conjectural product formula holds for c = 0, 1 and that it satisfies (15); that is,

$$R(a, b, c+1)R(a, b, c-1) = R(a, b, c)^2 - R(a+1, b-1, c)R(a-1, b+1, c).$$

<sup>&</sup>lt;sup>7</sup>Special care is required when 0s appear in the interior of the pyramid.

For more applications of Dodgson condensation, see for example [AZ01]

There is a wonderful connection between Dodgson condensation, Aztec diamonds, and **alter**nating sign matrices, which we now describe. Let us construct a square pyramid of numbers where levels n + 2 and n + 1 are given by two matrices  $\mathbf{y} = (y_{ij})_{1 \le i,j \le n+2}$  and  $\mathbf{x} = x_{ij})_{1 \le i,j \le n+1}$ , respectively, and levels  $n - 1, \ldots, 2, 1$  are computed in terms of the lower rows using Dodgson's recurrence f = (ad - bc)/e. Let  $f_n(\mathbf{x}, \mathbf{y})$  be the entry at the top of the pyramid.

Remarkably, all the entries of the resulting pyramid will be Laurent monomials in the  $x_{ij}$ s and  $y_{ij}$ s; that is, their denominators are always monomials. This is obvious for the first few levels, but it becomes more and more surprising as we divide by more and more intricate expressions.

The combinatorial explanation for this fact is that each entry in the (n - k)th level of the pyramid encodes the domino tilings of a Aztec diamond  $AD_k$ . For instance, if n = 3, the entry at the top of the pyramid is



Figure 3.11: A domino tiling of  $AD_2$  and the corresponding monomial in  $f_2(\mathbf{x}, \mathbf{y})$ .

There is a simple bijection between the 8 terms of  $f_3$  and the 8 domino tilings of  $AD_2$ . Regard a tiling of  $AD_2$  as a graph with vertices on the underlying lattice, and add a vertical edge above and below the tiling, and a horizontal edge to the left and to the right of T. Now rotate the tiling 45°. Record the degree of each vertex, ignoring the outside corners on the boundary of the diamond, and subtract 3 from each vertex. This leaves us with an  $n \times n$  grid of integers within an  $(n+1) \times (n+1)$  grid of integers. Assign to it the monomial whose x exponents are given by the outer grid and whose y exponents are given by the inner grid. For example, the tiling in Figure 3.11 corresponds to the monomial  $(x_{12}x_{21}x_{23}x_{32})/(x_{22}y_{23}y_{32})$ .

In general, this gives a bijection between the terms of  $f_n(\mathbf{x}, \mathbf{y})$  and the domino tilings of the Aztec diamond  $AD_n$ . One may also check that there are no cancellations, so Dodgson condensation tell us that the number  $m_n$  of terms in  $f_n$  satisfies  $m_{n-1}m_{n+1} = 2m_n^2$ . This gives an alternative proof that the Aztec diamond  $Aztec_n$  has  $2^{n(n+1)/2}$  domino tilings.

We may also consider the patterns formed by the  $x_{ij}$ s by themselves (or of the  $y_{ij}$ s by themselves). In each individual monomial of  $f_n(\mathbf{x}, \mathbf{y})$ , the exponents of the  $x_{ij}$ s form an  $n \times n$  alternating sign matrix (ASM): a matrix of 1s, 0s, and -1s such that the nonzero entries in any row or column alternate  $1, -1, \ldots, -1, 1$ . Similarly, the exponents of the  $y_{ij}s$  form an ASM of size n-1.

Alternating sign matrices are fascinating objects in their own right, with connections to representation theory, statistical mechanics, and other fields. The number of alternating sign matrices of size n is

$$\frac{1!\,4!\,7!\cdots(3n-2)!}{n!\,(n+1)!\,(n+2)!\cdots(2n-1)!}.$$

For details on the history and solution of this difficult enumeration problem see [BP99, Rob91, Zei96].

# 4 **Posets**

This section is devoted to the enumerative aspects of the theory of partially ordered sets (posets). Section 4.1 introduces key definitions and examples. Section 4.2 discusses some families of lattices that are of special importance. In Section 4.3 we count chains and linear extensions of posets.

The remaining sections are centered around the *Möbius Inversion Formula*, which is perhaps the most useful enumerative tool in the theory of posets. This formula helps us count sets which have an underlying poset structure; it applies to many combinatorial settings of interest.

In Section 4.4 we discuss the Inclusion-Exclusion Principle, a special case of great importance. In Section 4.5 we introduce Möbius functions and the Möbius Inversion Formula. In particular, we catalog the Möbius functions of many important posets. The incidence algebra, a nice algebraic framework for understanding and working with the Möbius function, is discussed in Section 4.6. In Section 4.7 we discuss methods for computing Möbius functions of posets, and sketch proofs for the posets of Section 4.5. Finally, in Section 4.8, we discuss Eulerian posets and the enumeration of their flags, which gives rise to the **ab**-index and **cd**-index.

#### 4.1 Basic definitions and examples

A partially ordered set or poset  $(P, \leq)$  is a set P together with a binary relation  $\leq$ , called a *partial order*, such that

- For all  $p \in P$ , we have  $p \leq p$ .
- For all  $p, q \in P$ , if  $p \leq q$  and  $q \leq p$  then p = q.
- For all  $p, q, r \in P$ , if  $p \leq q$  and  $q \leq r$  then  $p \leq r$ .

We say that p < q if  $p \le q$  and  $p \ne q$ . We say that p and q are **comparable** if p < q or p > q, and they are **incomparable** otherwise. We say that q **covers** p if q > p and there is no  $r \in P$  such that q > r > p. When q covers p we write q > p.

**Example 4.1.1.** Many sets in combinatorics come with a natural partial order, and often the resulting poset structure is very useful for enumerative purposes. Some of the most important examples are the following:

- 1. (Chain) The poset  $\mathbf{n} = \{1, 2, ..., n\}$  with the usual total order.  $(n \ge 1)$
- 2. (Boolean lattice) The poset  $2^A$  of subsets of a set A, where  $S \leq T$  if  $S \subseteq T$ .
- 3. (Divisor lattice) The poset  $D_n$  of divisors of n, where  $c \leq d$  if c divides d.  $(n \geq 1)$
- 4. (Young's lattice) The poset Y of integer partitions, where  $\lambda \leq \mu$  if  $\lambda_i \leq \mu_i$  for all i.
- 5. (Partition lattice) The poset  $\Pi_n$  of set partitions of [n], where  $\pi \leq \rho$  if  $\pi$  refines  $\rho$ ; that is, if every block of  $\rho$  is a union of blocks of  $\pi$ .  $(n \geq 1)$
- 6. (Non-crossing partition lattice) The subposet  $NC_n$  of  $\Pi_n$  consisting of the non-crossing set partitions of [n], where there are no elements a < b < c < d such that a, c are together in one block and b, d are together in a different block.  $(n \ge 1)$

- 7. (Bruhat order on permutations) The poset  $S_n$  of permutations of [n], where  $\pi$  covers  $\rho$  if  $\pi$  is obtained from  $\rho$  by choosing two adjacent numbers  $\rho_i = a < b = \rho_{i+1}$  in  $\rho$  and exchanging their positions.  $(n \ge 1)$
- 8. (Subspace lattice) The poset  $L(\mathbb{F}_q^n)$  of subspaces of a finite dimensional vector space  $\mathbb{F}_q^n$ , where  $U \leq V$  if U is a subspace of V.  $(n \geq 1, q \ a \ prime \ power)$
- 9. (Distributive lattice) The poset J(P) of order ideals (subsets  $I \subseteq P$  such that  $j \in P$  and i < j imply  $i \in P$ ) ordered by containment.
- 10. (Face poset of a polytope) The poset F(P) of faces of a polytope P, ordered by inclusion.
- 11. (Face poset of a subdivision of a polytope) The poset  $\widehat{\mathcal{T}}$  of faces of a subdivision  $\mathcal{T}$  of a polytope P ordered by inclusion, with an additional maximum element.
- 12. (Subgroup lattice of a group) The poset L(G) of subgroups of a group G, ordered by containment.

The **Hasse diagram** of a finite poset P is obtained by drawing a dot for each element of P and an edge going down from p to q if p covers q. The diagram below shows the Hasse diagrams of some of the posets above. In particular, the Hasse diagram of  $2^{[n]}$  is the 1-skeleton of the *n*-dimensional cube. Figure 4.1 shows some examples.



Figure 4.1: The Hasse diagrams of the chain 4, Boolean lattice  $2^{[3]}$ , divisor lattice  $D_{18}$ , partition lattice  $\Pi_3$ , and Bruhat order  $S_3$ .

A subset Q of P is a **chain** if every pair of elements is comparable, and it is an **antichain** if every pair of elements is incomparable. The length of a chain C is |C| - 1. If there is a **rank** function  $r: P \to \mathbb{N}$  such that r(x) = 0 for any minimal element x and r(y) = r(x) + 1 whenever y > x, then P is called **graded** or **ranked**. The largest rank is called the **rank** or **height** of P. The **rank-generating function** of a finite graded poset is

$$R(P;x) = \sum_{p \in P} x^{r(p)}$$

All the posets of Example 4.1.1 are graded.

A poset P induces a poset structure on any subset  $Q \subseteq P$ ; a special case of interest is the interval  $[p,q] = \{r \in P : p \leq r \leq q\}$ . We call a poset locally finite if all its intervals are finite.

Given posets P and Q on disjoint sets, the **direct sum** P + Q is the poset on  $P \cup Q$  inheriting the order relations from P and Q, and containing no additional order relations between elements

of P and Q. The **direct product**  $P \times Q$  is the poset on  $P \times Q$  where  $(p,q) \leq (p',q')$  if  $p \leq p'$  and  $q \leq q'$ .

We have already seen examples of product posets. The Boolean lattice is  $2^A \cong \mathbf{2} \times \cdots \times \mathbf{2}$ . Also, if  $n = p_1^{t_1} \cdots p_k^{t_k}$  is the prime factorization of n, then  $D_n \cong (\mathbf{t_1} + \mathbf{1}) \times \cdots \times (\mathbf{t_k} + \mathbf{1})$ .

# 4.2 Lattices

A poset is a **lattice** if every two elements p and q have a least upper bound  $p \lor q$  and a greatest lower bound  $p \land q$ , called their **meet** and **join**, respectively. We will see this additional algebraic structure can be quite beneficial for enumerative purposes.

**Example 4.2.1.** All the posets in Example 4.1.1 are lattices, except for the Bruhat order. In most cases, the meet and join have easy descriptions. In **n**, the meet and join are the minimum and maximum, respectively. In  $2^A$  they are the intersection and union. In  $D_n$  they are the greatest common divisor and least common multiple. In Y they are the componentwise minimum and maximum. In  $\Pi_n$  and in  $NC_n$  the meet of two partitions  $\pi$  and  $\rho$  is the collection of intersections of a block of  $\pi$  and a block of  $\rho$ . In  $L(\mathbb{F}_q^n)$  the meet and join are the intersection and the span. In J(P) they are the intersection and the union. In F(P) the meet is the intersection. In L(G) the meet is the intersection.

Any lattice must have a unique minimum element  $\hat{0}$  and maximum element  $\hat{1}$ . An element covering  $\hat{0}$  is called an **atom**; an element covered by  $\hat{1}$  is called a **coatom**. To prove that a finite poset P is a lattice, it is sufficient to check that it has a  $\hat{1}$  and that any  $x, y \in P$  have a meet; then the join of x and y will be the (necessarily non-empty) meet of their common upper bounds. Similarly, it suffices to check that P has a  $\hat{0}$  and that any  $x, y \in P$  have a join.

**Distributive Lattices.** A lattice L is **distributive** if the join and meet operations satisfy the distributive properties:

$$x \lor (y \land z) = (x \lor y) \land (x \lor z), \qquad x \land (y \lor z) = (x \land y) \lor (x \land z)$$
(16)

for all  $x, y, z \in L$ . To prove that L is distributive, it is sufficient to check that to verify that one of the equations in (16) holds for all  $x, y, z \in L$ .

**Example 4.2.2.** There are several distributive lattices in Example 4.1.1: the chains  $\mathbf{n}$ , the Boolean lattices  $2^A$ , the divisor lattices  $D_n$ , and Young's lattice Y. This follows from the fact that the pairs of operations (min, max), (gcd, lcm) and  $(\cap, \cup)$  satisfy the distributive laws. On the other hand,  $\Pi_3$  is not distributive.

The most important – and in fact, the only – source of finite distributive lattices is the following. Given a poset P, a **downset** or **order ideal** I is a subset of P such that if  $i \in I$  and j < i then  $j \in I$ . A **principal** order ideal is one of the form  $P_{\leq p} = \{q \in P : q \leq p\}$ . Let J(P) be the **poset** of order ideals of P, ordered by inclusion.

**Theorem 4.2.3.** (Fundamental Theorem for Finite Distributive Lattices.) A poset L is a distributive lattice if and only if there exists a poset P such that  $L \cong J(P)$ . Sketch of Proof. Since the collection of order ideals of a poset P is closed under union and intersection, J(P) is a sublattice of  $2^{P}$ . The distributivity of  $2^{P}$  then implies that J(P) is a distributive lattice.

For the converse, let L be a lattice, and let P be the set of join-irreducible elements of L; that is, the elements  $p > \hat{0}$  which cannot be written as  $p = q \lor r$  for q, r < p. These are precisely the elements of L that cover exactly one element. The set P inherits a partial order from L, and this is the poset such that  $L \cong J(P)$ . The isomorphism is given by

$$\begin{array}{cccc} \phi: J(P) & \longrightarrow & L \\ I & \longmapsto & \bigvee_{p \in I} p \end{array}$$

and the inverse map is given by  $\phi^{-1}(l) = \{p \in P : p \leq l\}.$ 



Figure 4.2: A poset and the corresponding distributive lattice.

Theorem 4.2.3 extends to some infinite posets with minor modifications. Let  $J_f(P)$  be the set of finite order ideals of a poset P. Then the map  $P \mapsto J_f(P)$  is a bijection between the posets whose principal order ideals are finite and the locally finite distributive lattices with  $\hat{0}$ .

**Example 4.2.4.** The posets P of join-irreducibles of the distributive lattices  $L \cong J(P)$  of Example 4.2.2 are as follows. For  $L = \mathbf{n}$ ,  $P = \mathbf{n} - \mathbf{1}$  is a chain. For  $L = 2^A$ ,  $P = \mathbf{1} + \cdots + \mathbf{1}$  is an antichain. For  $L = D_n$ , where  $n = p_1^{t_1} \cdots p_k^{t_k}$ ,  $P = \mathbf{t}_1 + \cdots + \mathbf{t}_k$  is the disjoint sum of k chains. For L = Y,  $P = \mathbb{N} \times \mathbb{N}$  is a "quadrant".

Theorem 4.2.3 explains the abundance of cubes in the Hasse diagram of a distributive lattice L. For any element  $l \in L$  covered by n elements  $l_1, \ldots, l_n$  of L, the joins of the  $2^n$  subsets of  $\{l_1, \ldots, l_n\}$  are distinct, and form a copy of the Boolean lattice  $2^{[n]}$  inside L. The dual result holds as well.

The width of a poset P is the size of the largest antichain of P. Dilworth's Theorem [Dil50] states that this is the smallest integer w such that P can be written as the disjoint union of w chains.

**Theorem 4.2.5.** The distributive lattice J(P) can be embedded as an induced subposet of the poset  $\mathbb{N}^w$ , where w is the width of P.

*Proof.* Decompose P as the disjoint union of w chains  $C_1, \ldots, C_c$ . The map

$$\phi: J(P) \longrightarrow \mathbb{N}^w$$

$$I \longmapsto (|I \cap C_1|, \dots, |I \cap C_w|)$$

gives the desired inclusion.

**Geometric Lattices.** Now we introduce another family of lattices of great importance in combinatorics. We say that a lattice L is:

- semimodular if the following two equivalent conditions hold:
  - o L is graded and  $r(p) + r(q) \ge r(p \land q) + r(p \lor q)$  for all  $p, q \in L$ .
  - o If p and q both cover  $p \wedge q$ , then  $p \vee q$  covers both p and q.
- **atomic** if every element is a join of atoms.
- geometric if it is semimodular and atomic.

**Example 4.2.6.** In Figure 4.1, the posets  $2^{[3]}$  and  $\Pi_3$  are geometric, while the posets **4** and  $D_{18}$  are not.

Not surprisingly, the prototypical example of a geometric lattice comes from a natural geometric construction, illustrated in Figure 4.3. Let  $A = \{v_1, \ldots, v_n\}$  be a set of vectors in a vector space V. A **flat** is an subspace of V generated by a subset of A. We identify a flat with the set of  $v_i$ s that it contains. Let  $L_A$  be the set of flats of A, ordered by inclusion. Then  $L_A$  is a geometric lattice.



Figure 4.3: A vector configuration  $\mathbf{t}, \mathbf{u}, \mathbf{v}, \mathbf{w}$  (where  $\mathbf{t}, \mathbf{u}, \mathbf{v}$  are coplanar) and the corresponding geometric lattice.

The theory of geometric lattices is equivalent to the rich theory of *matroids*, which is the subject of Section 7.

**Supersolvable Lattices.** A lattice L is **supersolvable** if there exists a maximal chain, called an M-chain, such that the sublattice generated by C and any other chain of L is distributive. [Sta72]

Again unsurprisingly, an important example comes from supersolvable groups, but there are several other interesting examples. Here is a list of supersolvable lattices, and an M-chain in each case.

1. Distributive lattices: every maximal chain is an M-chain.

- 2. Partition lattice  $\Pi_n: 1|2|\cdots |n < 12|3|\cdots |n < 123|4|\cdots |n < \cdots < 123\cdots n$
- 3. Noncrossing partition lattice  $NC_n$ : the same chain as above.
- 4. Lattice of subspaces  $L(\mathbb{F}_q^n)$  of the vector space  $\mathbb{F}_q^n$  over a finite field  $\mathbb{F}_q$ : every maximal chain is an M-chain.
- 5. Subgroup lattices of finite supersolvable groups G: an M-chain is given by any normal series  $1 = H_0 \triangleleft H_1 \triangleleft \cdots \triangleleft H_k = G$  where each  $H_i$  is normal and  $H_i/H_{i-1}$  is cyclic of prime order.

Fortunately, there is a simple criterion to verify semimodularity. An *R*-labeling of a poset *P* is a labeling of the edges of the Hasse diagram of *P* with integers such that for any  $s \leq t$  there exists a unique maximal chain *C* from *s* to *t*.

**Theorem 4.2.7.** [McN03] A finite graded lattice of rank n is supersolvable if and only if it has an R-labeling for which the labels on every maximal chain are a permutation of  $\{1, \ldots, n\}$ .

# 4.3 Chain polynomials, order polynomials

The **zeta polynomial** of a finite poset P counts the multichains of various lengths in P. A **multichain** of length k in P is a sequence of possibly repeated elements  $t_0, t_1, \ldots, t_k \in P$  such that  $t_0 \leq t_1 \leq \cdots \leq t_k$ . Let

 $Z_P(k) =$ number of multichains of length k - 2 in P  $(k \ge 2)$ . (17)

There is a unique polynomial  $Z_P(k)$  satisfying (17) for all integers  $k \ge 2$ ; it is given by

$$Z_P(k) = \sum_{i \ge 2} b_i \binom{k-2}{i-2},$$
(18)

where  $b_i$  is the number of chains of length i-2 in P. This polynomial is called the **zeta polynomial** of P.

**Example 4.3.1.** The following posets have particularly nice zeta polynomials:

- 1.  $P = \mathbf{n}$ :  $Z(k) = \binom{n+k-2}{n-1}$ 2.  $P = B_n$ :  $Z(k) = k^n$
- 3.  $P = NC_n$ : (Kreweras, [Kre72])

$$Z(k) = \frac{1}{n} \binom{kn}{n-1}$$

The order polynomial of P counts the order-preserving labelings of P; it is defined by

 $\Omega_P(k) =$  number of maps  $f: P \to [k]$  such that p < q implies  $f(p) \leq f(q)$ 

for  $k \in \mathbb{N}$ . The next proposition shows that, once again, there is a unique polynomial taking these values at the natural numbers.

**Proposition 4.3.2.** For any poset P,  $\Omega_P(k) = Z_{J(P)}(k)$ .

*Proof.* An order-preserving map  $f : P \to [k]$  gives rise to a sequence of order ideals  $f^{-1}(\{1\}) \subseteq f^{-1}(\{1,2\}) \subseteq \cdots f^{-1}(\{1,\ldots,k\})$ , which is a multichain in J(P). Conversely, every sequence arises uniquely in this way.

A linear extension of P is an order-preserving labeling of the elements of P with the labels  $1, \ldots, n = |P|$  which extends the order of P; that is, a bijection  $f: P \to [n]$  such that p < q implies f(p) < f(q). Let

e(P) = number of linear extensions of P.

It follows from Proposition 4.3.2 and (18) that the zeta polynomial  $\Omega_P$  has degree |P|, and leading coefficient e(P)/|P|!.

The following is a method for computing e(P) recursively.

**Proposition 4.3.3.** Define  $e: J(P) \to \mathbb{N}$  recursively by

$$e(I) = \begin{cases} 1 & \text{if } I = \widehat{0}, \\ \sum_{J \leqslant I} e(J) & \text{otherwise.} \end{cases}$$

Then  $e(\widehat{1})$  is the number e(P) of linear extensions of P.

*Proof.* Let  $p_1, \ldots, p_k$  be the maximal elements of P. In a linear extension of P, one of  $p_1, \ldots, p_k$  has to be labelled n, and therefore

$$e(P) = e(P \setminus \{p_1\}) + \dots + e(P \setminus \{p_k\}).$$

This is equivalent to the desired recurrence.

It is useful to keep in mind that J(P) is a subposet of  $\mathbb{N}^w$  for w = w(P). The recurrence of Proposition 4.3.3 generalizes Pascal's triangle, which corresponds to the case  $P = \mathbb{N} + \mathbb{N}$ . When we apply it to  $P = \mathbb{N} + \cdots + \mathbb{N}$ , we get the recursive formula for multinomial coefficients.

**Example 4.3.4.** In some special cases, the problem of enumerating linear extensions is of fundamental importance.

- $e(\mathbf{n}_1 + \dots + \mathbf{n}_k) = \binom{n_1 + \dots + n_k}{n_1, \dots, n_k}$
- $e(\mathbf{2} \times \mathbf{n}) = C_n = \frac{1}{n+1} \binom{2n}{n}.$
- Let T be a tree poset of n elements, such that the Hasse diagram is a tree rooted at  $\widehat{0}$ . For each vertex v let  $t_v = |T_{>v}| = |\{w \in T : w \ge v\}|$ . Then

$$e(T) = \frac{n!}{\prod_{v \in T} t_v}$$

• Let  $\lambda$  be a Ferrers diagram of n cells, partially ordered by decreeing that each cell is covered by the cell directly below and the cell directly to the right, if they are in  $\lambda$ . The **hook** of a cell c consists of cells on the same row and to the right of c, those on the same column and below c, and c itself. Let  $h_c = |H_c|$ . Then

$$e(\lambda) = \frac{n!}{\prod_{c \in D} h_c}.$$

This is the dimension of the irreducible representation of the symmetric group  $S_n$  corresponding to  $\lambda$ . [Sag01]

## 4.4 The Inclusion-Exclusion Formula

Our next goal is to discuss one of the most useful enumerative tools for posets: Möbius functions and the Möbius inversion theorem. Before we do that, we devote this section to a special case that preceded and motivated them: the inclusion-exclusion formula.

**Theorem 4.4.1.** (Inclusion-Exclusion Principle) For any finite sets  $A_1, \ldots, A_n \subseteq X$ , we have

1. 
$$|A_1 \cup \dots \cup A_n| = \sum_i |A_i| - \sum_{i < j} |A_i \cap A_j| + \sum_{i < j < k} |A_i \cap A_j \cap A_k| - \dots \pm |A_1 \cap \dots \cap A_n|.$$
  
2. 
$$|\overline{A_1} \cap \dots \cap \overline{A_n}| = |X| - \sum_i |A_i| + \sum_{i < j} |A_i \cap A_j| - \dots \pm |A_1 \cap \dots \cap A_n|.$$

*Proof.* It suffices to prove one of these two equivalent equations. To prove the first one, consider an element x appearing in  $k \ge 1$  of the given sets. The number of times that x is counted in the right hand side is  $k - \binom{k}{2} + \cdots \pm \binom{k}{k} = 1$ .

We now present a slightly more general formulation.

**Theorem 4.4.2.** (Inclusion-Exclusion Principle) Let A be a set and consider two functions  $f_{=}, f_{\geq}$ :  $2^A \longrightarrow \mathbb{k}$  be functions from  $2^A$  to a field  $\mathbb{k}$ . Then

$$1. \ f_{\geq}(S) = \sum_{T \supseteq S} f_{=}(T) \ \text{for } S \subseteq A \quad \Longleftrightarrow \quad f_{=}(S) = \sum_{T \supseteq S} (-1)^{|T-S|} f_{\geq}(T) \ \text{for } S \subseteq A.$$
$$2. \ f_{\leq}(S) = \sum_{T \subseteq S} f_{=}(T) \ \text{for } S \subseteq A \quad \Longleftrightarrow \quad f_{=}(S) = \sum_{T \subseteq S} (-1)^{|S-T|} f_{\leq}(T) \ \text{for } S \subseteq A.$$

The most common interpretation is the following. Suppose we have a set U of objects and a set A of properties that each object in U may or may not satisfy. If we know, for each  $S \subseteq A$ , the number  $f_{\geq}(S)$  of elements having at least the properties in S (or the number  $f_{\leq}(S)$  of elements having at most the properties in S), then we obtain, for each  $S \subseteq A$ , the number  $f_{\equiv}(S)$  of elements having exactly the properties in S. We are often interested in the number  $f_{\equiv}(\emptyset)$  or  $f_{\equiv}(A)$  of elements satisfying none of the given properties.

Theorem 4.4.2 has a simple linear algebraic interpretation. Consider the two  $2^A \times 2^A$  matrices C, D whose non-zero entries are  $C_{S,T} = 1$  for  $S \subseteq T$ , and  $D_{S,T} = (-1)^{|T-S|}$  for  $S \subseteq T$ . Then the inclusion-exclusion formula is equivalent to the assertion that C and D are inverse matrices. This

can be proved directly, but we prefer to deduce it as a special case of the Möbius inversion formula (Theorem 4.5.2). We now present two applications.

**Derangements**. One of the classic applications of inclusion-exclusion principle is the enumeration of the **derangements** of [n], which are the permutations  $\pi \in S_n$  such that  $\pi(i) \neq i$  for all i. Let  $A = \{A_1, \ldots, A_n\}$  where  $A_i$  is the property that  $\pi(i) = i$ . Then  $f_{\geq}(T) = (n - |T|)!$ , so the number  $D_n$  of derangements of [n] is

$$D_n = f_{\pm}(\emptyset) = \sum_T (-1)^{|T|} f_{\geq}(T) = \sum_{k=0}^n \binom{n}{k} (-1)^k (n-k)! = n! \left(\frac{1}{0!} - \frac{1}{1!} + \frac{1}{2!} - \dots \pm \frac{1}{n!}\right).$$

It follows that  $D_n$  is the integer closest to n!/e.

**Discrete derivatives.** Consider the k-vector space  $\Gamma$  of functions  $f : \mathbb{Z} \to \mathbb{k}$ . The **discrete derivative** of f is the function  $\Delta f$  given by  $\Delta f(n) = f(n+1) - f(n)$ . We now wish to show that, just as with ordinary derivatives,

 $\Delta^{d+1}f = 0$  if and only if f is a polynomial of degree at most d.

Recall that this was part of Theorem 2.4.2. Regarding  $\Delta$  as a linear operator on  $\Gamma$ , we have  $\Delta = E - 1$  where Ef(n) = f(n+1) and 1 is the identity. Then  $\Delta^k = (E-1)^k = \sum_{i=0}^k {k \choose i} E^i (-1)^{k-i}$ , so the k-th discrete derivative is  $\Delta^k f(n) = \sum_{i=0}^k (-1)^{k-i} {k \choose i} f(n+i)$ .

so the k-th discrete derivative is  $\Delta^k f(n) = \sum_{i=0}^k (-1)^{k-i} {k \choose i} f(n+i)$ . The functions  $f_{\leq}(S) = f(n+|S|)$  and  $f_{=}(S) = \Delta^k f(|S|)$  satisfy Theorem 4.4.2.2, so we have  $f(n+k) = \sum_{i=0}^k {k \choose i} \Delta^i f(n)$ . (This is equivalent to  $E^k = (\Delta+1)^k$ .) If  $\Delta^{d+1}f = 0$ , this gives  $f(k) = \sum_{i=0}^d {k \choose i} \Delta^i f(0)$  which is a polynomial in k of degree at most d. The converse follows from the observation that  $\Delta$  lowers the degree of a polynomial by 1.

## 4.5 Möbius functions and Möbius inversion

# 4.5.1 The Möbius function.

Given a locally finite poset P, let  $Int(P) = \{[x, y] : x, y \in P, x \leq y\}$  be the set of intervals of P. The (two-variable) **Möbius function** of a poset P is the function  $\mu : Int(P) \to \mathbb{Z}$  defined by

$$\sum_{p \le r \le q} \mu(p, r) = \begin{cases} 1 & \text{if } p = q, \\ 0 & \text{otherwise.} \end{cases}$$
(19)

Here we are denoting  $\mu(p,q) = \mu([p,q])$ . We will later see that the Möbius function can be defined equivalently by the equations:

$$\sum_{p \le r \le q} \mu(r, q) = \begin{cases} 1 & \text{if } p = q, \\ 0 & \text{otherwise.} \end{cases}$$
(20)

When P has a minimum element  $\hat{0}$ , the (one-variable) **Möbius function**  $\mu : P \to \mathbb{Z}$  is  $\mu(x) = \mu(\hat{0}, x)$ . If P also has a  $\hat{1}$ , the **Möbius number** of P is  $\mu(P) = \mu(\hat{0}, \hat{1})$ .

Computing the Möbius function is a very important problem, because the Möbius function is the poset analog of a derivative; and as such, it is a fundamental invariant of a poset. This problem often leads to very interesting enumerative combinatorics, as can be gleaned from the following gallery of Möbius functions. **Theorem 4.5.1.** The Möbius functions of some key posets are as follows.

1. (Chain) P = n:

$$\mu_{\mathbf{n}}(i,j) = \begin{cases} 1 & \text{if } j = i \\ -1 & \text{if } j = i+1 \\ 0 & \text{otherwise.} \end{cases}$$

2. (Boolean lattice)  $P = 2^A$ :

$$\mu_{2^A}(S,T) = (-1)^{T-S}.$$

3. (Divisor lattice)  $P = D_n$ : We have  $\mu_{D_n}(k, l) = \mu(l/k)$  where

$$\mu(m) = \begin{cases} (-1)^t & \text{if } m \text{ is a product of } t \text{ distinct primes, and} \\ 0 & \text{otherwise.} \end{cases}$$

is the classical Möbius function from number theory.

4. (Young's lattice) P = Y:

$$\mu(\lambda,\mu) = \begin{cases} (-1)^{|\mu-\lambda|} & \text{if } \mu-\lambda \text{ has no two adjacent squares, and} \\ 0 & \text{otherwise.} \end{cases}$$

5. (Partition lattice)  $P = \prod_n$ : The Möbius number of  $\prod_n$  is

$$\mu(\Pi_n) = (-1)^{n-1}(n-1)!,$$

from which a (less elegant) formula for the complete Möbius function can be derived.

6. (Non-crossing partition lattice)  $P = NC_n$ : The Möbius number of  $NC_n$  is

$$\mu(NC_n) = (-1)^{n-1}C_{n-1},$$

where  $C_{n-1}$  is the (n-1)-st Catalan number. This gives a (less elegant) formula for the complete Möbius function.

7. (Bruhat order)  $P = S_n$ :

$$\mu(u, v) = (-1)^{\ell(v) - \ell(u)},$$

where the **length**  $\ell(w)$  of a permutation  $w \in S_n$  is the number of **inversions** (i, j) where  $1 \leq i < j \leq n$  and  $w_i > w_j$ . (There is a generalization of this result to the Bruhat order on any Coxeter group W, or even on a parabolic subgroup  $W^J$ ; see Section 4.7.)

8. (Subspace lattice)  $P = L(\mathbb{F}_q^n)$ : For the subspace lattice of a finite vector space  $\mathbb{F}_q^n$ ,

$$\mu(U,V) = (-1)^d q^{\binom{d}{2}},$$

where  $d = \dim V - \dim U$ .

9. (Distributive lattice) L = J(P):

$$\mu(I,J) = \begin{cases} (-1)^{|J-I|} & \text{if } J - I \text{ is an antichain in } P, \text{ and} \\ 0 & \text{otherwise.} \end{cases}$$

10. (Face poset of a polytope) L = F(P):

$$\mu(F,G) = (-1)^{\dim G - \dim F}$$

11. (Face poset of a subdivision  $\mathcal{T}$  of a polytope P)  $L = \widehat{T}$ :

$$\mu(F,G) = \begin{cases} (-1)^{\dim G - \dim F} & \text{if } G < \widehat{1} \\ (-1)^{\dim P - \dim F + 1} & \text{if } G = \widehat{1} \text{ and } F \text{ is not on the boundary of } P \\ 0 & \text{if } G = \widehat{1} \text{ and } F \text{ is on the boundary of } P \end{cases}$$

12. (Subgroup lattice of a finite p-group) If  $|G| = p^n$  for p prime,  $n \in \mathbb{N}$ , then in L = L(G):

$$\mu(A,B) = \begin{cases} (-1)^k p^{\binom{k}{2}} & \text{if } A \text{ is a normal subgroup of } B \text{ and } B/A \cong \mathbb{Z}_p^k, \text{ and} \\ 0 & \text{otherwise.} \end{cases}$$

Some of the formulas above follow easily from the definitions, while others require more sophisticated methods. In the following sections we will develop some of the basic theory of Möbius functions and discuss the most common methods for computing them. Along the way, we will sketch proofs of all the formulas above.

It is worth remarking that a version of 11. holds more generally for the face poset of any finite regular cell complexes  $\Gamma$  such that the underlying space  $|\Gamma|$  is a manifold with or without boundary; see [Sta12, Prop. 3.8.9] for details.

#### 4.5.2 Möbius inversion

In enumerative combinatorics, there are many situations where have a set U of objects, and a natural way of assigning to each object u of U an element f(u) of a poset P. We are interested in counting the objects in U that map to a particular element  $p \in P$ . Often we find that it is much easier to count the objects in s that map to an element less than or equal to<sup>8</sup> p in P. The following theorem tells us that this easier enumeration is sufficient for our purposes, as long as we can compute the Möbius function of P.

**Theorem 4.5.2.** (Möbius Inversion formula) Let P be a poset and let  $f, g : P \to \Bbbk$  be functions from P to a field  $\Bbbk$ . Then

1. 
$$g(p) = \sum_{q \ge p} f(q)$$
 for all  $p \in P$   $\iff$   $f(p) = \sum_{q \ge p} \mu(p,q)g(q)$  for all  $p \in P$ , and  
2.  $g(p) = \sum_{q \le p} f(q)$  for all  $p \in P$   $\iff$   $f(p) = \sum_{q \le p} \mu(q,p)g(q)$  for all  $p \in P$ .

<sup>&</sup>lt;sup>8</sup>or greater than or equal to

In his paper [Rot64], which pioneered the use of the Möbius inversion formula as a tool for counting in combinatorics, Rota described this enumerative philosophy as follows:

It often happens that a set of objects to be counted possesses a natural ordering, in general only a partial order. It may be unnatural to fit the enumeration of such a set into a linear order such as the integers: instead, it turns out in a great many cases, that a more effective technique is to work with the natural order of the set. One is led in this way to set up a "difference calculus" relative to an arbitrary partially ordered set.

Indeed, one may think of the Möbius function as a poset-theoretic analog of the Fundamental Theorem of Calculus: g is analogous to the integral of f, as it stores the cumulative value of this function. Like the Fundamental Theorem of Calculus, the Möbius inversion formula tells us how to recover the function f from its cumulative values.

It is not difficult to prove Theorem 4.5.2 directly, but we will soon discuss an algebraic framework that really explains it. In the meantime, we discuss two key applications. We will see several other applications later on.

Example 4.5.3. Möbius inversion is particularly important for chains, Boolean and divisor lattices.

1. For  $P = \mathbb{N}$ , Theorem 4.5.2.1 is a simple but important result for partial sums:

$$g(n) = \sum_{i=0}^{n} f(i) \text{ for } n \in \mathbb{N} \text{ if and only if } f(n) = g(n) - g(n-1) \text{ for } n \in \mathbb{N}$$

where g(-1) = 0.

2. For  $P = 2^A$ , Theorem 4.5.2.1 is the inclusion-exclusion formula:

$$g(S) = \sum_{T \supseteq S} f(T) \text{ for } S \subseteq A \text{ if and only if } f(S) = \sum_{T \supseteq S} (-1)^{|T-S|} g(T) \text{ for } S \subseteq A.$$

3. For  $P = D_n$ , Theorem 4.5.2.2 is the Möbius inversion formula from number theory:

$$g(m) = \sum_{m|d|n} f(d) \quad \text{for } m|n \quad \text{ if and only if } \quad f(m) = \sum_{m|d|n} \mu(d/m)g(d) \quad \text{for } m|n.$$

A typical application is the computation of Euler's totient function

 $\varphi(n) = |\{u : 1 \le u \le n, \ \gcd(u, n) = 1\}|,$ 

or more generally  $f(m) = |\{u : 1 \le u \le n, gcd(u, n) = m\}|$  for m|n. Here U = [n], and we assign to each  $u \in U$  the divisor gcd(n, u) of n. There are g(m) = n/m multiples of m in U, so

$$\varphi(n) = f(1) = \sum_{d|n} \mu(d) \frac{n}{d} = n \left(1 - \frac{1}{p_1}\right) \cdots \left(1 - \frac{1}{p_k}\right)$$

We will be interested in applying the Möbius inversion formula in many other contexts, and for that reason it is important that we gain a deeper understanding of Möbius functions, and develop techniques to compute them. This is one of the main goals of the following sections.
#### 4.6 The incidence algebra.

The Möbius function has a very natural algebraic interpretation, which we discuss in this section. Given a field k, recall that a k-algebra A is a vector space over k equipped with a bilinear product.

The **incidence algebra** I(P) of a locally finite poset P is the k-algebra of functions f: Int $(P) \rightarrow k$  from the intervals of P to k, equipped with the **convolution product**  $f \cdot g$  given by

$$f \cdot g(p,r) = \sum_{p \le q \le r} f(p,q) g(q,r) \quad \text{for } p \le r$$

Alternatively, let Mat(P) be the set of  $P \times P$  matrices A with entries in  $\Bbbk$  whose only nonzero entries  $a_{pq} \neq 0$  occur in positions where  $p \leq q$  in P. There is no canonical way of listing the rows and columns of the matrix P in a linear order. We normally list them in the order given by a linear extension of P, so that the matrices in Mat(P) will be upper triangular. It is clear from the definitions that

$$I(P) \cong \operatorname{Mat}(P).$$

The unit and inverses. The product in I(P) is clearly associative, and has a unit

$$\mathbf{1}(p,q) = \begin{cases} 1 & \text{if } p = q, \\ 0 & \text{if } p < q. \end{cases}$$

which is a (two-sided) multiplicative identity. An element  $f \in I(P)$  has a (necessarily unique) left and right multiplicative inverse  $f^{-1}$  if and only if  $f(p, p) \neq 0$  for all  $p \in P$ .

The zeta function and counting chains. An important element of I(P) is the zeta function

$$\zeta(p,q) = 1$$
 for all  $p \le q$ .

Notice that  $\zeta^k(p,q)$  is the number of multichains of length k from p to q.

If P has a  $\widehat{0}$  and  $\widehat{1}$ , then  $\zeta_P^k(\widehat{0},\widehat{1})$  counts all multichains of length k-2 in P, so the zeta polynomial and zeta function are related by

$$Z_P(k) = \zeta_P^k(\widehat{0}, \widehat{1}) \qquad \text{for } k \ge 2.$$
(21)

Similarly,  $(\zeta - \mathbf{1})^k(p, q)$  is the number of chains of length k from p to q. The sum  $\mathbf{1} + (\zeta - \mathbf{1}) + (\zeta - \mathbf{1})^2 + \cdots$  has finitely many non-zero terms, and it equals  $(\mathbf{2} - \zeta)^{-1}$ , where  $\mathbf{2} = 2 \cdot \mathbf{1}$ ; so

 $(\mathbf{2}-\zeta)^{-1}(p,q) =$  total number of chains in P from p to q.

The Möbius function and Möbius inversion. The (equivalent) equations (19) and (20) defining the Möbius function can be rewritten as  $\mu \zeta = \mathbf{1}$  and  $\zeta \mu = \mathbf{1}$ , respectively. This explains why these two equations are equivalent: they say that

$$\mu = \zeta^{-1}.$$

Proof of the Möbius inversion formula. Consider the left action of the incidence algebra I(P) on the vector space  $\mathbb{k}^P$  of functions  $f: P \to \mathbb{k}$ , given by

$$a \cdot f(p) = \sum_{q \ge p} a(p,q) f(q)$$

for  $a \in I(P)$  and  $f: P \to k$ . The Möbius inversion formula then states that  $g = \zeta \cdot f$  if and only if  $\mu \cdot g = f$ ; this follows immediately from  $\mu = \zeta^{-1}$ .

# 4.7 Computing Möbius functions.

To apply the Möbius inversion Formula successfully, it is often necessary to compute the Möbius functions of the posets involved. In this section, we discuss some of the main tools for computing Möbius functions. Along the way, we prove the formulas for the Möbius functions of the posets of Theorem 4.5.1.

1. (Chain **n**) We can check the formula for  $\mu_{\mathbf{n}}$  manually.

**Möbius functions of products.** A simple but important fact is that Möbius functions behave well under poset multiplication:

$$\mu_{P \times Q}((p,q), (p',q')) = \mu_P(p,p')\mu_Q(q,q')$$

This is easily verified directly, and also follows from the fact that  $I(P \times Q) \cong I(P) \otimes_{\mathbb{K}} I(Q)$ .

- 2. (Boolean lattice  $2^A$ ) Since  $2^A \cong \mathbf{2} \times \cdots \times \mathbf{2}$ , the Möbius function of  $\mathbf{2}$  tells us that the Möbius function of  $2^A$  is  $\mu(S,T) = (-1)^{|T-S|}$ .
- 3. (Divisor lattice  $D_n$ ) Since  $D_n \cong (\mathbf{t_1} + \mathbf{1}) \times \cdots \times (\mathbf{t_k} + \mathbf{1})$  for  $n = p_1^{t_1} \cdots p_k^{t_k}$ , we get the formula for  $\mu_{D_n}$  from the formula for the Möbius function of a chain.

**Möbius functions through Möbius inversion.** So far we have thought of the Möbius function as a tool to apply Möbius inversion. Somewhat counterintuitively, it is possible to use Möbius inversion in the other direction, as a tool to compute Möbius functions. We carry out this approach to compute the Möbius function of the partition lattice  $\Pi_n$ .

5. (Partition lattice  $\Pi_n$ ) Let  $\Pi_A$  denote the lattice of set partitions of a set A ordered by refinement. Every interval  $[\pi, \rho]$  in  $\Pi_A$  is a product of partition lattices, as illustrated by the following example: in  $\Pi_9$  we have  $[18|2|37|4|569, 13478|2569] \cong \Pi_{\{18,37,4\}} \times \Pi_{\{2,569\}} \cong \Pi_3 \times \Pi_2$ . Since the Möbius function is multiplicative, to compute the Möbius function of the partition lattices it suffices to show that  $\mu_{\Pi_n}(\widehat{0}, \widehat{1}) = (-1)^{n-1}(n-1)!$ .

Let W be the set of words  $w = w_1 \dots w_n$  of length n in the alphabet  $\{0, 1, \dots, q-1\}$ . Classify the words according to the equalities among their coordinates; namely, to each word w, associate the partition of [n] where i and j are in the same block when  $w_i = w_j$ .

Let  $f(\pi)$  be the number of words whose partition is  $\pi$ , and let  $g(\pi)$  be the number of words whose partition is a coarsening of  $\pi$ . This is a situation where both  $f(\pi)$  and  $g(\pi)$  are easily computed: if  $\pi$  has b blocks then we have

$$f(\pi) = q(q-1)\cdots(q-b+1), \qquad g(\pi) = q^b.$$

Since  $g(\pi) = \sum_{\rho \ge \pi} f(\rho)$  we get  $f(\pi) = \sum_{\rho \ge \pi} \mu(\pi, \rho) g(\rho)$ . For  $\pi = \widehat{0}$  this says that  $q(q-1)\cdots(q-n+1) = \sum_{\rho \in \Pi_n} \mu(\widehat{0}, \rho) q^{|\rho|}$ .

Equating the coefficients of  $q^1$  we get the desired result.

Möbius functions through closures. A function  $\overline{\cdot}: P \to P$  is a closure operator if

- $p \leq \overline{p}$  for all  $p \in P$ ,
- $p \leq q$  implies  $\overline{p} \leq \overline{q}$  for all  $p, q \in P$ , and
- $\overline{\overline{p}} = \overline{p}$  for all  $p \in P$ .

An element of p is closed if  $\overline{p} = p$ ; let Cl(P) be the subposet of closed elements of P.

**Proposition 4.7.1.** If  $\overline{\cdot}: P \to P$  is a closure operator, then for any  $p \leq q$  in P,

$$\sum_{r:\bar{r}=q} \mu(p,r) = \begin{cases} \mu_{Cl(P)}(p,q) & \text{if } p \text{ and } q \text{ are closed} \\ 0 & \text{otherwise} \end{cases}$$

*Proof.* We have an inclusion of incidence algebras  $I(Cl(P)) \to I(P)$  given by  $f \mapsto \overline{f}$  where

$$\overline{f}(p,q) = \begin{cases} f(p,q) & \text{if } p \text{ and } q \text{ are closed} \\ 0 & \text{otherwise} \end{cases}$$

Let  $\overline{1}, \overline{\zeta}$ , and  $\overline{\mu}$  be the image of the unit, zeta, and Möbius functions of Cl(P) in I(P). Also consider the "closure function"  $c \in I(P)$  whose non-zero values are  $c(p, \overline{p}) = 1$  for all p. Note that  $c\overline{\zeta} = \zeta \overline{1}$ because when q is closed,  $\overline{p} \leq q$  if and only if  $p \leq q$ . Now we compute in I(P):

$$\mu c \overline{1} = \mu c \overline{\zeta} \overline{\mu} = \mu \zeta \overline{1} \overline{\mu} = 1 \overline{\mu} = \overline{\mu}$$

This is equivalent to the desired equality.

**Möbius functions of lattices.** When L is a lattice, there are two other methods for computing Möbius functions. The first one gives an alternative to the defining recursion for  $\mu(p,q)$ . This new recurrence is usually much shorter, at the expense of requiring some understanding of the join operation.

**Proposition 4.7.2.** (Weisner's Theorem) For any  $p < a \leq q$  in a lattice L we have

$$\sum_{p \leq r \leq q \, : \, r \vee a = q} \mu(p,r) = 0$$

Weisner's Theorem follows from Proposition 4.7.1 because  $r \mapsto r \lor a$  is a closure operation, whose closed sets are the elements greater than or equal to a. There is also a dual version, obtained by applying the result to the reverse lattice  $L^{\text{op}}$  obtained from L by reversing all order relations.

Let us apply Weisner's Theorem to three examples.

8. (Subspace lattice  $L(\mathbb{F}_q^n)$ ) Since an interval of height r in  $L(\mathbb{F}_q^n)$  is isomorphic to  $L(\mathbb{F}_q^r)$ , it suffices to compute  $\mu(L(\mathbb{F}_q^n)) =: \mu_n$ . Let  $p = \hat{0}, q = \hat{1}$ , and let a be any line. The only subspaces  $r \neq \hat{1}$  with  $r \lor a = \hat{1}$  are the  $q^{n-1}$  hyperplanes not containing a. Each one of these hyperplanes satisfies  $[\hat{0}, r] \cong L(\mathbb{F}_q^{n-1})$ , so Weisner's Theorem gives  $\mu_n + q^{n-1}\mu_{n-1} = 0$ , from which  $\mu_n = (-1)^n q^{n(n-1)/2}$ .

- 5. (Partition lattice  $\Pi_n$ , revisited) A similar (and easier) argument may be used to compute  $\mu_n = \mu(\Pi_n)$ , though it is now easier to use the dual to Weisner's Theorem. Let  $p = \hat{0}$ ,  $q = \hat{1}$ , and let a be the coatom  $12 \dots n 1 | n$ . The only partitions  $r \neq \hat{0}$  with  $r \wedge a = \hat{0}$  are those with only one non-singleton block of the form  $\{i, n\}$  for  $i \neq n$ . Each such partition  $\pi$  has  $[\pi, \hat{1}] \cong \Pi_{n-1}$ . The dual to Weisner's Theorem tells us that  $\mu_n + (n-1)\mu_{n-1} = 0$ , from which  $\mu_n = (-1)^n (n-1)!$ .
- 11. (Subgroup lattice L(G) of a finite p-group G) A similar argument may be used for the subgroup lattice L(G) of a p-group G. However, now one needs to invoke some facts about p-groups; see [Wei35].

**Theorem 4.7.3.** (Crosscut Theorem) Let L be a lattice and let X be the set of atoms of L. Then

$$\mu(\widehat{0},\widehat{1}) = \sum_{k} (-1)^k N_k$$

where  $N_k$  is the number of k-subsets of X whose join is  $\hat{1}$ .

We will sketch a proof at the end of this section. Meanwhile, we point out a simple corollary of the Crosscut Theorem:

If the join of the atoms is not 
$$\hat{1}$$
 then  $\mu(0, \hat{1}) = 0.$  (22)

- 9. (Distributive lattice L = J(P)) This result suffices to compute the Möbius function of a distributive lattice L = J(P). If J I is an antichain, then [I, J] is a Boolean lattice and  $\mu_L(I, J) = (-1)^{J-I}$ . Otherwise, the join of the atoms of [I, J] is  $I \cup \min(J I) \neq J$ , and hence  $\mu_L(I, J) = 0$ .
- 4. (Young's lattice Y) We obtain this Möbius function for free since Y is distributive.

Naturally, there are dual formulations to the previous two propositions, obtained by reversing the order of L. Also, there are many different versions of the crosscut theorem; see for example [Rot64].

**Möbius functions through multichains.** If we know the zeta polynomial of a poset P with  $\hat{0}$  and  $\hat{1}$ , we can obtain its Möbius number  $\mu(P)$  immediately.

**Proposition 4.7.4.** If P is a poset with  $\hat{0}$  and  $\hat{1}$ ,

$$Z_P(-1) = \mu_P(\widehat{0}, \widehat{1}).$$
 (23)

*Proof.* We saw in (21) that  $Z_P(k) = \zeta_P^k(\widehat{0}, \widehat{1})$  for all integers  $k \ge 2$ . It would be irresponsible to just set k = -1, but it is very tempting, since  $\zeta^{-1} = \mu$ .

In the spirit of **combinatorial reciprocity**, this is an instance where such irresponsible behavior pays off, with a bit of extra care. We know that  $Z_P(k)$  is polynomial for  $k \in \mathbb{Z}$ , and we leave it as an exercise to show that  $\zeta^k(\widehat{0}, \widehat{1})$  is also polynomial for  $k \in \mathbb{Z}$ . Since these two polynomials agree on infinitely many values, they also agree for k = -1, and the result follows. In light of (23), the zeta polynomial of P will give us the Möbius number  $\mu(P)$  automatically. This is advantageous because sometimes the zeta polynomial is easier to compute than the Möbius function, as it is the answer to an explicit enumerative question.

6. (Non-crossing partition lattice  $NC_n$ ) From the zeta polynomial of  $NC_n$  in Example 4.3.1 we immediately obtain that  $\mu(NC_n) = (-1)^{n-1}C_{n-1}$  where  $C_{n-1}$  is the (n-1)st Catalan number. This gives a formula for the full Möbius function, since every interval in  $NC_n$  is a product of smaller non-crossing partition lattices.

**Möbius functions through topology.** Equation (23) has a topological interpretation which is an extremely powerful method for computing Möbius functions.

**Proposition 4.7.5.** (Phillip Hall's Theorem) Let P be a finite poset with a  $\hat{0}$  and  $\hat{1}$ , and let  $c_i$  be the number of chains  $\hat{0} = p_0 < p_1 < \cdots < p_i = \hat{1}$  of length i from  $\hat{0}$  to  $\hat{1}$  in P. Then

$$\mu_P(\widehat{0},\widehat{1}) = c_0 - c_1 + c_2 - \cdots$$

This formula is equivalent to (23) in light of (18) and the relations  $b_i = c_i + 2c_{i-1} + c_{i-2}$ ; it may also be proved directly in the incidence algebra of P.

Let us now interpret this result topologically. The order complex  $\Delta(P)$  of a poset P is the simplicial complex whose vertices are the elements of P, and whose faces are the chains of P.

**Theorem 4.7.6.** Let P be a finite poset with a  $\hat{0}$  and  $\hat{1}$ , and let  $\overline{P} = P - \{\hat{0}, \hat{1}\}$ . Then

$$\mu_P(\widehat{0},\widehat{1}) = \widetilde{\chi}(\Delta(\overline{P}))$$

is the reduced Euler characteristic of  $\Delta(\overline{P})$ .

Proof. In light of Proposition 4.7.5, this follows immediately from the combinatorial formula

$$\widetilde{\chi}(\Delta) = \sum_{k=-1}^{d} (-1)^k f_k$$

for the Euler characteristic of  $|\Delta|$ , where  $f_k$  is the number of k-dimensional faces of  $\Delta$ .

Let us use this topological description to sketch proofs of the remaining Möbius functions of Theorem 4.5.1.

10. (Face lattice of a polytope L(P)) The **barycentric subdivision** sd(P) is a simplicial complex with a vertex v(F) at the barycenter of each proper face F. It has a simplex connecting vertices  $v(F_1), \ldots, v(F_k)$  whenever  $F_1 \subset \cdots \subset F_k$ . As abstract simplicial complexes, sd(P)equals  $\Delta(\overline{F(P)})$ . Geometrically, sd(P) is a subdivision of the boundary of the polytope P, and hence is homeomorphic to the sphere  $\mathbb{S}^{\dim P-1}$ . Therefore

$$\mu(\widehat{0},\widehat{1})=\widetilde{\chi}(\Delta(\overline{P}))=\widetilde{\chi}(\mathbb{S}^{\dim P-1})=(-1)^{\dim P}.$$

We will see in Section 5 that every interval of F(P) is itself the face poset of a polytope. It then follows that  $\mu(F, G) = (-1)^{\dim G - \dim F}$ .

- 11. (Face lattice of a subdivision  $\mathcal{T}$  of a polytope P) A similar argument holds, though the details are slightly more subtle; see [Sta12, Prop. 3.8.9].
- 7. (Bruhat order in permutations  $S_n$ ) There are several known proofs of the fact that  $\mu(u, v) = (-1)^{l(u)-l(v)}$ , none of which is easy. The first proof was an ad hoc combinatorial due to Verma [Ver71]. Later Kazhdan and Lusztig [KL79] and Stembridge [Ste07] gave algebraic proofs using Kazhdan-Lusztig polynomials and Hecke algebras, respectively. Björner and Wachs [BW82] gave a topological proof, based on Theorem 4.7.6 and further tools from topological combinatorics. As we remarked earlier, there are similar formulas for the Möbius function of an arbitrary parabolic quotient of an arbitrary Coxeter group; see [BB05] or [Ste07].

Theorem 4.7.6 tells us that in order to compute Möbius functions of posets of interest, it can be very useful to understand the topology of their underlying order complexes. Conversely, combinatorial facts about Möbius functions may lead to stronger statements about topological spaces. This is the motivation for the very rich study of **poset topology**. We refer the reader to the survey [Wac07] for further information on this topic.

## 4.8 Eulerian posets, flag *f*-vectors and *h*-vectors.

Let P be a graded poset of height r. The flag f-vector  $(f_S : S \subseteq [0, r])$  and the flag h-vector  $(h_S : S \subseteq [0, r])$  are defined by

 $f_S$  = number of chains  $p_1 < p_2 < \dots < p_k$  with  $\{r(p_1), \dots, r(p_k)\} = S$ , and

$$h_S = \sum_{T \subseteq S} (-1)^{|S| - |T|} f_T, \qquad f_S = \sum_{T \subseteq S} h_T$$

for  $S \subseteq [0, r]$ .

**Example 4.8.1.** Let H be a hexagonal prism of Figure 4.4, and let  $P = L(H) - \{\widehat{0}, \widehat{1}\}$  be its face lattice, with the top and bottom element (the empty face and the full face H) removed. It is not so enlightening to draw the poset, but we can still compute its flag f and h-vectors. For example,  $f_{\{0,2\}} = 36$  counts the pairs (v, f) of a vertex v contained in a 2-face f. We get:

	Ø	0	1	2	01	02	12	012
$f_S$	1	12	18	8	36	36	36	72
$h_S$	1	11	17	$\gamma$	$\gamma$	17	11	1



Figure 4.4: A hexagonal prism.

This example suggests that there may be additional structure in the flag *h*-vector; for instance in this example we have  $h_S = h_{[0,d]-S}$  for all S. This equation is not true in general, but it does hold for an important class of posets, which we now discuss.

**Eulerian posets.** Say a graded poset P is **Eulerian** if  $\mu(x, y) = (-1)^{r(y)-r(x)}$  for all  $x \leq y$ .

In Theorem 4.5.1 we saw three important families of Eulerian posets: Boolean lattices, the Bruhat order, and face posets of polytopes. In particular, the poset P of Example 4.8.1 is Eulerian.

For an Eulerian poset, there are many linear relations among the  $f_S$ s, which are easier to describe in terms of the  $h_S$ s; for example,  $h_S = h_{[0,d]-S}$  for all S. To describe them all, we further encode the flag f-vector in a polynomial in non-commuting variables **a** and **b** called the **ab-index**, which is defined to be

$$\Phi_P[\mathbf{a}, \mathbf{b}] = \sum_{S \subseteq [d]} h_S u_S$$

where  $u_S = u_1 \dots u_d$  and  $u_i = \mathbf{a}$  if  $i \notin S$  and  $u_i = \mathbf{b}$  if  $i \in S$ . This is an element of the ring  $\mathbb{Z}\langle \mathbf{a}, \mathbf{b} \rangle$  of integer polynomials in the non-commutative variables  $\mathbf{a}$  and  $\mathbf{b}$ .

**Theorem 4.8.2.** [BB85, BK91, Sta94] The **ab**-index of an Eulerian poset P can be expressed uniquely as a polynomial in

$$\mathbf{c} = \mathbf{a} + \mathbf{b}, \qquad \mathbf{d} = \mathbf{a}\mathbf{b} + \mathbf{b}\mathbf{a}$$

called the cd-index  $\Psi_P(\mathbf{c}, \mathbf{d})$  of *P*. Furthermore, if *P* is the face poset of a polytope, then the coefficients of the cd-index are non-negative.

Example 4.8.3. The ab-index of the hexagonal prism in Example 4.8.1 is

$$\Phi(\mathbf{a}, \mathbf{b}) = \mathbf{a}\mathbf{a}\mathbf{a} + 11\mathbf{b}\mathbf{a}\mathbf{a} + 17\mathbf{a}\mathbf{b}\mathbf{a} + 7\mathbf{a}\mathbf{a}\mathbf{b} + 7\mathbf{b}\mathbf{b}\mathbf{a} + 17\mathbf{b}\mathbf{a}\mathbf{b} + 11\mathbf{a}\mathbf{b}\mathbf{b} + \mathbf{b}\mathbf{b}\mathbf{b}$$
  
=  $(\mathbf{a} + \mathbf{b})^3 + 6(\mathbf{a} + \mathbf{b})(\mathbf{a}\mathbf{b} + \mathbf{b}\mathbf{a}) + 10(\mathbf{a}\mathbf{b} + \mathbf{b}\mathbf{a})(\mathbf{a} + \mathbf{b})$ 

so the cd-index is

$$\Psi(\mathbf{c}, \mathbf{d}) = \mathbf{c}^3 + 6\mathbf{c}\mathbf{d} + 10\mathbf{d}\mathbf{c}.$$

In general, the 8 entries of the flag f-vector of a 3-polytope (or any other Eulerian poset of rank 3) are determined completely by only 3 numbers, namely, the coefficients of  $\mathbf{c}^3$ ,  $\mathbf{cd}$ , and  $\mathbf{dc}$  in the  $\mathbf{cd}$ -index.

The **cd**-index encodes optimally the linear relations among the entries of flag f-vectors of polytopes. Since the number of monomials in **c** and **d** of **ab**-degree d is the Fibonacci number  $F_{d+1}$ , this is the smallest number of entries in  $(f_S)$  from which we can recover the whole flag f-vector – this is much smaller than  $2^d$ .

**Corollary 4.8.4.** [BB85, BK91] The subspace of  $\mathbb{R}^{2^d}$  spanned by the flag f-vectors of Eulerian posets of rank d (or by the flag f-vectors of d-polytopes) has dimension equal to the Fibonacci number  $F_{d+1}$ .

# PART 2. DISCRETE GEOMETRIC METHODS

Part 2 is devoted to discrete geometry, which studies the connections between combinatorics and the geometry of subspaces (points, lines, planes, ..., hyperplanes) in a Euclidean space. Configurations in the plane and 3-space have received great attention, and feature many interesting results and open questions; however, very few of them involve exact enumeration in a meaningful way. Instead, we will focus on the discrete geometry of higher dimensions, where:

• studying "general" geometric configurations leads to interesting enumerative questions, and

• we have enough room to construct "special" geometric configurations that model various combinatorial structures of interest.

This second part is divided into three chapters on closely interrelated topics. In Section 5 we discuss polytopes, which are the higher dimensional generalization of polygons. Section 6 discusses arrangements of hyperplanes in a vector space. Finally Section 7 is devoted to matroids, which are combinatorial objects that simultaneously abstract arrangements of vectors, graphs, and matching problems, among others.

# 5 **Polytopes**

The theory of polytopes is a vast area of study, with deep connections to pure (algebraic geometry, commutative algebra, representation theory) and applied mathematics (optimization). Again, we focus on aspects related to enumeration. For a general introduction to polytopes, see [Grü03, Zie95].

After discussing the basic definitions and facts in Section 5.1 and some important examples in Section 5.2, we turn to enumerative questions. Section 5.3 is devoted to the enumeration of faces of various dimensions. Section 5.4 is on *Ehrhart theory*, which measures polytopes by counting the lattice points that they (and their dilations) contain.

## 5.1 Basic definitions and constructions

Recall that a set S in Euclidean space  $\mathbb{R}^d$  is **convex** if for every pair of points u, v in S, the line segment uv is in S. The **convex hull** conv(S) of a set  $S \subseteq \mathbb{R}^d$  is the minimal convex set containing S. If  $S = \{\mathbf{v}_1, \ldots, \mathbf{v}_n\}$  then

$$\operatorname{conv}\{\mathbf{v}_1,\ldots,\mathbf{v}_n\} = \{\lambda_1\mathbf{v}_1+\cdots+\lambda_n\mathbf{v}_n : \lambda_1,\ldots,\lambda_n \ge 0, \lambda_1+\cdots+\lambda_n = 1\}$$
  
= intersection of all convex sets containing  $\mathbf{v}_1,\ldots,\mathbf{v}_n$ .

We will only be interested in convex polytopes, and when we talk about polytopes, it will be assumed that they are convex.

A hyperplane H in Euclidean space  $\mathbb{R}^d$  is an affine subspace of dimension d-1; it is given by a linear equation

 $H = \{ \mathbf{x} \in \mathbb{R}^d : \mathbf{a} \cdot \mathbf{x} = b \}$  for some  $\mathbf{a} \in \mathbb{R}^d - \{ \mathbf{0} \}$  and  $b \in \mathbb{R}$ .

It separates  $\mathbb{R}^d$  into the two halfspaces given by the inequalities  $\mathbf{a} \cdot \mathbf{x} \leq b$  and  $\mathbf{a} \cdot \mathbf{x} \geq b$ , respectively.

There are two equivalent ways of defining convex polytopes: the *V*-description gives a polytope in terms of its vertices, and the *H*-description gives it in terms of its defining inequalities.

The V-description. A convex polytope P is the convex hull of finitely many points  $\mathbf{v}_1, \ldots, \mathbf{v}_n$  in a Euclidean space  $\mathbb{R}^d$ :

$$P = \operatorname{conv}\{\mathbf{v}_1,\ldots,\mathbf{v}_n\}$$

The H-description. A convex polytope P is a bounded intersection of finitely many halfspaces in a Euclidean space  $\mathbb{R}^d$ :

$$P = \left\{ \mathbf{x} \in \mathbb{R}^d : \mathbf{a}_1 \cdot \mathbf{x} \le b_1, \dots, \mathbf{a}_m \cdot \mathbf{x} \le b_m \right\}$$
$$= \left\{ x \in \mathbb{R}^d : A\mathbf{x} \le \mathbf{b} \right\}$$

where  $\mathbf{a}_1, \ldots, \mathbf{a}_m \in \mathbb{R}^d$  and  $b_1, \ldots, b_m \in \mathbb{R}$ . In the second expression we think of  $\mathbf{x}$  as a column vector, A is the  $m \times d$  matrix with rows  $\mathbf{a}_1, \ldots, \mathbf{a}_m$  and  $\mathbf{b}$  is the column vector with entries  $b_1, \ldots, b_m$ .

**Theorem 5.1.1.** A subset  $P \subseteq \mathbb{R}^d$  is the convex hull of a finite set of points if and only if it is a bounded intersection of finitely many halfspaces.

For theoretical and practical purposes, it is useful to have **both** the V-description and the H-description of a polytope. For example, it is clear from the H-description (but not at all from the V-description) that the intersection of two polytopes is a polytope, and it is clear from the V-description (but not at all from the H-description) that a projection of a polytope is a polytope. It is a nontrivial task to translate one description into the other; see [Zie95, Notes to Chapter 1] for a discussion and references on this problem. Here are some simple examples.

**Example 5.1.2.** The standard simplex  $\Delta_{d-1}$ , the cube  $\Box_d$ , and the crosspolytope  $\Diamond_d$  are:

1. 
$$\Delta_{d-1} = \operatorname{conv} \{ \mathbf{e}_1, \dots, \mathbf{e}_d \}$$
  

$$= \{ \mathbf{x} \in \mathbb{R}^d : x_1 + \cdots x_d = 1 \text{ and } x_i \ge 0 \text{ for } i = 1, \dots, d \},$$
  
2. 
$$\Box_d = \operatorname{conv} \{ \pm \mathbf{e}_1 \pm \cdots \pm \mathbf{e}_d \text{ for any choice of signs} \}$$
  

$$= \{ \mathbf{x} \in \mathbb{R}^d : -1 \le x_i \le 1 \text{ for } i = 1, \dots, d \},$$
  
3. 
$$\Diamond_d = \operatorname{conv} \{ -\mathbf{e}_1, \mathbf{e}_1, \dots, -\mathbf{e}_d, \mathbf{e}_d \}$$
  

$$= \{ \mathbf{x} \in \mathbb{R}^d : \pm x_1 \pm \cdots \pm x_d \le 1 \text{ for any choice of signs} \},$$

respectively, where  $\mathbf{e}_1, \ldots, \mathbf{e}_d$  are the standard basis in  $\mathbb{R}^d$ . These polytopes are illustrated in Figure 5.1 for d = 3.



Figure 5.1: The triangle  $\Delta_2$ , the cube  $\Box_3$ , and the octahedron  $\diamond_3$ .

The **dimension** of a polytope P is the dimension of the affine subspace spanned by P:

aff(P) = 
$$\left\{\lambda_1 \mathbf{v}_1 + \dots + \lambda_k \mathbf{v}_k : \mathbf{v}_1, \dots, \mathbf{v}_k \in \mathbb{R}^d, \lambda_1, \dots, \lambda_k \in \mathbb{R}, \lambda_1 + \dots + \lambda_k = 0\right\}.$$

For example, even though  $\Delta_{d-1}$  lives in  $\mathbb{R}^d$ , its dimension is d-1 because it lies on the hyperplane  $x_1 + \cdots + x_d = 1$ .

The **interior** int(P) of a polytope  $P \subseteq \mathbb{R}^d$  is the topological interior of P. It is often more useful to consider the **relative interior** relint(P), which is its topological interior as a subset of the affine space aff(P).

**Polar polytopes**. The similarity of our descriptions of  $\Box_d$  and  $\Diamond_d$  is a manifestation of a general notion of duality between V-descriptions and H-descriptions of polytopes.

Let P be a polytope in  $\mathbb{R}^d$  such that  $\mathbf{0} \in \mathbf{int}(P)$ .<sup>9</sup> The **polar polytope**  $P^{\triangle}$  of P is

$$P^{\Delta} = \left\{ \mathbf{a} \in \mathbb{R}^d : \mathbf{a} \cdot \mathbf{x} \le 1 \text{ for all } \mathbf{x} \in P \right\}.$$

If P has vertices  $\mathbf{v}_1, \ldots, \mathbf{v}_n$  and facets  $\mathbf{a}_1 \cdot \mathbf{x} \leq 1, \ldots, \mathbf{a}_m \cdot \mathbf{x} \leq 1$  (and hence contains **0**), then the polar  $P^{\triangle}$  has vertices  $\mathbf{a}_1, \ldots, \mathbf{a}_m$  and facets  $\mathbf{v}_1 \cdot \mathbf{x} \leq 1, \ldots, \mathbf{v}_n \cdot \mathbf{x} \leq 1$ . It follows that if  $\mathbf{0} \in P$ , then

$$(P^{\triangle})^{\triangle} = P.$$

**Faces**. For each vector  $\mathbf{a} \in \mathbb{R}^d$ , let

$$P_{\mathbf{a}} = \{ \mathbf{x} \in P : \mathbf{a} \cdot \mathbf{x} \ge \mathbf{a} \cdot \mathbf{y} \text{ for all } \mathbf{y} \in P \}$$

be the subset of P where the linear function  $\mathbf{a} \cdot \mathbf{x}$  is maximized. Such a set is called a **face** of P. It is customary to allow the empty set to be a face of P as well. If  $\dim(P) = d$ , then the faces of dimension 1, 2, d - 2, d - 1 are called **vertices**, **edges**, **ridges**, and **facets**, respectively. We let V(P) be the set of vertices of P. Faces other than P are called **proper faces**. We collect a few basic facts about faces:

- A polytope is the convex hull of its vertices:  $P = \operatorname{conv}(V(P))$ .
- A polytope is the intersection of the halfspaces determined by its facets.
- The vertices of a face F of P are the vertices of P contained in  $F: V(F) = V(P) \cap F$ .
- A face F of P equals the intersection of the facets of P containing F.
- If F is a face of P, then any face of F is a face of P.
- The intersection of two faces of P is a face of P.

• A polytope is the disjoint union of the relative interiors of its faces:  $P = \bigcup_{F \text{ face}} \operatorname{relint}(F)$ .

The face lattice. The face lattice L(P) is the poset of faces of P, ordered by containment. It is indeed a lattice with  $F \wedge G = F \cap G$  and  $F \vee G = \operatorname{aff}(F \cup G) \cap P$ . It is graded with  $\operatorname{rk}(F) = \dim F + 1$ . We say that polytopes P and Q are combinatorially isomorphic if  $L(P) \cong L(Q)$ . We collect some basic properties of face lattices:

• For each face F of P, the interval [0, F] of L(P) is isomorphic to the face lattice of F.

• For each face F of P, the interval  $[F, \hat{1}]$  of L(P) is isomorphic to the face lattice of a polytope, called the **face figure** P/F of F.

 $<sup>^{9}</sup>$ We can apply an affine transformation to any full-dimensional polytope to make its interior contain the origin.

• Every interval [F, G] of L(P) is isomorphic to the face lattice of some polytope.

• The face lattice  $L(P^{\Delta})$  of the polar polytope  $P^{\Delta}$  is isomorphic to the opposite poset  $L(P)^{\text{op}}$ , obtained by reversing the order relations of L(P).

There are different constructions of the face figure P/F, giving rise to combinatorially isomorphic polytopes. We discuss one such construction. Let  $F^{\Diamond}$  be the face of  $P^{\triangle}$  corresponding to the face F of P under the isomorphism  $L(P^{\triangle}) \cong L(P)^{\text{op}}$ . Then we can define the face figure to be the polar polytope of  $F^{\Diamond}$ ; that is,  $P/F = (F^{\Diamond})^{\triangle}$ . When  $F = \mathbf{v}$  is a vertex, there is a more direct construction: let  $P/\mathbf{v} = P \cap H$ , where H is a hyperplane separating v from all other vertices of P.

We say a *d*-polytope P is **simplicial** if every face is a simplex. We say P is **simple** if every vertex is on exactly *d* facets (or, equivalently, on *d* edges). Note that the convex hull of generically chosen points is a simplicial polytope. Similarly, a bounded intersection of generically chosen half-spaces is a simple polytope. Also note that P is simplicial if and only if its polar  $P^{\Diamond}$  is simple.

**Triangulations and subdivisions**. In many contexts, it is useful to subdivide a polytope into simpler polytopes (most often simplices). It is most convenient to do it in such a way that the pieces of the subdivision meet face to face:

A subdivision of a polytope P is a finite collection  $\mathcal{T}$  of polytopes such that

- P is the union of the polytopes in  $\mathcal{T}$ ,
- if  $P \in \mathcal{T}$  then every face of P is in  $\mathcal{T}$ , and
- if  $P, Q \in \mathcal{T}$  then  $P \cap Q \in \mathcal{T}$ .

The elements of  $\mathcal{T}$  are called the **faces** of  $\mathcal{T}$ ; the full-dimensional ones are called **facets**. If all the faces are simplices, then  $\mathcal{T}$  is called a **triangulation** of P.

In some situations, it is useful to assume that a subdivision does not introduce new vertices; that is, that the only points in  $\mathcal{T}$  are the vertices of P. We will assume that throughout the rest of this chapter.

# **Theorem 5.1.3.** Every convex polytope has a triangulation.

Sketch of Proof. Let V be the set of vertices of our polytope  $P \in \mathbb{R}^d$ . For each **height function**  $h: V \to \mathbb{R}$  consider the set of lifted points  $P^h = \{(v, h(v)), v \in V\}$  in  $\mathbb{R}^{d+1}$ . Let Q be the convex hull of  $P^h$ , and consider the set  $\mathcal{F}$  of "lower facets" of Q that are visible from below, that is, the facets maximizing some linear function  $\mathbf{a} \cdot \mathbf{x}$  with  $a_{d+1} = -1$ . For each such facet F, let  $\pi(F)$ be its projection back down to  $\mathbb{R}^d$ . One may check that  $\{\pi(F) : F \text{ is a lower facet of } Q\}$  is a subdivision of P. If the height function is chosen generically, then this subdivision is actually a triangulation.

The subdivisions that may be obtained from the lifting construction above are called **regular** or **coherent**. In general, not every subdivision is regular, and the question of distinguishing regular and non-regular triangulations is subtle. In the simplest case, when P is a convex polygon, every subdivision is regular, and the number of subdivisions is a Catalan number, as we saw in Example 14b. of Section 2.2.2. Few other exact enumerative results of this sort are known.

In any case, regular subdivisions are of great importance for several reasons. One reason is that they are easy to define and construct; for example, choosing the heights  $h(v) = v_1^2 + \cdots v_d^2$  leads to the **Delaunay triangulation**, which is very easy to compute and has several desirable properties. Secondly, they have a very elegant structure. Every polytope P gives rise to a **secondary polytope**, whose faces are in bijection with the regular subdivisions of P; and faces F



Figure 5.2: A regular subdivision of a hexagon.

and G satisfy that  $F \subset G$  if and only if the subdivision of F refines the subdivision of G. Thirdly, they are widely applicable, in particular, because they play a key role in the theory of Gröbner bases in commutative algebra. For our enumerative purposes the regularity question will not be too important. For readers interested in this and other aspects of triangulations, we recommend [DLRS10, GKZ08, Stu96].

# 5.2 Examples

The following polytopes have particularly nice enumerative properties. We give references to some revelant results at the end of Section 5.4.

- 1. (Product of two simplices)  $\Delta_{c-1} \times \Delta_{d-1} = \operatorname{conv}\{(\mathbf{e}_i, \mathbf{f}_j) : 1 \le i \le c, 1 \le j \le d\}$  in  $\mathbb{R}^c \times \mathbb{R}^d$ where  $\{\mathbf{e}_1, \ldots, \mathbf{e}_c\}$  and  $\{\mathbf{f}_1, \ldots, \mathbf{f}_d\}$  are the standard bases for  $\mathbb{R}^c$  and  $\mathbb{R}^d$ , respectively.
- 2. (Hypersimplex)  $\Delta(r,d) = \operatorname{conv} \{ \mathbf{e}_{i_1} + \dots + \mathbf{e}_{i_r} : 1 \le i_1 < \dots < i_r \le d \}$  in  $\mathbb{R}^d$ .
- 3. (Permutahedron)  $\Pi_{d-1} = \operatorname{conv}\{(a_1, \ldots, a_d) : \{a_1, \ldots, a_d\} \text{ is a permutation of } [d]\}.$
- 4. (Zonotopes) The zonotope of a vector configuration  $A = \{\mathbf{a}_1, \ldots, \mathbf{a}_k\} \subset \mathbb{R}^d$  is the *Minkowski* sum  $Z(A) = [\mathbf{0}, \mathbf{a}_1] + \cdots + [\mathbf{0}, \mathbf{a}_k] := \{\lambda_1 \mathbf{a}_1 + \cdots + \lambda_k \mathbf{a}_k : 0 \leq \lambda_1, \ldots, \lambda_k \leq 1\}$ . The permutahedron  $\Pi_{d-1}$  is the zonotope of the root system  $\{\mathbf{e}_i - \mathbf{e}_j : 1 \leq i < j \leq d\}$ .
- 5. (Cyclic polytope)  $C_d(m) = \operatorname{conv}\{(1, t_i, t_i^2, \dots, t_i^{d-1}) | 1 \le i \le m\}$  for  $t_1 < \dots < t_m$ .
- 6. (Order polytope of a poset P)  $\mathcal{O}(P) = \{ \mathbf{x} \in \mathbb{R}^P : x_i \ge 0 \text{ for all } i \text{ and } x_i \le x_j \text{ if } i < j \text{ in } P \}.$
- 7. (Chain polytope of a poset P)  $\mathcal{C}(P) = \{ \mathbf{x} \in \mathbb{R}^P : x_i \ge 0 \text{ for all } i \text{ and } x_{i_1} + \dots + x_{i_k} \le 1 \text{ for each chain } i_1 < \dots < i_k \text{ in } P \}.$
- 8. (Type A root polytope)  $A_{d-1} = \operatorname{conv} \{ \mathbf{e}_i \mathbf{e}_j : 1 \le i \ne j \le d \}$  in  $\mathbb{R}^d$ .
- 9. (Type A positive root polytope)  $A_{d-1}^+ = \operatorname{conv} \{ \mathbf{e}_i \mathbf{e}_j : 1 \le i < j \le d \}$  in  $\mathbb{R}^d$ .
- 10. (Flow polytope) Given a directed graph G = (V, E) and a vector  $\mathbf{b} \in \mathbb{R}^V$ , the flow polytope is  $F_G(\mathbf{b}) = \{\mathbf{f} \in \mathbb{R}^E : f_e \ge 0 \text{ for all } e \in E \text{ and } \sum_{vw \in E} f_{vw} \sum_{uv \in E} f_{uv} = b_v \text{ for all } v \in V\}$ . We think of  $f_e$  as a flow on edge e, so that the excess flow or leak at each vertex v equals  $b_v$ .

- 11. (CRY polytope)  $\operatorname{CRY}_n = F_{K_{n+1}}(1, 0, \dots, 0, -1)$  where  $K_{n+1}$  is the complete graph on [n+1] with edges directed  $i \to j$  for i < j.
- 12. (Associahedron) Assoc<sub>d-1</sub> is a polytope whose faces are in bijection with the ways of subdividing a convex (d+3)-gon into polygons without introducing new vertices. Faces F and G satisfy that  $F \subset G$  if and only if the subdivision of F refines the subdivision of G. There are several different polytopal realizations of Assoc<sub>d-1</sub>; see [CSZ11] for a survey.
- 13. (Matroid polytope) If  $M = \{v_1, \ldots, v_d\}$  is a set of vectors spanning a vector space  $\mathbb{k}^r$ , the matroid (basis) polytope is  $P_M = \operatorname{conv}\{\mathbf{e}_{i_1} + \cdots + \mathbf{e}_{i_r} : \{v_{i_1}, \ldots, v_{i_r}\}$  is a basis of  $\mathbb{k}^r\}$  in  $\mathbb{R}^d$ . When M is generic we get  $P_M = \Delta(r, d)$ . This construction is better understood in the context of matroids; see Section 7.
- 14. (Generalized permutahedra / polymatroids) Many interesting polytopes are deformations of the permutahedron, obtained by moving the vertices of  $\Pi_{d-1}$  while respecting all the edge directions. Examples include the polytopes  $\Delta_{d-1}, \Delta_{e-1} \times \Delta_{d-e-1}, \Delta(r, d), A_{d-1}, P_M, \text{Assoc}_{d-1}$ above. Such polytopes are called *polymatroid base polytopes* or *generalized permutahedra*; see [Fuj05, Pos09].

Many of these polytopes are related to the permutation group  $S_d$  and the corresponding type A root system conv $\{\mathbf{e}_i - \mathbf{e}_j : 1 \le i \ne j \le d\}$ . Many have generalizations to the wider context of finite Coxeter groups and root systems; see for example [ABH+11, BGW03, FR07, FH91, MM12] and the references therein.

The faces of most of these polytopes can be described combinatorially. This should help us appreciate these polytopes, because there are not many interesting families of polytopes for which we can do this. We describe the most interesting ones.

3. (Permutahedron) The H-description of  $\Pi_{d-1}$  is

 $x_1 + \dots + x_d = d(d+1)/2, \qquad x_{i_1} + \dots + x_{i_k} \ge k(k+1)/2 \text{ for } \emptyset \subsetneq \{i_1, \dots, i_k\} \subsetneq [d]$ 

There is a bijection  $\mathcal{S} \leftrightarrow F_{\mathcal{S}}$  between the ordered set partitions of [d] and the faces of the permutahedron  $\Pi_{d-1}$ .

- 4. (Zonotope) The face enumeration of Z(A) is the subject of the upcoming Theorem 6.5.2.
- 5. (Cyclic polytope) Let  $\mathbf{t}_i = (1, t_i, \dots, t_i^{d-1})$ . The facets of the cyclic polytope  $C_d(m)$  are the simplices conv $\{\mathbf{t}_s : s \in S\}$  for the *d*-subsets  $S \subseteq [m]$  satisfying **Gale's evenness condition**: between any i < j not in S there is an even number of elements of S. There is a similar description for all faces of the cyclic polytope. In particular, the combinatorics of  $C_d(m)$  is independent of  $t_1, \dots, t_m$ . Two other remarkable facts are the following.
  - Every subset of at most d/2 vertices forms a face of  $C_d(m)$ .
  - (McMullen's Upper Bound Theorem [MS71]) Among all *d*-polytopes with *m* vertices,  $C_d(m)$  maximizes the number of faces of dimension *k* for all  $2 \le k \le d-1$ .

The rich theory of *positroids* can be seen as a generalization of the study of cyclic polytopes; see Section 7 and [ARW13a, Pos06] for this connection.

- 6. (Order polytope) The vertices of  $\mathcal{O}(P)$  are  $\sum_{i \notin I} \mathbf{e}_i$  for the order ideals  $I \subseteq P$ .
- 7. (Chain polytope) The vertices of  $\mathcal{C}(P)$  are  $\sum_{i \in A} \mathbf{e}_i$  for the antichains  $A \subseteq P$ .
- 13. (Matroid polytope) The matroid polytope is cut out (non-minimally) by the inequalities

$$\sum_{e \in E} x_e = r(E), \qquad \sum_{e \in S} x_e \le r(S) \text{ for } E \subset S$$

The facets are characterized in [FS05].

14. (Generalized permutahedra) There are several interesting result on various classes of generalized permutahedra; see [PRW08].

# 5.3 Counting faces

The f-vector of a d-polytope P is

$$f_P = (f_0, f_1, \dots, f_{d-1}, f_d)$$

where  $f_i$  is the number of (i-1)-dimensional faces of P for  $0 \le i \le d$ .<sup>10</sup> The problem of characterizing the f-vectors of various kinds of polytopes (or more general polyhedral complexes) is a central one in combinatorics. We offer a very brief discussion; for more detailed accounts, see [BB97, MN13, Zie95]. A one sentence summary is this: we completely understand the f-vectors of simplicial (or equivalently, of simple) polytopes, but we are far from understanding the f-vectors of general polytopes. There are many interesting results and (mostly) open questions in between.

*f*-vectors. The most important result about *f*-vectors of arbitrary polytopes is the following.

**Theorem 5.3.1.** [MS71] (McMullen's Upper Bound Theorem) For any polytope P of dimension d with m vertices, we have

$$f_i(P) \le f_i(C_d(m))$$
 for  $i = 0, 1, \dots, d-1$ ,

where  $C_d(m)$  is the cyclic polytope.

*h*-vectors. If *P* is a simplicial *d*-polytope (or more generally, any simplicial complex of dimension d-1), we define the *h*-vector  $h_P = (h_0, h_1, \ldots, h_d)$  by the equivalent equations:

$$h_0 x^d + h_1 x^{d-1} + \dots + h_d x^0 = f_0 (x-1)^d + f_1 (x-1)^{d-1} + \dots + f_d (x-1)^0$$
, or  
 $i$   
 $i$   
 $i$   
 $(d-i)$ 

$$h_i = \sum_{j=0}^{i} (-1)^{i-j} {d-j \choose i-j} f_j$$
 and  $f_i = \sum_{j=0}^{i} {d-j \choose i-j} h_j$  for  $0 \le i \le d$ .

The *h*-vector is a more economical way of storing the *f*-vector, due to the *Dehn-Sommerville* relations:  $h_i = h_{d-i}$  for  $0 \le i \le d$ .

In fact, the *g*-theorem characterizes completely the *f*-vectors of simplicial polytopes [BL81, Sta80c]! This spectacular result (which McMullen conjectured, and described as "even more intriguing, if rather less plausible" [MS71]) is one of the most important achievements of algebraic and geometric combinatorics to date. To state it, we need some definitions.

<sup>&</sup>lt;sup>10</sup>Note that we include the empty face and omit the full-dimensional face P in this enumeration.

We say a sequence of nonnegative integers  $(m_0, m_1, \ldots, m_d)$  is an **M-sequence** if there exists a set S of monomials in  $x_1, \ldots, x_n$ , containing exactly  $m_i$  monomials of degree *i* for  $i = 0, 1, \ldots, d$ , such that  $m' \in S$  and m|m' implies  $m \in S$ . Stanley gave a numerical characterization, as follows.

For any positive integers a and i, there is a unique representation  $a = \binom{a_i}{i} + \binom{a_{i-1}}{i-1} + \dots + \binom{a_j}{j}$ for some  $a_i > a_{i-1} > \dots > a_j \ge j \ge 1$ . We then define  $a^{\langle i \rangle} = \binom{a_i}{i+1} + \binom{a_{i-1}}{i} + \dots + \binom{a_j}{j+1}$ . Then  $(m_0, m_1, \dots, m_d)$  is an M-sequence if and only if  $m_0 = 1$  and  $m_{i+1} \le m_i^{\langle i \rangle}$  for  $i = 1, 2, \dots, d-1$ .

**Theorem 5.3.2.** [BL81, Sta80c] (Billera-Lee-Stanley's g-theorem) A sequence  $(h_0, \ldots, h_d)$  of positive integers is the h-vector of a simplicial d-polytope if and only if

- 1.  $h_i = h_{d-i}$  for i = 0, 1, ..., |d/2| (Dehn-Somerville equations), and
- 2.  $(g_0, g_1, \ldots, g_{\lfloor d/2 \rfloor})$  is an *M*-sequence, where  $g_0 = h_0$  and  $g_i = h_i h_{i-1}$  for  $1 \le i \le \lfloor d/2 \rfloor$ .

For general polytopes, the situation is much less clear, even in dimension 4, a case which has been studied extensively. [Zie02] Stanley [Sta87] defined a more subtle **toric** h-vector. It coincides with the h-vector when P is simplicial, and it also satisfies the Dehn-Sommerville equations. His definition may be seen as a combinatorial formula for the dimensions of the intersection cohomology groups of the corresponding projective toric variety (if P is a rational polytope). In a different direction, for "cubical" polytopes P whose proper faces are all cubes, Adin [Adi96] defined the **cubical** h-vector, which also satisfies the Dehn-Sommerville relations. Characterizing these toric and cubical h-vectors is an important open problem.

**Flag** *f*-vectors and *h*-vectors. The flag *f*-vector  $(f_D)_{D\subseteq[0,d-1]}$  of a *d*-polytope enumerates the flags  $F_1 \subset \cdots \subset F_k$  of given dimensions  $D = \{d_1 < \cdots < d_k\}$  for all  $D \subseteq [0, d-1]$ . As we described in Section 4.8 (in the wider context of Eulerian posets), this information can be more economically stored in the **cd**-index of *P*. This encoding incorporates all linear relations among the flag *f*-vector.

As we saw in Theorem 4.8.2, the **cd**-index of a polytope P is non-negative; this is not true for general Eulerian posets. Another very interesting question, which is wide open, is to classify the **cd**-indices of polytopes.

# 5.4 Counting lattice points: Ehrhart theory

In this section we are interested in "measuring" a polytope P by counting the lattice points in its integer dilations  $P, 2P, 3P, \ldots$  We limit our attention to **lattice polytopes**, whose vertices are lattice points, and to **rational polytopes**, whose vertices have rational coordinates; at the moment there is no good theory for general polytopes.

**Theorem 5.4.1.** Let  $P \subset \mathbb{R}^d$  be a lattice polytope. There are polynomials  $L_P(x)$  and  $L_{P^o}(x)$  of degree dim P, called the **Ehrhart polynomial** and interior Ehrhart polynomial of P, such that

1. the number of lattice points in the nth dilation of P and its interior  $P^{o}$  are

$$L_P(n) = |nP \cap \mathbb{Z}^d| \qquad L_{P^o}(n) = |nP^o \cap \mathbb{Z}^d| \qquad \text{for all } n \in \mathbb{N}.$$

2. the Ehrhart reciprocity law holds:

$$(-1)^{\dim P} L_P(-x) = L_{P^o}(x).$$

**Note.** If P is a rational polytope, then  $L_P(n)$  is instead given by a **quasipolynomial**; that is, there exist an integer m and polynomials  $L_1(x), \ldots, L_m(x)$  such that  $|nP \cap \mathbb{Z}^d| = L_k(n)$  whenever  $n \equiv k \pmod{m}$ .

Sketch of Proof of Theorem 5.4.1. By working in one dimension higher, we can consider the various dilations of P all at once. We embed  $\mathbb{R}^d$  into  $\mathbb{R}^{d+1}$  by mapping  $\mathbf{v}$  to  $(\mathbf{v}, 1)$ , and consider the cone

$$\operatorname{cone}(P) = \{\lambda_1(\mathbf{v}_1, 1) + \dots + \lambda_n(\mathbf{v}_n, 1) : \lambda_1, \dots, \lambda_n \ge 0\}$$



Figure 5.3: The cone of a polytope and its slices at height 0, 1, 2, 3.

Then for each  $n \in \mathbb{N}$ , the slice  $x_{d+1} = n$  of cone(P) is a copy of the dilation nP, as shown in Figure 5.3. Two key ingredients of the proof will be the **lattice point enumerator** of cone(P) and the **Ehrhart series** of P:

$$\sigma_{\operatorname{cone}(P)}(\mathbf{z}) = \sum_{\mathbf{p}\in\operatorname{cone}(P)} \mathbf{z}^{\mathbf{p}}, \qquad \operatorname{Ehr}_{P}(z) = \sum_{n\geq 0} L_{P}(n) z^{n} = \sigma_{\operatorname{cone}(P)}(1,\ldots,1,z), \qquad (24)$$

where  $\mathbf{z}^{\mathbf{p}} = z_1^{p_1} \cdots z_{d+1}^{p_{d+1}}$ .

We first prove the theorem for simplices, and then use triangulations to prove the general case.

Step 1. (Simplices) First we prove the result when P is a simplex, which we may assume is fulldimensional. In this case cone(P) is an orthant generated by d + 1 linearly independent vectors  $\mathbf{w}_1, \ldots, \mathbf{w}_{d+1}$ , where  $\mathbf{w}_i = (\mathbf{v}_i, 1)$ .

1. Define the lower and upper fundamental (half-open) parallelepipeds

$$\Pi = \{\lambda_1 \mathbf{w}_1 + \dots + \lambda_{d+1} \mathbf{w}_{d+1} : 0 \le \lambda_1, \dots, \lambda_{d+1} < 1\}$$
  
$$\Pi^+ = \{\lambda_1 \mathbf{w}_1 + \dots + \lambda_{d+1} \mathbf{w}_{d+1} : 0 < \lambda_1, \dots, \lambda_{d+1} \le 1\}.$$

Then cone(P) is tiled by the various non-negative **w**-integer translates of  $\Pi$ , namely, the parallelepipeds  $\Pi + k_1 \mathbf{w}_1 + \cdots + k_{d+1} \mathbf{w}_{d+1}$  for  $\mathbf{k} = (k_1, \ldots, k_{d+1}) \in \mathbb{N}^{d+1}$ . Similarly, the interior int(cone(P)) is tiled by the various non-negative **w**-integer translates of  $\Pi^+$ . Therefore

$$\sigma_{\operatorname{cone}(P)}(\mathbf{z}) = \sigma_{\Pi}(\mathbf{z}) \left(\frac{1}{1-\mathbf{z}^{\mathbf{w}_{1}}}\right) \cdots \left(\frac{1}{1-\mathbf{z}^{\mathbf{w}_{d+1}}}\right)$$
  
$$\sigma_{\operatorname{int}(\operatorname{cone}(P))}(\mathbf{z}) = \sigma_{\Pi^{+}}(\mathbf{z}) \left(\frac{1}{1-\mathbf{z}^{\mathbf{w}_{1}}}\right) \cdots \left(\frac{1}{1-\mathbf{z}^{\mathbf{w}_{d+1}}}\right)$$

Using (24) we get

Ehr<sub>P</sub>(z) = 
$$\frac{\sigma_{\Pi}(1,\ldots,1,z)}{(1-z)^{d+1}}$$
, Ehr<sub>P<sup>o</sup></sub>(z) =  $\frac{\sigma_{\Pi^+}(1,\ldots,1,z)}{(1-z)^{d+1}}$ ,

which are rational functions. Then, by Theorem 2.4.2,  $L_P(n)$  and  $L_{P^o}(n)$  are polynomial functions of n of degree d, as desired.

2. Now observe that  $\sigma_{\Pi^+}(\mathbf{z}) = \mathbf{z}^{\mathbf{w}_1 + \dots + \mathbf{w}_{d+1}} \sigma_{\Pi}(1/z_1, \dots, 1/z_{d+1})$ , because  $\Pi^+$  is the translation of  $-\Pi$  by the vector  $\mathbf{w}_1 + \dots + \mathbf{w}_{d+1}$ . This gives

$$\operatorname{Ehr}_{P^{o}}(z) + (-1)^{d} \operatorname{Ehr}_{P}(1/z) = 0.$$

It remains to invoke the fact that if f is a polynomial, then  $F^+(z) = \sum_{n\geq 0} f(n)z^n$  (which is a rational function of z) and  $F^-(z) = \sum_{n<0} f(n)z^n$  (which is a rational function of 1/z, and hence of z) satisfy  $F^+(z) + F^-(z) = 0$  as rational functions. This implies that  $(-1)^{\dim P} L_P(-n) = L_{P^o}(n)$  for every positive integer n, and hence these two polynomials are equal.

Step 2. (The general case) Now let P be a general lattice polytope, and let  $\mathcal{T}$  be a triangulation of P. Recall that  $\mathcal{T}^o$  is the set of non-boundary faces of  $\mathcal{T}$ .

1. We have

$$L_P(n) = \sum_{F \in \mathcal{T}} L_{F^o}(n), \qquad L_{P^o}(n) = \sum_{F \in \mathcal{T}^o} L_{F^o}(n)$$

which implies that  $L_P$  and  $L_{P^o}$  are polynomials of degree dim P, by Step 1.

2. Let  $\widehat{\mathcal{T}}$  be the face poset of  $\mathcal{T}$ , with an additional maximum element  $\widehat{1}$ . We have the Möbius dual relations

$$L_F = \sum_{G \le F} L_{G^o}, \qquad L_{F^o} = \sum_{G \le F} \mu(G, F) L_G \qquad \text{for all } F \in \mathcal{T}$$

omitting the arguments of the polynomials in question. This gives us

$$L_P = \sum_{F \in \mathcal{T}} L_{F^o} = \sum_{F \in \mathcal{T}} \sum_{G \le F} \mu(G, F) L_G = \sum_{G \in \mathcal{T}} -\mu(G, \widehat{1}) L_G = \sum_{G \in \mathcal{T}^o} (-1)^{\dim P - \dim G} L_G.$$

where in the last step we are invoking Theorem 4.5.1.11. Now, using that the simplices  $G \in \mathcal{T}^o$  satisfy Ehrhart reciprocity: we obtain

$$L_P(-n) = \sum_{G \in \mathcal{T}^o} (-1)^{\dim P - \dim G} L_G(-n) = (-1)^{\dim P} \sum_{G \in \mathcal{T}^o} L_{G^o}(n) = (-1)^{\dim P} L_{P^o}(n),$$
  
red.

as desired.

Next we observe that the volume of a lattice polytope can be recovered from its Ehrhart polynomial. If dim P < d, there is a small subtlety: Let V be the affine span of P, and  $\Lambda = V \cup \mathbb{Z}^d$ . Then we need to normalize the volumes along V, so that any "unit" cube, generated by a  $\mathbb{Z}$ -basis of  $\Lambda$ , has normalized volume 1. We let vol(P) denote the **normalized volume** or **lattice volume** of P. When P is full-dimensional, this is the usual volume.

For example, the segment from (1, 1) to (4, 7) has normalized volume (lattice length) 3, because in this case the lattice  $\Lambda$  is generated by the primitive vector (1, 2), and  $(4, 7) - (1, 1) = \mathbf{3}(1, 2)$ . **Proposition 5.4.2.** If  $L_P(t) = c_d t^d + \cdots + c_1 t + c_0$  is the Ehrhart polynomial of a lattice polytope P, then  $c_d = \operatorname{vol}(P)$  and  $c_0 = 1$ .

Sketch of Proof. We obtain better and better approximations of the volume of P by choosing finer and finer grids  $(\frac{1}{n}\mathbb{Z})^d$ , and placing a cube of volume  $\frac{1}{n^d}$  centered at each lattice point in  $P \cap (\frac{1}{n}\mathbb{Z})^d$ . Therefore

$$\operatorname{vol}(P) = \lim_{n \to \infty} \frac{1}{n^d} \left| P \cap \left(\frac{1}{n}\mathbb{Z}\right)^d \right| = \lim_{n \to \infty} \frac{1}{n^d} \left| nP \cap \mathbb{Z}^d \right| = \lim_{n \to \infty} \frac{L_P(n)}{n^d} = c_d$$

Since the only lattice point in 0P is the origin,  $c_0 = L_P(0) = 1$ .

A lattice simplex S in  $\mathbb{R}^d$  with vertices  $\mathbf{v}_1, \ldots, \mathbf{v}_k$  is **unimodular** if the vectors  $\mathbf{v}_2 - \mathbf{v}_1, \ldots, \mathbf{v}_d - \mathbf{v}_1$  are a lattice basis for the lattice aff  $S \cap \mathbb{Z}^d$ . A triangulation is **unimodular** if all its simplices are unimodular. The following result shows that unimodular triangulations are particularly useful for enumerative purposes.

**Proposition 5.4.3.** 1. For every lattice polytope  $P \subset \mathbb{R}^d$  the **Ehrhart**  $h^*$ -polynomial  $h_P^*(x) = h_0^* + h_1^* x + \cdots + h_d^* x^d$ , which is defined by

$$\operatorname{Ehr}_{P}(z) = \frac{h_{P}^{*}(z)}{(1-z)^{d+1}}$$

has nonnegative coefficients:  $h_k^* \ge 0$  for all k. 2. If P has a unimodular triangulation  $\mathcal{T}$ , then

$$h_P^*(z) = h_\mathcal{T}(z),$$

where the h-polynomial is the generating function  $h_{\mathcal{T}}(z) = h_0 + h_1 z + \cdots + h_d z^d$  for the h-vector  $(h_0, \ldots, h_d)$  of  $\mathcal{T}$ , as defined in Section 5.3.

*Proof.* 1. is due to Stanley [Sta80b]; for a short proof see [BR07, Theorem 3.12]. For 2., we have

$$\operatorname{Ehr}_{P}(z) = \sum_{F \in \mathcal{T}} \operatorname{Ehr}_{F^{o}}(z) = \sum_{F \in \mathcal{T}} \frac{\sigma_{\Pi_{F}^{+}(1,\dots,1,z)}}{(1-z)^{\dim F+1}}$$

where  $\Pi_+(F)$  is the upper fundamental parallelepiped of F. Since F is unimodular, the only lattice point in  $\Pi_F^+$  is the sum of its generators, which is at height dim F + 1, so

$$\operatorname{Ehr}_{P}(z) = \sum_{F \in \mathcal{T}} \left( \frac{z}{1-z} \right)^{\dim F+1} = \sum_{k=0}^{d+1} f_k \left( \frac{z}{1-z} \right)^k = \frac{\sum_{k=0}^{d+1} h_k z^k}{(1-z)^{d+1}}$$

as desired.

**Examples**. The following polytopes have particularly nice Ehrhart polynomials and Ehrhart series.

1. (Polygon) P a lattice polygon in  $\mathbb{R}^2$ :

$$L_P(n) = \left(I + \frac{B}{2} - 1\right)n^2 + \frac{B}{2}n + 1, \qquad \text{Ehr}_P(z) = \frac{Iz^2 + (I + B - 3)z + 1}{(1 - z)^3}$$

where P contains I lattice points in its interior and B lattice points on its boundary. This is equivalent to Pick's formula for the area of a lattice polygon.

2. (Simplex)  $\Delta_{d-1}$ :

$$L_{\Delta_{d-1}}(n) = \binom{n+d-1}{d-1}, \qquad \text{Ehr}_{\Delta_{d-1}}(z) = \frac{1}{(1-z)^d}.$$

3. (Cube)  $\Box_d$ :

$$L_{\Box_d}(n) = (n+1)^d$$
,  $\operatorname{Ehr}_{\Box_d}(z) = \frac{\sum_{k=0}^d A(d,k) z^k}{(1-z)^{d+1}}$ ,

where the **Eulerian number** A(d,k) is the number of permutations  $\pi$  of [d] with k-1 descents; that is, positions i with  $\pi(i) > \pi(i+1)$ .

4. (Crosspolytope)  $\Diamond_d$ :

$$L_{\Diamond_d}(n) = \sum_{k=0}^d 2^k \binom{d}{k} \binom{n}{k}, \qquad \text{Ehr}_{\Diamond_d}(z) = \frac{(1+z)^d}{(1-z)^{d+1}}.$$

5. (Product of two simplices)  $P = \Delta_{c-1} \times \Delta_{d-1}$ :

$$L_P(n) = \binom{n+c-1}{c-1} \binom{n+d-1}{d-1}, \qquad \text{Ehr}_P(z) = \frac{\sum_{k=0}^{\min(c-1,d-1)} \binom{c-1}{k} \binom{d-1}{k} z^k}{(1-z)^{c+d-1}}.$$

Every triangulation of  $\Delta_{c-1} \times \Delta_{d-1}$  is unimodular, with *h*-vector given by the *h*<sup>\*</sup>-vector above. These triangulations are very interesting combinatorially, and play an important role in tropical geometry and other contexts; see [AB07, AD09, DS04, San05] and the references therein.

6. (Hypersimplex)  $\Delta(r, d)$ :

$$L_{\Delta(r,d)}(n) = [z^{rn}] \left(\frac{1-z^{n+1}}{1-z}\right)^d, \qquad \text{vol}(\Delta_{r,d}) = \frac{A(d-1,r)}{(d-1)!}$$

where, again, A(d-1,r) denotes the Eulerian numbers. No simple formula is known for the Ehrhart series of  $\Delta(r, d)$ . There is a nice formula for the "half-open hypersimplex"; see [HLO13]. An elegant triangulation of  $\Delta(r, d)$  was given (using four different descriptions) by Lam-Postnikov, Stanley, Sturmfels, and Ziegler; see [LP07].

7. (Permutahedron)  $P = \prod_{d=1}$ :

$$L_{\Pi_{d-1}}(n) = \sum_{i=0}^{d-1} f_i n^i, \qquad \text{vol}(\Pi_{d-1}) = d^{d-2}$$

where  $f_i$  is the number of forests on [d] with *i* vertices. [Sta91]

8. (Zonotope) P = Z(A):

$$L_{Z(A)}(n) = n^r M\left(1 + \frac{1}{n}, 1\right)$$

where r is the rank of A and M(x, y) is the arithmetic Tutte polynomial of Section 7.9. [Sta91] This polynomial is difficult to compute in general; when A is a root system, explicit formulas are given in [ACH14].

9. (Cyclic polytope)  $P = C_d(t_1, \ldots, t_m)$ :

$$L_P(n) = \sum_{k=0}^d \operatorname{vol} C_k(t_1, \dots, t_m) n^k.$$

See [Liu05]. The triangulations of the cyclic polytope are unusually well behaved; see [DLRS10, Ram97] and the references therein. In particular, when m = d + 4, this is one of the few polytopes whose triangulations have been enumerated exactly (and non-trivially) [AS02].

10. (Order polytope and chain polytope):  $\mathcal{O}(P), \mathcal{C}(P)$ :

$$L_{\mathcal{O}(P)} = L_{\mathcal{C}(P)} = \Omega_P(n+1), \qquad \operatorname{vol}(\mathcal{O}(P)) = \operatorname{vol}(\mathcal{C}(P)) = e(P)/|P|!,$$

where  $\Omega_P$  is the order polynomial of P and e(P) is the number of linear extensions, as discussed in Section 4.3. [Sta86] Remarkably,  $\mathcal{O}(P)$  and  $\mathcal{C}(P)$  have the same Ehrhart polynomial, even though they are not metrically equivalent in general. There is a nice characterization of the posets P such that  $\mathcal{O}(P)$  and  $\mathcal{C}(P)$  may be obtained from one another by a unimodular change of basis [HL12].

11. (Root polytope):  $A_{d-1}$ :

Ehr<sub>A<sub>d</sub></sub>(z) = 
$$\frac{\sum_{k=0}^{d} {\binom{d}{k}}^2 z^k}{(1-x)^d}$$
, vol(A<sub>d</sub>) =  $\frac{{\binom{2d}{d}}}{d!}$ .

There are similar formulas for the other classical root polytopes  $B_d, C_d, D_d$ , as well as for the positive root polytopes. For example,  $\operatorname{vol}(A_d^+) = C_d/d!$  where  $C_d$  is the *d*th Catalan number. Explicit unimodular triangulations were constructed in [ABH<sup>+</sup>11].

12. (CRY polytope / Flow polytope)  $CRY_n = F_{K_{n+1}}(1, 0, \dots, 0, -1)$ :

$$\operatorname{vol}(\operatorname{CRY}_n) = C_0 C_1 C_2 \cdots C_{n-2},$$

a product of Catalan numbers. [Zei99]. No combinatorial proof of this fact is known. Flow polytopes are of great importance due to their close connection with the Kostant partition function, which Gelfand described as *"the transcendental element which accounts for many* of the subtleties of the Cartan-Weyl theory" of representations of semisimple Lie algebras. [Ret13]. There are many other interesting combinatorial results; see for example [BV08, MM12]

- 13. (Matroid polytopes) A combinatorial formula for the volume of a matroid polytope is given in [ABD10].
- 14. (Generalized permutahedra / polymatroids) There are many interesting results about the volumes and lattice points of various families of generalized permutahedra; see [Pos09].
- 15. (Cayley polytope)  $\mathbf{C}_n = \{ \mathbf{x} \in \mathbb{R}^n : 1 \le x_1 \le 2 \text{ and } 1 \le x_i \le 2x_{i-1} \text{ for } 2 \le i \le n \}$ :

$$\operatorname{vol}(\mathbf{C}_n) = \frac{c_{n+1}}{n!}$$

where  $c_n$  is the number of connected graphs on [n]. The related *Tutte polytope* has volume given by an evaluation of the Tutte polynomial (see Section 7.6) of the complete graph. [KP13]

# 6 Hyperplane arrangements

We now discuss arrangements of hyperplanes in a vector space. The questions that we ask depend on whether the underlying field is  $\mathbb{R}, \mathbb{C}$ , or a finite field  $\mathbb{F}_q$ ; but in every case, the underlying combinatorics plays an important role. The presentation of this section is heavily influenced by [Sta07]. See [OT92] for a great introduction to more algebraic and topological aspects of the theory of hyperplane arrangements.

After developing the basics in Section 6.1, in Section 6.2 we introduce the *characteristic polynomial*, which plays a crucial role in this theory. In Section 6.3 we discuss some of its important properties, and in Section 6.4 we develop several techniques for computing it. We illustrate these techniques by computing the characteristic polynomials of many arrangements of interest. Finally, in Section 6.5, we give a remarkable formula for the **cd**-index of an arrangement, which enumerates the flags of faces of given dimensions.

# 6.1 **Basic definitions**

Let k be a field and  $V = k^d$ . A hyperplane arrangement  $\mathcal{A} = \{H_1, \ldots, H_n\}$  is a collection of affine hyperplanes in V, say,

$$H_i = \{x \in V : v_i \cdot x = b_i\}$$

for nonzero normal vectors  $v_1, \ldots, v_n \in V$  and constants  $b_1, \ldots, b_n \in \mathbb{k}$ . We say  $\mathcal{A}$  is **central** if all hyperplanes have a common point – in the most natural examples, the origin is a common point. Figure 6.1 shows a central arrangement of 4 hyperplanes in  $\mathbb{R}^3$ .



Figure 6.1: An arrangement.

In some ways, central arrangements are slightly better behaved than affine arrangements. We can **centralize** an affine arrangement  $\mathcal{A}$  in  $\mathbb{k}^n$  to obtain the **cone** of  $\mathcal{A}$ , an arrangement  $c\mathcal{A}$  in  $\mathbb{k}^{n+1}$ , by turning the hyperplane  $a_1x_1 + \cdots + a_nx_n = a$  in  $\mathbb{k}^n$  into the hyperplane  $a_1x_1 + \cdots + a_nx_n = ax_{n+1}$  in  $\mathbb{k}^{n+1}$ , and adding the hyperplane  $x_{n+1} = 0$ 

Sometimes arrangements are "too central", in the sense that their intersection is a subspace L of positive dimension. In that case, there is little harm in intersecting our arrangement with the orthogonal complement  $L^{\perp}$ . We define the **essentialization** of  $\mathcal{A}$  to be the arrangement  $ess(\mathcal{A}) = \{H \cap L^{\perp} : H \in \mathcal{A}\}$  in  $L^{\perp}$ . The result is an **essential** arrangement, where the intersection of the hyperplanes is the origin. In most problems of interest, there is no important difference between  $\mathcal{A}$  and  $ess(\mathcal{A})$ .

A key object is the **complement** 

$$V(\mathcal{A}) = V \setminus \left(\bigcup_{H \in \mathcal{A}} H\right).$$

and we now introduce a polynomial which is a fundamental tool in the study of  $V(\mathcal{A})$ .

#### 6.2 The characteristic polynomial.

There is a combinatorial polynomial which knows a tremendous amount about the complement  $V(\mathcal{A})$  of an arrangement  $\mathcal{A}$ . The kinds of questions that we ask about  $V(\mathcal{A})$  depend on the underlying field.

• If  $\mathbb{k} = \mathbb{R}$  then every hyperplane  $v_i \cdot x = b_i$  divides V into two half-spaces, where  $v_i \cdot x < b_i$  and  $v_i \cdot x > b_i$ , respectively. Therefore an arrangement  $\mathcal{A}$  divides  $\mathbb{R}^d$  into  $a(\mathcal{A})$  regions, which are the connected components of the complement  $\mathbb{R}^d \setminus \mathcal{A}$ . Let  $b(\mathcal{A})$  be the number of those regions which are bounded.<sup>11</sup> A central question about real hyperplane arrangements is to compute the numbers  $a(\mathcal{A})$  and  $b(\mathcal{A})$  of regions and bounded regions.

• If  $\mathbb{k} = \mathbb{C}$ , then it is possible to walk around a hyperplane without crossing it; this produces a loop in the complement  $V(\mathcal{A})$ . Therefore  $V(\mathcal{A})$  has nontrivial topology, and it is natural to ask for its Betti numbers.

• If  $\mathbb{k} = \mathbb{F}_q$  is a finite field, where q is a prime power, then  $V(\mathcal{A})$  is a finite set, and the simplest question we can ask is how many points it contains.

Amazingly, the **characteristic polynomial**  $\chi_{\mathcal{A}}(q)$  can answer these questions immediately. Let us define it.

Define a flat of  $\mathcal{A}$  to be an affine subspace obtained as an intersection of hyperplanes in  $\mathcal{A}$ . The intersection poset  $L_{\mathcal{A}}$  is the set of flats partially ordered by reverse inclusion. If  $\mathcal{A}$  is central, then  $L_{\mathcal{A}}$  is a geometric lattice, as discussed in Section 4.2. If  $\mathcal{A}$  is not central, then  $L_{\mathcal{A}}$  is only a geometric meet semilattice [WW86]. The rank  $r = r(\mathcal{A})$  of  $\mathcal{A}$  is the height of  $L_{\mathcal{A}}$ .

The characteristic polynomial of  $\mathcal{A}$  is

$$\chi_A(q) = \sum_{F \in L_{\mathcal{A}}} \mu(\widehat{0}, F) q^{\dim F}$$

Figure 6.2 shows the intersection poset of the arrangement  $\mathcal{A}$  in Figure 6.1; its characteristic polynomial  $\chi_{\mathcal{A}}(q) = q^3 - 4q^2 + 5q - 2$  is easily computed by adding the Möbius numbers on each level of  $L_{\mathcal{A}}$ .

The reader may wish to compute some examples, such as the characteristic polynomial of an arrangement of lines in  $\mathbb{R}^2$ , or the arrangement of *n* coordinate hyperplanes in  $\mathbb{k}^n$ . We will see some general techniques to compute characteristic polynomials in Section 6.4.

**Theorem 6.2.1.** The characteristic polynomial  $\chi_{\mathcal{A}}(x)$  contains the following information about the complement  $V(\mathcal{A})$  of a hyperplane arrangement  $\mathcal{A}$ .

<sup>&</sup>lt;sup>11</sup>If  $\mathcal{A}$  is not essential, we let  $b(\mathcal{A})$  be the number of **relatively bounded** regions. These are the regions which become bounded in the essentialization ess( $\mathcal{A}$ ).



Figure 6.2: The intersection poset of  $\mathcal{A}$ .

1.  $(\mathbb{k} = \mathbb{R})$  [Zas75] Let  $\mathcal{A}(A)$  be a **real** hyperplane arrangement in  $\mathbb{R}^n$ . The number of regions and relatively bounded regions of the complement  $V(\mathcal{A})$  are

$$a(\mathcal{A}) = (-1)^n \chi_{\mathcal{A}}(-1), \qquad b(\mathcal{A}) = (-1)^{r(\mathcal{A})} \chi_{\mathcal{A}}(1).$$

2.  $(\mathbb{k} = \mathbb{C})$  [GM88, OS80] Let  $\mathcal{A}(A)$  be a **complex** hyperplane arrangement in  $\mathbb{C}^n$ . The complement  $V(\mathcal{A})$  has Poincaré polynomial

$$\sum_{k\geq 0} \operatorname{rank} H^k(V(A), \mathbb{Z})q^k = (-q)^n \chi_{\mathcal{A}}\left(\frac{-1}{q}\right).$$

3.  $(\mathbb{k} = \mathbb{F}_q)$  [Ath96, CR70] Let  $\mathcal{A}(A)$  be a hyperplane arrangement in  $\mathbb{F}_q^n$  where  $\mathbb{F}_q$  is the finite field of q elements for a prime power q. The complement  $V(\mathcal{A})$  has size

$$|V(A)| = \chi_{\mathcal{A}}(q).$$

*Proof.* 3. This is a typical enumerative problem where our set of objects (the points of  $\mathbb{F}_q^n$ ) are partially ordered, according to how special their position is with respect to  $\mathcal{A}$ . This is a natural setting to apply Möbius inversion.

For each flat F, let  $f_{=}(F)$  be the number of points of  $\mathbb{F}_q^n$  which are on F, and on no smaller flat. Since there are  $f_{\geq}(F) = q^{\dim F}$  points on F, we have

$$q^{\dim F} = \sum_{G \ge F} f_{=}(G)$$

which inverts to

$$f_{=}(F) = \sum_{G \ge F} \mu(F, G) q^{\dim G}$$

Setting  $F = \hat{0}$  gives the desired result.

1. For each flat F let  $\mathcal{A}/F$  be the arrangement inside F obtained by intersecting the hyperplanes  $\mathcal{A} - F$  with F. The arrangement subdivides each flat F into (relatively) open faces, namely, the

regions of  $\mathcal{A}/G$  for the flats  $G \geq F$ . Since the Euler characteristic of  $F \cong \mathbb{R}^{\dim F}$  is  $(-1)^{\dim F}$ , we get

$$(-1)^{\dim F} = \sum_{G \ge F} (-1)^{\dim G} a(\mathcal{A}/G),$$

which inverts to

$$(-1)^{\dim F}a(\mathcal{A}/F) = \sum_{G \ge F} \mu(F,G)(-1)^{\dim G}.$$

Setting  $F = \hat{0}$  gives the desired result. The same strategy works for  $b(\mathcal{A})$ , using the result that the union of the bounded faces of  $\mathcal{A}$  is contractible, and hence has Euler characteristic equal to 1. [BLVS<sup>+</sup>99, Theorem 4.5.7(b)]

2. is beyond the scope of this writeup; see [GM88, OT92].

# 6.3 **Properties of the characteristic polynomial.**

Whitney's formula and the Tutte polynomial. Since the intersection poset  $\mathcal{A}$  is constructed from its atoms (the hyperplanes of  $\mathcal{A}$ ), it is natural to invoke the Crosscut Theorem 4.7.3 and obtain the following result.

**Theorem 6.3.1.** (Whitney's Theorem) The characteristic polynomial of an arrangement  $\mathcal{A}$  is given by

$$\chi_{\mathcal{A}}(q) = \sum_{\substack{\mathcal{B} \subseteq \mathcal{A} \\ \mathcal{B} \text{ central}}} (-1)^{|\mathcal{B}|} q^{n-r(\mathcal{B})}$$

where  $r(\mathcal{B}) = n - \dim \bigcap_{H \in \mathcal{B}} H$ .

The **Tutte polynomial** is another important polynomial associated to an arrangement:

$$T_{\mathcal{A}}(x,y) = \sum_{\substack{\mathcal{B} \subseteq \mathcal{A} \\ \mathcal{B} \text{ central}}} (x-1)^{r-r(\mathcal{B})} (y-1)^{|\mathcal{B}|-r(\mathcal{B})}.$$

Whitney's Theorem can then be rephrased as:

$$\chi_{\mathcal{A}}(q) = (-1)^r q^{n-r} T_{\mathcal{A}}(1-q,0).$$

#### 6.3.1 Deletion and contraction.

A common technique for inductive arguments in hyperplane arrangement  $\mathcal{A}$  is to choose a hyperplane H and study how  $\mathcal{A}$  behaves without H (in the deletion  $\mathcal{A} \setminus H$ ) and how H interacts with the rest of  $\mathcal{A}$  (in the contraction  $\mathcal{A}/H$ ).

For a hyperplane H of an arrangement  $\mathcal{A}$  in V, the **deletion** 

$$\mathcal{A} \backslash H = \{ A \in \mathcal{A} \, : \, A \neq H \}$$

is the arrangement in V consisting of the hyperplanes other than H, and the contraction

$$\mathcal{A}/H = \{A \cap H : A \in \mathcal{A}, A \neq H\}$$

is the arrangement in H consisting of the intersections of the other hyperplanes with H. Figure 6.3 shows a hyperplane arrangement  $\mathcal{A}$  in  $\mathbb{R}^3$  and the deletion  $\mathcal{A}\backslash 4$  and contraction  $\mathcal{A}/4$ .



Figure 6.3: An arrangement and its deletion and contraction.

**Proposition 6.3.2.** (Deletion-Contraction) If  $\mathcal{A}$  is an arrangement and H is a hyperplane in  $\mathcal{A}$  then

$$\chi_{\mathcal{A}}(q) = \chi_{\mathcal{A} \setminus H}(q) - \chi_{\mathcal{A}/H}(q)$$

*Proof.* Whitney's formula gives

$$\chi_{\mathcal{A}}(q) = \sum_{\substack{H \notin \mathcal{B} \subseteq \mathcal{A} \\ \mathcal{B} \text{ central}}} (-1)^{|\mathcal{B}|} q^{n-r(\mathcal{B})} + \sum_{\substack{H \in \mathcal{B} \subseteq \mathcal{A} \\ \mathcal{B} \text{ central}}} (-1)^{|\mathcal{B}|} q^{n-r(\mathcal{B})} = \chi_{\mathcal{A} \setminus H}(q) - \chi_{\mathcal{A}/H}(q),$$

where we use the fact that if  $H \in \mathcal{B}$  then  $r(\mathcal{B}) = r_{\mathcal{A}/H}(\mathcal{B}\backslash H) + 1$ .

Unfortunately hyperplane arrangements are not closed under contraction; in the example of Figure 6.3, the image of 3 in  $(\mathcal{A}/1)/2$  is not a hyperplane. In particular, we only apply the deletion-contraction formula when there is only one copy of H in  $\mathcal{A}$ .

These small but annoying difficulties are elegantly solved by working in the wider class of "matroids" (for central arrangements) or semimatroids (for affine arrangements).

We leave it to the reader to check that the characteristic polynomial of the cone  $c\mathcal{A}$  can be expressed as follows.

**Proposition 6.3.3.** If cA is the cone of arrangement A then

$$\chi_{c\mathcal{A}}(q) = (q-1)\chi_{\mathcal{A}}(q).$$

#### 6.3.2 Sign alternation and unimodality.

More can be said about the individual coefficients of the characteristic polynomial.

**Theorem 6.3.4.** The characteristic polynomial of an arrangement

$$\chi_{\mathcal{A}}(q) = q^n - a_{n-1}q^{n-1} + a_{n-2}q^{n-2} - \dots + (-1)^n a_n q^0$$

has coefficients alternating signs, so  $a_i \ge 0$ . Furthermore, the coefficients are unimodal and even log-concave, that is,

$$a_1 \le a_2 \le \dots \le a_{i-1} \le a_i \ge a_{i+1} \ge \dots \ge a_n$$
 for some *i*, and  
 $a_{j-1}a_{j+1} \le a_j^2$  for all *j*.

*Proof.* The sign alternation is easily proved by induction using deletion-contraction. The second result is much deeper. It was conjectured by Rota in 1970 [Rot71], and recently proved by Huh [Huh12] for fields of characteristic 0 and by Huh and Katz [HK12] for arbitrary fields, drawing from toric geometry, tropical geometry, and matroid theory. This result is conjectured to be true for any geometric lattice; this is still open.

# 6.3.3 Graphs and proper colorings.

There is a special case of interest, corresponding to arrangements coming from graphs. To each graph G on vertex set [n] we associate the **graphical arrangement** in  $\mathbb{k}^n$ , consisting of the hyperplanes  $x_i = x_j$  for all edges ij of G.

Given q colors, a **proper** q-coloring of G assigns a color to each vertex of G so that two vertices i and j which share an edge ij in G must have different colors. The number of proper q-colorings of G is given by the **chromatic polynomial**  $\chi_G(q)$ .

Say an orientation of the edges of G is **acyclic** if it creates no directed cycles.

**Theorem 6.3.5.** The chromatic polynomial of a graph G equals the characteristic polynomial of its graphical arrangement  $\mathcal{A}_G$ 

$$\chi_{\mathcal{A}_G(q)} = \chi_G(q).$$

and the number of acyclic orientations of G equals the number of regions of  $\mathcal{A}_G$  in  $\mathbb{R}^n$ .

*Proof.* A proper q-coloring of G is equivalent to a point  $\mathbf{a} \in \mathbb{F}_q^n$  which is on none of the hyperplanes  $x_i = x_j$  for  $ij \in G$ . Also, the arrangement  $\mathcal{A}_G$  in  $\mathbb{k}^n$  has the same intersection lattice for any field  $\mathbb{k}$ . (This is not true for every arrangement.) The first equality then follows from Theorem 6.2.1.3.

Given a region R of  $\mathcal{A}_G$  we give each edge ij the orientation  $i \to j$  if  $x_i > x_j$  in R, and  $i \leftarrow j$  if  $x_i < x_j$  in R. This is a bijection between the regions of  $\mathcal{A}_G$  and the acyclic orientations of G.  $\Box$ 

In particular, this theorem proves that  $\chi_G(q)$  is indeed given by a polynomial in q when q is a positive integer. It also gives us a reciprocity theorem, telling us what happens when we irresponsibly substitute the negative integer q = -1 into this polynomial:

**Corollary 6.3.6.** The graph G has  $|\chi_G(-1)|$  acyclic orientations of G.

## 6.3.4 Free arrangements.

Far more often than we might expect, characteristic polynomials of hyperplane arrangements factor as products of linear forms. [Sag99] Such factorizations are often a manifestation of the underlying algebraic structure. The theory of free arrangements gives one possible explanation for this phenomenon.

Let  $\mathcal{A}$  be a real central arrangement and let  $R = \mathbb{R}[x_1, \ldots, x_n]$  be the polynomial ring in n variables, graded by total degree. A **derivation** is a linear map D satisfying Leibniz's law

$$D(fg) = f(Dg) + (Df)g$$
 for all  $f, g \in R$ 

The set Der of derivations is an *R*-module; that is, if  $p \in R$  and  $D \in$  Der then  $pD \in$  Der. It is a graded module, where *D* is homogeneous of degree *d* if it takes polynomials of degree *k* to polynomials of degree k + d. It is a free module with basis  $\frac{\partial}{\partial x_1}, \dots, \frac{\partial}{\partial x_n}$ ; that is,

$$Der = \left\{ p_1 \frac{\partial}{\partial x_1} + \dots + p_n \frac{\partial}{\partial x_n} : p_i \in R \right\}$$

Indeed, if D is a derivation with  $Dx_i = p_i \in R$  for  $1 \leq i \leq n$ , then  $D = p_1 \frac{\partial}{\partial x_1} + \cdots + p_n \frac{\partial}{\partial x_n}$  by linearity.

Now consider the submodule of  $\mathcal{A}$ -derivations:

$$Der(\mathcal{A}) = \{ D \in Der : \alpha_H \mid D(\alpha_H) \text{ for all } H \in A \}$$

where  $\alpha_H$  is a linear form defining hyperplane H, so  $H = \{v \in \mathbb{k}^n : \alpha_H(v) = 0\}$ . Let  $Q = \prod_{H \in \mathcal{A}} \alpha_H$  be the **defining polynomial** of  $\mathcal{A}$ .

For example, if  $\mathcal{H}$  is the arrangement of coordinate hyperplanes with defining polynomial  $Q_{\mathcal{H}} = x_1 \cdots x_n$ , then  $E_i = x_i \frac{\partial}{\partial x_i}$  is an  $\mathcal{H}$ -derivation for  $1 \leq i \leq n$  since  $E_i(x_i) = x_i$  and  $E_i(x_j) = 0$  for  $j \neq i$ .

We say the arrangement  $\mathcal{A}$  is **free** if  $\text{Der}(\mathcal{A})$  is a free *R*-module. This notion is not well understood at the moment; it is not even known if it is a combinatorial condition. For example, Ziegler gave an example of a free arrangement over  $\mathbb{F}_2$  and a non-free arrangement over  $\mathbb{F}_3$  with isomorphic intersection posets.

**Conjecture 6.3.1.** [Ter80] If two real arrangements  $A_1$  and  $A_2$  have isomorphic intersection posets and  $A_1$  is free, then  $A_2$  is free.

Mysterious as it is, freeness is a very useful property.

**Theorem 6.3.7.** (Terao's Factorization Theorem) [Ter80] If  $\mathcal{A}$  is free then  $D(\mathcal{A})$  has a homogeneous basis  $D_1, \ldots, D_n$  whose degrees  $d_1, \ldots, d_n$  only depend on  $\mathcal{A}$ . Furthermore, the Poincaré polynomial of  $V(\mathcal{A})$  is  $q^n \chi_{\mathcal{A}}(-1/q)$  where

$$\chi_A(q) = (q - d_1 - 1) \cdots (q - d_n - 1).$$

Freeness is made more tractable thanks to the following two useful criteria.

**Theorem 6.3.8.** (Saito Criterion) [Sai80] Let  $D_1, \ldots, D_n$  be  $\mathcal{A}$ -derivations and let  $Q(\mathcal{A})$  be the defining form of  $\mathcal{A}$ . Then  $Der(\mathcal{A})$  is free with basis  $D_1, \ldots, D_n$  if and only if

$$\det(D_i(x_j))_{1 \le i,j \le n} = c \cdot Q_{\mathcal{A}}$$

for some constant c.

**Theorem 6.3.9.** [Ter80] Let  $\mathcal{A}$  be an arrangement and H be a hyperplane of  $\mathcal{A}$ . Any two of the following statements imply the third:

- $\mathcal{A}/H$  is free with exponents  $b_1, \ldots, b_{n-1}$ .
- $\mathcal{A} \setminus H$  is free with exponents  $b_1, \ldots, b_{n-1}, b_n 1$ .
- $\mathcal{A}$  is free with exponents  $b_1, \ldots, b_{n-1}, b_n$

Theorem 6.3.8 can be very easy to use if we have the right candidate for a basis. For example, we saw that  $E_i = x_i \frac{\partial}{\partial x_i}$  is an  $\mathcal{H}_n$ -derivation for  $1 \leq i \leq n$ . The matrix  $E_i(x_j)$  is diagonal with determinant  $x_1 \cdots x_n$ , so this must in fact be a basis for  $\text{Der}(\mathcal{H})$  with exponents  $0, \ldots, 0$ , and

$$\chi_{\mathcal{H}_n}(q) = (q-1)^n.$$

To use Theorem 6.3.9 to prove that an arrangement  $\mathcal{A}$  is free, we need  $\mathcal{A}$  to belong to a larger family of free arrangements with predictable exponents, which behaves well under deletion and contraction. For example, to prove inductively that  $\mathcal{H}_n$  is free, we need the stronger statement that an arrangement of k coordinate hyperplanes in  $\mathbb{k}^n$  (where  $k \leq n$ ) has exponents 0 (k times) and -1 (n - k times). A more interesting example is given in Section 6.4.

### 6.3.5 Supersolvability.

There is a combinatorial counterpart to the notion of freeness that produces similar results. Recall that a lattice L is **supersolvable** if it has an **M-chain** C such that the sublattice generated by C and any other chain is distributive.

In this section we are interested in arrangements whose intersection poset is supersolvable. Since this poset is a geometric lattice, it is semimodular, so the following theorem applies.

**Theorem 6.3.10.** [Sta72] Let L be a finite supersolvable semimodular lattice, and suppose that  $\widehat{0} = t_0 < t_1 < \cdots < t_n = \widehat{1}$  is an M-chain. Let  $a_i$  be the number of atoms s such that  $s \leq t_i$  but  $s \not\leq t_{i-1}$ . Then

$$\chi_L(q) = (q - a_1) \cdots (q - a_n).$$

Graphical arrangements are an interesting special case. Say a graph G is **chordal** if there exists an ordering of the vertices  $v_1, \ldots, v_n$  such that for each *i*, the vertices among  $\{v_1, \ldots, v_{i-1}\}$  which are connected to  $v_i$  form a complete subgraph. An equivalent characterization is that G has no induced cycles of length greater than 3.

It is very easy to compute the chromatic polynomial of a chordal graph. Suppose we wish to assign colors from [q] to  $v_1, \ldots, v_n$  in order, to get a proper q-coloring. If  $v_i$  is connected to  $a_i$  vertices among  $\{v_1, \ldots, v_{i-1}\}$ , since these are all connected pairwise, they must have different colors, so there are exactly  $q - a_i$  colors available for  $b_i$ . It follows that  $\chi_G(q) = (q - a_1) \cdots (q - a_n)$ . The similarity in these formulas is not a coincidence.

**Theorem 6.3.11.** [Sta72] The intersection lattice of the graphical arrangement  $\mathcal{A}_G$  is supersolvable if and only if the graph G is chordal.

# 6.4 Computing the characteristic polynomial.

The results of the previous section, and Theorem 6.2.1 in particular, show the importance of computing  $\chi_{\mathcal{A}}(q)$  for arrangements of interest. In this section we discuss the most common techniques for doing this. **Computing the Möbius function directly.** In Section 4.7 we saw many techniques for computing Möbius functions, and we can use them to compute  $\chi_{\mathcal{A}}(x)$ .

1. (Generic arrangement)  $\mathcal{A}_{n,r}$ : *n* generic hyperplanes in  $\mathbb{k}^r$ .

Consider a **generic** arrangement of n hyperplanes in  $\mathbb{k}^r$ , where any  $k \leq r$  hyperplanes have an intersection of codimension k. There are  $\binom{n}{m}$  flats of rank m, and for each flat F we have  $[\widehat{0}, F] \cong 2^{[m]}$ , so  $\mu(\widehat{0}, F) = (-1)^m$ . Therefore the characteristic polynomial is

$$\chi_{\mathcal{A}_{n,r}}(x) = \sum_{m=0}^{r} (-1)^m \binom{n}{m} x^m$$

and the number of regions and bounded regions are

$$a(\mathcal{A}_{n,r}) = \binom{n}{r} + \binom{n}{r-1} + \dots + \binom{n}{0},$$
  
$$b(\mathcal{A}_{n,r}) = \binom{n}{r} - \binom{n}{r-1} + \dots \pm \binom{n}{0} = \binom{n-1}{r}.$$

This method works in some examples, but when there is a nice formula for the characteristic polynomial, this is usually not the most efficient technique.

The finite field method. Theorem 6.2.1.3 is about arrangements over finite fields, but it may also be used as a powerful technique for computing  $\chi_{\mathcal{A}}(x)$  for real or complex arrangements. The idea is simple: most arrangements  $\mathcal{A}$  we encounter "in nature" (that is, in mathematics) are given by equations with integer coefficients.

We can use the equations of  $\mathcal{A}$  to determine an arrangement  $\mathcal{A}_q$  over a prime q. For large enough q, the arrangements  $\mathcal{A}$  and  $\mathcal{A}_q$  will have the same intersection poset, so

$$\chi_{\mathcal{A}}(q) = |\mathbb{F}_q^n \setminus \bigcup_{H \in \mathcal{A}_q} H|.$$

We have thus reduced the computation of  $\chi_A$  to an enumerative problem in a finite field. By now we are pretty good at counting, and we can solve these problems for many arrangements of interest.

2. (Coordinate arrangement)  $\mathcal{H}_n$ :  $x_i = 0$   $(1 \le i \le n)$ Here  $\chi_{\mathcal{H}_n}(q)$  is the number of *n*-tuples  $(a_1, \ldots, a_n) \in \mathbb{F}_q^n$  with  $a_i \ne 0$  for all *i*, so

$$\chi_{\mathcal{H}_n}(q) = (q-1)^n, \qquad a(\mathcal{H}_n) = 2^n$$

There is an easy bijective proof for the number of regions: each region R of  $\mathcal{H}_n$  is determined by whether  $x_i < 0$  or  $x_i > 0$  in R.

3. (Braid arrangement)  $\mathcal{A}_{n-1}$ :  $x_i = x_j$   $(1 \le i < j \le n)$ .

Here  $\chi_{\mathcal{A}_{n-1}}(q)$  is the number of *n*-tuples  $(a_1, \ldots, a_n) \in \mathbb{F}_q^n$  such that  $a_i \neq a_j$  for  $i \neq j$ . Selecting them in order,  $a_i$  can be any element of  $\mathbb{F}_q$  other than  $a_1, \ldots, a_{i-1}$ , so

$$\chi_{\mathcal{A}_{n-1}}(q) = q(q-1)(q-2)\cdots(q-n+1), \qquad a(\mathcal{A}_{n-1}) = n$$

A region is determined by whether  $x_i > x_j$  or  $x_i < x_j$  for all  $i \neq j$ ; that is, by the relative linear order of  $x_1, \ldots, x_n$ .

4. (Threshold arrangement)  $\mathcal{T}_n$ :  $x_i + x_j = 0$ ,  $(1 \le i < j \le n)$ .

Let q be an odd prime. We need to count the points  $\mathbf{a} \in \mathbb{F}_q^n$  on none of the hyperplanes, that is, those satisfying  $a_i + a_j \neq 0$  for all  $i \neq j$ . To specify one such point, we may first choose the partition  $[n] = S_0 \sqcup S_1 \sqcup \cdots \sqcup S_{(q-1)/2}$  where  $S_i$  is the set of positions j such that  $a_j \in \{i, -i\}$ . Note that  $S_0$  can have at most one element. Then, for each non-empty block  $S_i$  with i > 0we need to decide whether  $a_j = i$  or  $a_j = -i$  for all  $j \in S_i$ . The techniques of Section 2.3 then give

$$\sum_{n>0} \chi_{\tau_n}(x) \frac{z^n}{n!} = (1+z)(2e^z - 1)^{(x-1)/2}.$$

The regions of  $\mathcal{T}_n$  are in bijection with the **threshold graphs** on [n]. These are the graphs for which there exist vertex weights w(i) for  $1 \le i \le n$  and a "threshold" w such that edge ijis present in the graph if and only if w(i) + w(j) > w. Threshold graphs have many interesting properties and applications; see [MP95].

There are many interesting deformations of the braid arrangement, obtained by considering hyperplanes of the form  $x_i - x_j = a$  for various constants a. The left panel of Figure 6.4 shows the braid arrangement  $A_2$ . This is really an arrangement in  $\mathbb{R}^3$ , but since all hyperplanes contain the line x = y = z, we draw its essentialization by intersecting it with the plane x + y + z = 0. Similarly, the other panels show the Catalan, Shi, Ish, and Linial arrangements, which we now discuss.



Figure 6.4: The arrangements  $\mathcal{A}_2$ , Cat<sub>2</sub>, Shi<sub>2</sub>, and Ish<sub>2</sub>.

6. (Catalan arrangement)  $Cat_{n-1}$ :  $x_i - x_j = -1, 0, 1$   $(1 \le i < j \le n).$ 

To compute  $\chi_{\operatorname{Cat}_{n-1}}(q)$ , we need to count the *n*-tuples  $\mathbf{a} = (a_1, \ldots, a_n) \in \mathbb{F}_q^n$  where  $a_i$  and  $a_j$  are never equal or adjacent modulo q. There are q choices for  $a_1$ , and once we have chosen  $a_1$  we can "unwrap"  $\mathbb{F}_q - \{a_1\}$  into a linear sequence of q - 1 dots. The set  $A = \{a_2, \ldots, a_n\}$  consists of n - 1 non-adjacent dots, and choosing them is equivalent to choosing a partition of q - n into n parts, corresponding to the gaps between the dots; there are  $\binom{q-n-1}{n-1}$  choices. Finally there are (n - 1)! ways to place A in a linear order in  $\mathbf{a}$ . Therefore

$$\chi_{\operatorname{Cat}_{n-1}}(q) = q(q-n-1)(q-n-2)\cdots(q-2n+1)$$
$$a(\operatorname{Cat}_{n-1}) = n!C_n, \qquad b(\operatorname{Cat}_{n-1}) = n!C_{n-1},$$

where  $C_n$  is the *n*th Catalan number. It is not too difficult to show bijectively that each region of the braid arrangement  $\mathcal{A}_{n-1}$  contains  $C_n$  regions of the Catalan arrangement,  $C_{n-1}$  of which are bounded. [Sta07, Section 5.4]

7. (Shi arrangement)  $\operatorname{Shi}_{n-1}$ :  $x_i - x_j = 0, 1$   $(1 \le i < j \le n)$ .

Consider a point  $\mathbf{a} = (a_1, \ldots, a_n) \in \mathbb{F}_q^n$  not on any of the Shi hyperplanes. Consider dots  $0, 1, \ldots, q-1$  around a circle, and mark dot  $a_i$  with the number *i* for  $1 \leq i \leq n$ . Mark the remaining dots  $\bullet$ . Now let *w* be the word of length *q* obtained by reading the labels clockwise, starting at the label 1. Note that each block of consecutive numbers must be listed in increasing order. By recording the sets between adjacent  $\bullet$ s, and dropping the initial 1, we obtained an ordered partition  $\Pi$  of  $\{2, 3, \ldots, n\}$  into q - n parts. There are  $(q - n)^{n-1}$  such partitions. Figure 6.5 shows an example of this construction.



Figure 6.5:  $\mathbf{a} = (2, 7, 3, 0, 8, 4) \in \mathbb{F}_{11}^6, \ w = 136 \bullet \bullet 25 \bullet \bullet 4 \bullet, \ \Pi = (\{3, 6\}, \emptyset, \{2, 5\}, \emptyset, \{4\}).$ 

To recover a from this partition, we only need to know where to put marking 1 on the circle; that is, we need to know  $a_1$ . It follows that

$$\chi_{\mathrm{Shi}_{n-1}}(q) = q(q-n)^{n-1}$$
$$a(\mathrm{Shi}_{n-1}) = (n+1)^{n-1}, \qquad b(\mathrm{Shi}_{n-1}) = (n-1)^{n-1}$$

8. (Ish arrangement)  $\operatorname{Ish}_{n-1}$ :  $x_i = x_j$ ,  $x_1 - x_j = i$   $(1 \le i < j \le n)$ .

To choose a point **a** in the complement of the Ish arrangement over  $\mathbb{F}_q$ , choose  $a_1 \in \mathbb{F}_q$ , and then choose  $a_n, a_{n-1}, \ldots, a_2$  subsequently. At each step we need to choose  $a_i \notin \{a_1, a_1 - 1, \ldots, a_1 - i + 1, a_n, a_{n-1}, \ldots, a_{i+1}\}$ . These *n* forbidden values are distinct, so

$$\chi_{\mathrm{Ish}_{n-1}}(q) = q(q-n)^{n-1},$$
  
$$a(\mathrm{Ish}_{n-1}) = (n+1)^{n-1}, \qquad b(\mathrm{Ish}_{n-1}) = (n-1)^{n-1}.$$

The Shi and Ish arrangements share several features, which are generally easier to verify for the Ish arrangement. Together, they give a nice description of the "q, t-Catalan numbers". [Arm13]

9. (Linial arrangement)  $\mathcal{L}_{n-1}$ :  $x_i - x_j = 1$ ,  $(1 \le i < j \le n)$ .

The Linial arrangement has characteristic polynomial

$$\chi_{\mathcal{L}_{n-1}}(q) = \frac{q}{2^n} \sum_{k=0}^n \binom{n}{k} (q-k)^{n-1}$$

The number of regions of  $\mathcal{L}_{n-1}$  equals the number of **alternating trees** with vertex set [n + 1], where each vertex is either smaller than all its neighbors or greater than all its neighbors. There are two different proofs, using Whitney's Theorem [PS00] and the finite field method [Ath99] respectively. Both are somewhat indirect, combining combinatorial tricks and algebraic manipulations. To date, there is no known bijection between the regions of the Linial arrangement and alternating trees.

Here are a few other nice examples.

10. (Coxeter arrangement)  $\mathcal{BC}_n$ :  $x_i \pm x_j = 0$ ,  $x_i = 0$   $(1 \le i < j \le n)$ .

We need to count the *n*-tuples  $(a_1, \ldots, a_n) \in \mathbb{F}_q^n$  with  $a_i \neq a_j$  for all  $i \neq j$ . If we choose  $a_1, \ldots, a_n$  successively, we find there are q - 2i + 1 choices for  $a_i$ , namely, any number other than  $0, \pm a_1, \ldots, \pm a_{i-1}$  (which are all distinct). Therefore

$$\chi_{\mathcal{BC}_n}(q) = (q-1)(q-3)\cdots(q-2n+3)(q-2n+1), \qquad r(\mathcal{BC}_n) = 2^n \cdot n!.$$

11. (Coxeter arrangement)  $\mathcal{D}_n$ :  $x_i \pm x_j = 0$   $(1 \le i < j \le n)$ .

Here there are  $(q-1)(q-3)(q-5)\cdots(1-2n+3)(q-2n+1)$  *n*-tuples with no  $a_i$  equal to 0, and  $n(q-1)(q-3)\cdots(q-2n+3)$  *n*-tuples with one  $a_i$  equal to 0. Therefore

$$\chi_{\mathcal{D}_n}(q) = (q-1)(q-3)\cdots(q-2n+3)(q-n+1), \qquad r(\mathcal{D}_n) = 2^{n-1}\cdot n^{n-1}$$

12. (Finite projective space)  $\mathcal{A}(p,n)$ : all linear hyperplanes in  $\mathbb{F}_p^n$ 

The equations  $\sum_i c_i x_i = 0$  (where  $c_i \in \mathbb{F}_p$ ) define an arrangement over  $\mathbb{F}_q$  where  $q = p^k$ , which has the same intersection poset. Recall that  $\mathbb{F}_q$  is a k-dimensional vector space over  $\mathbb{F}_p$ . Now let us count the points  $(a_1, \ldots, a_n)$  in  $\mathbb{F}_q$  which are not on any hyperplane. We choose  $a_1, \ldots, a_n$  subsequently and at each step we need  $a_i \notin \operatorname{span}_{\mathbb{F}_p}(a_1, \ldots, a_{n-1})$ . Therefore

$$\chi_{\mathcal{A}(p,n)}(q) = (q-1)(q-p)(q-p^2)\cdots(q-p^{n-1}).$$

See [BR, Mph00].

13. (All-subset arrangement)  $\operatorname{All}_n$ :  $\sum_{i \in A} x_i = 0$   $(A \subseteq [n], A \neq \emptyset, [n]).$ 

This arrangement appears naturally in combinatorics, representation theory, algebraic geometry, and physics, among many other contexts. [BMM<sup>+</sup>12] To date, we do not know a simple formula for the characteristic polynomial, though we do have a nice bound. [BMM<sup>+</sup>12].

The reduction of All<sub>n</sub> modulo 2 is the arrangement  $\mathcal{A}(2,n)$  in  $\mathbb{F}_2^n$  considered above. The map from All<sub>n</sub> to  $\mathcal{A}(2,n)$  changes the combinatorics; for example, the subset of hyperplanes  $x_1 + x_2 = 0, x_2 + x_3 = 0, x_3 + x_1 = 0$  decreases from rank 3 to rank 2. However, this map is a rank-preserving *weak map*, in the sense that the rank of a subset never increases, and the total rank stays the same. This implies [KN86, Cor. 9.3.7] that the coefficients of  $\chi_{\text{All}_n}(q)$  are greater than the respective coefficients of  $\chi_{\mathcal{A}(2,n)}(q)$  in absolute value, so

$$r(\operatorname{All}_n) = |\chi_{\operatorname{All}_n}(-1)| > |\chi_{\mathcal{A}(2,n)}(-1)| = \prod_{i=0}^{n-2} (2^i + 1) > 2^{\binom{n-1}{2}}$$

It is also known that  $r(All_n) < 2^{(n-1)^2}$ ; see [BMM<sup>+</sup>12].

Whitney's formula. We can sometimes identify combinatorially the terms in Whitney's Theorem 6.3.1 to obtain a useful formula for the characteristic polynomial.

14. (Coordinate arrangement)  $\mathcal{H}_n$ :  $x_i = 0$   $(1 \le i \le n)$ Whitney's formula gives

$$\chi_{\mathcal{H}_n}(q) = \sum_{A \subseteq [n]} (-1)^{|A|} q^{n-|A|} = (q-1)^n.$$

15. (Generic deformation of  $\mathcal{A}_n$ )  $\mathcal{G}_n$ :  $x_i - x_j = a_{ij}$   $(1 \le i < j \le n, a_{ij} \text{ generic })$ Let  $H_{ij}$  represent the hyperplane  $x_i - x_j = a_{ij}$ . By the genericity of the  $a_{ij}$ s, a subarrangement  $H_{i_1j_1}, \ldots, H_{i_kj_k}$  is central if and only if the graph with edges  $i_1j_1, \ldots, i_nj_n$  has no cycles; that is, it is a forest. Therefore

$$\chi_{\mathcal{G}_n}(q) = \sum_{F \text{ forest on } [n]} (-1)^{|F|} q^{n-|F|}, \qquad r(\mathcal{G}_n) = \text{forests}(n)$$

where forests(n) is the number of forests on vertex set [n]. There is no simple formula for this number, though we can use the techniques of Section 2.3 to compute the exponential generating function for  $\chi_{\mathcal{G}_n}(q)$ .

16. (Other deformations of  $\mathcal{A}_n$ ) The same approach can work for any arrangement consisting of hyperplanes of the form  $x_i - x_j = a_{ij}$ , which correspond to edges marked  $a_{ij}$ . If we have enough control over the  $a_{ij}$ s to describe combinatorially which subarrangements are central, we will obtain a combinatorial formula for the characteristic polynomial.

For simple arrangements like the Shi and Catalan arrangement, the finite field method gives slicker proofs than Whitney's formula. However, this unified approach is also very powerful; for many interesting examples, see [PS00].

**Freeness.** Terao's Factorization Theorem 6.3.7 is a powerful algebraic technique for computing characteristic polynomials which factor completely into linear factors.

17. (Braid arrangement, revisited) It is not difficult to describe the  $\mathcal{A}_{n-1}$ -derivations for the braid arrangement  $\mathcal{A}_{n-1}$ . Note that  $F_d = x_1^d \frac{\partial}{\partial x_1} + \cdots + x_n^d \frac{\partial}{\partial x_n}$  is an  $\mathcal{A}_{n-1}$ -derivation because  $F_d(x_i - x_j) = x_i^d - x_j^d$ . By Saito's criterion, since

$$\det(F_i(x_j))_{0 \le i,j \le n-1} = \det(x_j^i)_{1 \le i,j \le n} = \prod_{1 \le i < j \le n} (x_i - x_j) = Q_{\mathcal{A}},$$

 $Der(\mathcal{A})$  is free with basis  $F_0, \ldots, F_{n-1}$  of degrees  $-1, 0, \ldots, n-2$ . Therefore

$$\chi_{\mathcal{A}_n}(q) = q(q-1)\cdots(q-n+1).$$

18. (Coxeter arrangements  $\mathcal{BC}_n$  and  $\mathcal{D}_n$ , revisited) Now  $F_d(x_i + x_j) = x_i^d + x_j^d$ , so  $F_d$  is a  $\mathcal{D}_n$ derivation only for n odd. Fortunately,

$$\det(F_{2i-1}(x_j))_{1 \le i,j,\le n} = \prod_{1 \le i < j \le n} (x_i - x_j)(x_i + x_j),$$

as can be seen by identifying linear factors as in Section 3.2. It follows that  $Der(\mathcal{D}_n)$  is free with basis  $F_1, F_3, F_5, \ldots, F_{2n-1}$  and degrees  $0, 2, 4, \ldots, 2n-2$ .

For  $\mathcal{BC}_n$  we need the additional derivation  $F = x_1 \cdots x_n \sum_i \frac{1}{x_i} \frac{\partial}{\partial x_i}$ . This is indeed a  $\mathcal{BC}_n$ derivation because  $F(x_i \pm x_j) = (x_j \pm x_i)x_1 \cdots x_n/x_ix_j$ . Once again it is easy to check Saito's criterion to see that  $\text{Der}(\mathcal{BC}_n)$  is free with basis  $F_1, F_3, \ldots, F_{2n-3}, F$  and degrees  $0, 2, 4, \cdots, 2n-4, n-2$ .

19. (Coxeter arrangements, in general) A finite reflection group is a finite group generated by reflections through a family of hyperplanes. For example the reflections across the hyperplanes  $x_i = x_j$  of the braid arrangement  $\mathcal{A}_{n-1}$  correspond to the transpositions (ij), and generate the symmetric group  $S_n$ . Every finite reflection group is a direct product of irreducible ones. Here we focus on the **crystallographic** ones, which can be written with integer coordinates. The irreducible groups  $\Phi$  come in three infinite families  $A_n, BC_n, D_n$  and five exceptional groups  $G_2, F_4, E_6, E_7, E_8$ . The corresponding hyperplane arrangements  $\mathcal{A}_{\Phi}$  are the following.

Φ	$\mathbb{k}^n$	equations
$\mathcal{A}_{n-1}$	$\mathbb{k}^n$	$x_i = x_j$
$\mathcal{BC}_n$	$\mathbb{k}^n$	$x_i = \pm x_j,  x_i = 0$
$\mathcal{D}_n$	$\mathbb{k}^n$	$x_i = \pm x_j$
$\mathcal{G}_2$	$k^3$	$x_i = \pm x_j,  2x_i = x_j + x_k$
$\mathcal{F}_4$	$k^4$	$x_i = \pm x_j,  x_i = 0,  \pm x_1 \pm x_2 \pm x_3 \pm x_4 = 0.$
$\mathcal{E}_6$	$\mathbb{k}^9$	$x_i = x_j,$ $\sum 2x_i = \sum x_i \begin{pmatrix} 1 \le a \le 3\\ 4 \le b \le 6\\ 7 \le c \le 9 \end{pmatrix}.$
$\mathcal{E}_7$	k <sup>8</sup>	$x_{i} = x_{j},  \sum_{i \in A}^{i \in \{a,b,c\}} x_{i} = \sum_{i \notin A}^{i \notin \{a,b,c\}} x_{i}  \begin{pmatrix} A \subseteq [8] \\  A  = 4 \end{pmatrix}$
$\mathcal{E}_8$	$k^8$	$x_i = \pm x_j,  \sum_{i=1}^{8} \epsilon_i x_i = 0  \begin{pmatrix} \epsilon_i = \pm 1, \\ \epsilon_1 \cdots \epsilon_8 = 1 \end{pmatrix}$

Note that arrangement  $\Phi_r$  has rank r, but we have embedded some of them them in  $\mathbb{k}^d$  for higher d in order to obtain nicer equations. Here the indices range over the corresponding  $\mathbb{k}^n$ ; for instance,  $\mathcal{G}_2$  includes the hyperplanes  $2x_1 = x_2 + x_3$ ,  $2x_2 = x_1 + x_3$ ,  $2x_3 = x_1 + x_2$ . We now list the characteristic polynomials and number of regions for these arrangements.

Φ	characteristic polynomial	regions
$\mathcal{BC}_n$	$(q-1)(q-3)(q-5)\cdots(q-2n+3)(q-2n+1)$	$2^n n!$
$\mathcal{D}_n$	$(q-1)(q-3)(q-5)\cdots(q-2n+3)(q-n+1)$	$2^{n-1}n!$
$\mathcal{G}_2$	(q-1)(q-5)	12
$\mathcal{F}_4$	(q-1)(q-5)(q-7)(q-11)	1,152
$\mathcal{E}_6$	(q-1)(q-4)(q-5)(q-7)(q-8)(q-11)	51,840
$\mathcal{E}_7$	(q-1)(q-5)(q-7)(q-9)(q-11)(q-13)(q-17)	2,903,040
$\mathcal{E}_8$	(q-1)(q-7)(q-11)(q-13)(q-17)(q-19)(q-23)(q-29)	696,729,600

It follows from the theory of reflection groups that the number of regions equals the order of the reflection group:

$$r(\mathcal{A}_{\Phi}) = |\Phi|.$$

We already computed in two ways the characteristic polynomials of the classical Coxeter groups  $\mathcal{A}_n, \mathcal{BC}_n$ , and  $\mathcal{D}_n$ . We first gave simple finite field proofs; it would be interesting to do this for the other ("exceptional") groups. We also computed explicit bases for the modules of  $\mathcal{A}$ -derivations; doing this for the other Coxeter groups is not straightforward, but it can be done. [OT92, Appendix B]

However, these beautiful formulas clearly illustrate that there are deeper things at play, and they make it very desirable to have a unified explanation. Indeed, invariant theory produces an elegant case-free proof that all Coxeter arrangement are free; for details, see [OT92, Theorem 6.60].

20. (Shi arrangement) Given its characteristic polynomial, it is natural to guess that the Shi arrangement  $S_{n-1}$  is free. Since we only defined freeness for central arrangements, we consider the cone  $cS_{n-1}$  instead. Following [Ath98], we may prove that this is a free arrangement inductively using Theorem 6.3.9. The strategy is to remove the hyperplanes of  $S_{n-1}$  one at a time, by finding in each step a hyperplane H such that  $A \setminus H$  and A/H are free with predictable exponents. This leads us to consider a more general family of arrangements.

We claim that for any  $m \ge 0$  and  $2 \le k \le n+1$ , the arrangement  $\mathcal{S}_{n-1}^{m,k}$ 

$$\begin{aligned} x_i - x_j &= 0, 1 & \text{for } 2 \le i < j \le n \\ x_1 - x_j &= 0, 1, \dots, m & \text{for } 2 \le j < k \\ x_1 - x_j &= 0, 1, \dots, m, m + 1 & \text{for } k \le j \le n \end{aligned}$$

is free with exponents  $n + m - 1, \ldots, n + m - 1$  (n - k + 1 times),  $n + m - 2, \ldots, n + m - 2$ (k - 2 times), and -1.

Obviously it takes some care to assemble this family of free arrangements and their exponents. However, once we have done this, it is straightforward to prove this more general statement by induction using Theorem 6.3.9, by removing one hyperplane at a time while staying within the family of arrangements  $\mathcal{S}_{n-1}^{m,k}$ . Setting k = 2 and m = 0 we get that the Shi arrangement is free.

Incidentally, Athanasiadis also characterized all the arrangements  $\mathcal{A}$  with  $\mathcal{A}_{n-1} \subseteq \mathcal{A} \subseteq \mathcal{S}_{n-1}$ such that  $c\mathcal{A}$  is free [Ath98]. These arrangements have Ish counterparts which have the same characteristic polynomial, although their cones are not free. [AR12, Corollary 3.3]

# 6.5 The cd-index of an arrangement.

We now discuss a vast strengthening of Zaslavsky's Theorem due to Billera, Ehrenborg, and Readdy. [BER97] They showed that the enumeration of chains of faces of a real hyperplane arrangement  $\mathcal{A}$  (or, equivalently, of a zonotope) depends only on the intersection poset of  $L_{\mathcal{A}}$ . The equivalence with zonotopes is explained by the following result.

**Proposition 6.5.1.** Let A be an arrangement of non-zero real vectors, and let  $\mathcal{A}$  be the arrangement of hyperplanes perpendicular to the vectors of A. Then there is an order-reversing bijection between the faces of the zonotope Z(A) and the faces of the real arrangement  $\mathcal{A}$ .

From this result, it follows easily that the **cd**-indices of Z(A) and  $F(\mathcal{A})$  are reverses of each other; that is, they are related by the linear map  $* : \mathbb{Z}\langle \mathbf{c}, \mathbf{d} \rangle \to \mathbb{Z}\langle \mathbf{c}, \mathbf{d} \rangle$  that reverses any word in



Figure 6.6: A vector arrangement, the normal hyperplane arrangement, and its dual zonotope.

**c** and **d**, so that  $(v_1 \ldots v_k)^* = v_k \ldots v_1$ . To describe these **cd**-indices, we introduce the linear map  $\omega : \mathbb{Z}\langle \mathbf{a}, \mathbf{b} \rangle \to \mathbb{Z}\langle \mathbf{c}, \mathbf{d} \rangle$  obtained by first replacing each occurrence of **ab** with 2**d**, and then replacing every remaining letter with a **c**.

**Theorem 6.5.2.** [BER97] The flag f-vector of a real hyperplane arrangement (or equivalently, of the dual zonotope) depends only on its intersection poset.

More explicitly, let  $\mathcal{A}$  be a real hyperplane arrangement. Let  $F(\mathcal{A})$  be its face poset and  $L_{\mathcal{A}}$  be its intersection poset. The cd-indices of the face poset  $F(\mathcal{A})$  and the dual zonotope  $Z(\mathcal{A})$  are:

$$\Psi_{F(\mathcal{A})}(\mathbf{c},\mathbf{d}) = \left[\omega(\mathbf{a}\,\Phi_{L_{\mathcal{A}}}(\mathbf{a},\mathbf{b}))\right]^*, \qquad \Psi_{Z(A)}(\mathbf{c},\mathbf{d}) = \omega(\mathbf{a}\,\Phi_{L_{\mathcal{A}}}(\mathbf{a},\mathbf{b})),$$

Billera, Ehrenborg, and Readdy [BER97] extended this result to *orientable matroids*, and asked for an interesting interpretation of  $\omega(\mathbf{a} \Phi_L(\mathbf{a}, \mathbf{b}))$  for an arbitrary geometric lattice L.

To illustrate this result we revisit Example 4.8.1, where we computed the **cd**-index of a hexagonal prism. This is the dual zonotope to the arrangement of Figure 6.6, whose intersection poset  $L_{\mathcal{A}}$  is Figure 6.2. The flag f and h-vectors of  $L_{\mathcal{A}}$  are  $(f_{\emptyset}, f_{\{1\}}, f_{\{2\}}, f_{\{1,2\}}) = (1, 4, 4, 9)$  and  $(h_{\emptyset}, h_{\{1\}}, h_{\{2\}}, h_{\{1,2\}}) = (1, 3, 3, 2)$ , so its **ab**-index is  $\Phi_{L_{\mathcal{A}}}(\mathbf{a}, \mathbf{b})) = \mathbf{aa} + 3\mathbf{ab} + 3\mathbf{ba} + 2\mathbf{bb}$ . Therefore

$$\Psi_{Z(\mathcal{A})}(\mathbf{c}, \mathbf{d}) = \omega(\mathbf{a}\mathbf{a}\mathbf{a} + 3\mathbf{a}\mathbf{b}\mathbf{b} + 3\mathbf{a}\mathbf{b}\mathbf{a} + 2\mathbf{a}\mathbf{b}\mathbf{b}) = \mathbf{c}\mathbf{c}\mathbf{c} + 3\mathbf{c}(2\mathbf{d}) + 3(2\mathbf{d})\mathbf{c} + 2(2\mathbf{d})\mathbf{c} = \mathbf{c}^3 + 6\mathbf{c}\mathbf{d} + 10\mathbf{d}\mathbf{c}$$

and

$$\Psi_{F(\mathcal{A})}(\mathbf{c}, \mathbf{d}) = \mathbf{c}^3 + 6\mathbf{d}\mathbf{c} + 10\mathbf{c}\mathbf{d}.$$

in agreement with the computation in Section 4.8.
# 7 Matroids

Matroid theory is a combinatorial theory of independence which has its roots in linear algebra and graph theory, but which turns out to have deep connections with many fields, and numerous applications in pure and applied mathematics. Rota was a particularly enthusiastic ambassador:

"It is as if one were to condense all trends of present day mathematics onto a single finite structure, a feat that anyone would a priori deem impossible, were it not for the fact that matroids do exist." [Rot97]

There are natural notions of independence in linear algebra, graph theory, matching theory, the theory of fields, and the theory of routings, among others. Matroids capture the combinatorial essence that those notions share.

In this section we will mostly focus on enumerative aspects of matroid theory. For a more complete introduction, see [Oxl11, Wel76] and the three volume series [Whi86, Whi87, Whi92].

In Section 7.1 we discuss some combinatorial aspects of vector configurations and linear independence, and introduce some terminology. This example and terminology strongly motivates the definition(s) of a matroid, which we give in Section 7.2. Section 7.3 discusses many important families of matroids arising in algebra, combinatorics, and geometry. Section 7.4 introduces some basic constructions, and Section 7.5 gives a very brief overview of structural matroid theory. The Tutte polynomial is our main enumerative tool; it is introduced in Section 7.6. In Section 7.7 we answer many enumeration problems in terms of Tutte polynomials, and in Section 7.8 we see how we can actually compute the polynomial in some examples of interest. Section 7.9 is devoted to two generalizations: the multivariate and the arithmetic Tutte polynomials. Finally, we give a brief overview of matroid subdivisions in Section 7.10, including a discussion on matroid valuations and the Derksen–Fink invariant.

# 7.1 The main motivating example: Vector configurations and linear matroids.

Let  $E \subset \mathbb{k}^d$  be a finite set of vectors. For simplicity, we assume that E does not contain **0**, does not contain repeated vectors, and spans  $\mathbb{k}^d$ . The **matroid** of E is given by any of the following definitions:

- The independent sets of E are the subsets of E which are linearly independent over k.
- The **bases** of *E* are the subsets of *E* which are bases of  $\mathbb{k}^d$ .
- The circuits of E are the minimal linearly dependent subsets of E.
- The rank function  $r: 2^E \to \mathbb{N}$  is  $r(A) = \dim(\operatorname{span} A)$  for  $A \subseteq E$ .

• The flats F of E are the subspaces of  $\mathbb{k}^d$  spanned by a subset of E. We identify a flat with the set of vectors of E that it contains.

- The lattice of flats  $L_M$  is the poset of flats ordered by containment.
- The matroid (basis) polytope is  $P_M = \operatorname{conv} \{ \mathbf{e}_{b_1} + \dots + \mathbf{e}_{b_r} : \{ b_1, \dots, b_r \}$  is a basis $\} \subset \mathbb{R}^E$ .

More concretely, if we know one of the following objects, we can determine them all: the list  $\mathcal{I}$  of independent sets, the list  $\mathcal{B}$  of bases, the list  $\mathcal{C}$  of circuits, the rank function r, the list  $\mathcal{F}$  of flats, the lattice  $L_M$  of flats, or the matroid polytope  $P_M$ . For this reason, we define the **matroid** of E to be any one (and all) of these objects. A matroid that arises in this way from a set of vectors in a vector space (over  $\mathbf{k}$ ) is called a **linear** or **representable matroid** (over  $\mathbf{k}$ ).

**Example 7.1.1.** Let  $\mathbf{t} = (1, -1, 0)$ ,  $\mathbf{u} = (0, 1, -1)$ ,  $\mathbf{v} = (-1, 0, 1)$ ,  $\mathbf{w} = (1, 1, 1)$  be the vector configuration shown in the left panel of Figure 6.6. Then<sup>12</sup>

 $\mathcal{I} = \{\emptyset, \mathbf{t}, \mathbf{u}, \mathbf{v}, \mathbf{w}, \mathbf{tu}, \mathbf{tv}, \mathbf{tw}, \mathbf{uv}, \mathbf{uw}, \mathbf{vw}, \mathbf{tuw}, \mathbf{tvw}, \mathbf{uvw}\},\$   $\mathcal{B} = \{\mathbf{tuw}, \mathbf{tvw}, \mathbf{uvw}\},\$   $\mathcal{C} = \{\mathbf{tuv}\},\$   $\mathcal{F} = \{\emptyset, \mathbf{t}, \mathbf{u}, \mathbf{v}, \mathbf{w}, \mathbf{tuv}, \mathbf{tw}, \mathbf{uw}, \mathbf{vw}, \mathbf{tuvw}\}$ 

The rank function is  $r(\mathbf{tuvw}) = 3$ ,  $r(\mathbf{tuv}) = 2$ , and r(A) = |A| for all other sets A. The matroid polytope is the equilateral triangle with vertices (1, 1, 0, 1), (1, 0, 1, 1), (0, 1, 1, 1) in  $\mathbb{R}^4$ .

A vector arrangement  $E \subset \mathbb{k}^d$  determines a hyperplane arrangement  $\mathcal{A}$  in  $\mathbb{k}^d$  consisting of the normal hyperplanes  $\{\mathbf{x} \in \mathbb{k}^d : \mathbf{a} \cdot \mathbf{x} = 0\}$  for  $\mathbf{a} \in E$ . The rank function and lattice of flats of E are the same as the rank function and intersection poset of  $\mathcal{A}$ , as defined in Section 6.2. For instance, the lattice of flats of the example above is the same as the intersection poset of the corresponding arrangement, which is illustrated in Figure 6.2.

#### 7.2 Basic definitions.

A simple but important insight of matroid theory is that many of the properties of linear independence (and many other notions of independence in mathematics) are inherently combinatorial. That combinatorial structure is unexpectedly rich, and matroids provide a general combinatorial framework that is ideally suited for exploring it.

A matroid  $M = (E, \mathcal{I})$  consists of a finite set E and a collection  $\mathcal{I}$  of subsets of E such that (I1)  $\emptyset \in \mathcal{I}$ 

(I2) If  $J \in \mathcal{I}$  and  $I \subseteq J$  then  $I \in \mathcal{I}$ .

(I3) If  $I, J \in \mathcal{I}$  and |I| < |J| then there exists  $j \in J - I$  such that  $I \cup j \in \mathcal{I}$ .

The sets in  $\mathcal{I}$  are called **independent**.

The following result, which is an immediate consequence of (I3), shows that this simple definition already captures the notion of dimension:

**Proposition 7.2.1.** In a matroid  $M = (E, \mathcal{I})$ , all the **bases** (the maximal elements of  $\mathcal{I}$ ) have the same size, called the **rank** of the matroid.

Now we develop the combinatorial versions of the definitions of the previous section. Let  $M = (E, \mathcal{I})$  be a matroid.

- An independent set is a set in  $\mathcal{I}$ .
- A **basis** is a maximal independent set.
- The rank function  $r: 2^E \to \mathbb{N}$  is r(A) =(size of the largest independent subset of A).
- A circuit is a minimal dependent set; i.e., a minimal set not in  $\mathcal{I}$ .
- A flat is a set F such that  $r(F \cup e) > r(F)$  for all  $e \notin F$ .
- The lattice of flats  $L_M$  is the poset of flats ordered by containment.
- The matroid (basis) polytope  $P_M = \operatorname{conv} \{ \mathbf{e}_{b_1} + \dots + \mathbf{e}_{b_r} : \{ b_1, \dots, b_r \}$  is a basis}.

Again, we let  $\mathcal{B}$ , r,  $\mathcal{C}$  and  $\mathcal{F}$  denote the set of bases, the rank function, the set of circuits, and the set of flats of M. If we know the ground set E and any one of  $\mathcal{B}$ , r,  $\mathcal{C}$ ,  $\mathcal{F}$ ,  $L_M$ , or  $P_M^{13}$ , we

<sup>&</sup>lt;sup>12</sup>We are omitting brackets for clarity, so for example we write  $\mathbf{tuv}$  for  $\{\mathbf{t}, \mathbf{v}, \mathbf{v}, \}$ .

<sup>&</sup>lt;sup>13</sup>assuming that we know the labels of the elements of  $L_M$  and the embedding of the polytope  $P_M$  in  $\mathbb{R}^E$ .

know them all. For that reason, we call any one (and all) of  $(E, \mathcal{I}), (E, \mathcal{B}), (E, r), (E, \mathcal{C})$  and  $(E, \mathcal{F})$ "the matroid M". Each one of these points of view has its own axiomatization:

**Proposition 7.2.2.** We have the following characterizations of the possible bases  $\mathcal{B}$ , rank function r, circuits  $\mathcal{C}$ , flats  $\mathcal{F}$ , and matroid polytopes  $P_M$  of a matroid.

- The collection B ⊆ 2<sup>E</sup> is the set of bases of a matroid if and only if
   (B1) B is nonempty.
   (B2) If B<sub>1</sub>, B<sub>2</sub> ∈ B and b<sub>1</sub> ∈ B<sub>1</sub> − B<sub>2</sub>, there exists b<sub>2</sub> ∈ B<sub>2</sub> − B<sub>1</sub> such that (B<sub>1</sub> − b<sub>1</sub>) ∪ b<sub>2</sub> ∈ B.
- 2. The function r: 2<sup>E</sup> → N is the rank function of a matroid if and only if:
  (R1) 0 ≤ r(A) ≤ |A| for all A ⊆ E.
  (R2) If A ⊆ B ⊆ E then r(A) ≤ r(B).
  - (R3)  $r(A) + r(B) \ge r(A \cup B) + r(A \cap B)$  for all  $A, B \subseteq E$ .
- The collection C ⊆ 2<sup>E</sup> is the set of circuits of a matroid if and only if (C1) Ø ∉ C.
  - (C2) If  $C \in \mathcal{C}$  and  $C \subsetneq D$  then  $D \notin \mathcal{C}$ .

(C3) If  $C_1, C_2 \in \mathcal{C}$  and  $e \in C_1 \cap C_2$  then there exists  $C \in \mathcal{C}$  with  $C \subseteq C_1 \cap C_2 - e$ .

- 4. The poset L is the lattice of flats of a matroid if and only if it is a geometric lattice.
- The collection I ⊆ 2<sup>E</sup> is the set of independent sets of a matroid if and only if
   (I1) Ø ∈ I.
  - (I2) If  $J \in \mathcal{I}$  and  $I \subseteq J$  then  $I \in \mathcal{I}$ .
  - (I3') For every weight function  $w: E \to \mathbb{R}$ , the following greedy algorithm:

Start with  $B = \emptyset$ . Then, at each step, add to B an element  $e \notin B$  of minimum weight w(e) such that  $B \cup e \in \mathcal{I}$ . Stop when B is maximal in  $\mathcal{I}$ .

produces a maximal element B of  $\mathcal{I}$  of maximum weight  $w(B) = \sum_{b \in B} w(b)$ .

6. [GGMS87] The collection  $\mathcal{B} \subseteq 2^E$  is the set of bases of a matroid if and only if every edge of the polytope conv{ $\mathbf{e}_{b_1} + \cdots + \mathbf{e}_{b_r} : \{b_1, \ldots, b_r\} \in \mathcal{B}$ } in  $\mathbb{R}^E$  is a translate of  $\mathbf{e}_i - \mathbf{e}_j$  for some  $i, j \in E$ .

This proposition allows us to give six other definitions of matroids in terms of bases, ranks, circuits, flats, independent sets, and polytopes. The first three axiom systems are "not too far" from each other, but it is useful to have them all. For instance, to prove that linear matroids are indeed matroids, it is much easier to check the circuit axioms (C1)-(C3) than the independence axioms (I1)-(I3). The last three axiom systems suggest deeper, fruitful connections to posets, optimization, and polytopes.

In fact, there are several other equivalent (and useful) definitions of matroids. It is no coincidence that matroid theory gave birth to the notion of a **cryptomorphism**, which refers to the equivalence of different axiomatizations of the same object. This feature of matroid theory can be frustrating at first; but as one goes deeper, it becomes indispensable and extremely powerful. A

very valuable resource is [Bry86], a multilingual cryptomorphism dictionary that translates between these different points of view.

We say  $(E, \mathcal{B})$  and  $(E', \mathcal{B}')$  are **isomorphic** if there is a bijection between E and E' that induces a bijection between  $\mathcal{B}$  and  $\mathcal{B}'$ . A **loop** of M is an element  $e \in E$  of rank 0, so  $\{e\}$  is dependent. A **coloop** is an element e which is independent of E - e, so r(E - e) = r - 1. Non-loops  $\mathbf{e}$  and  $\mathbf{f}$ are **parallel** if  $r(\mathbf{ef}) = 1$ ; they are indistinguishable inside the matroid. A matroid is **simple** if it contains no loops or parallel elements. The map  $M \mapsto L_M$  is a bijection between simple matroids and geometric lattices.

# 7.3 Examples.

We now describe a few contexts where matroids arise naturally. These statements are nice (and not always trivial) exercises in their respective fields. Most proofs may be found, for example, in [Ox111]. We give references for the rest.

- 0. (Uniform matroids) For  $n \ge k \ge 1$ , the **uniform matroid**  $U_{k,n}$  is the matroid on [n] where every k-subset is a basis. It is the linear matroid of n generic vectors in a k-dimensional space.
- 1a. (Vector configurations) Let E be a set of vectors in  $\mathbb{k}^d$  and let  $\mathcal{I}$  be the set of linearly independent subsets of E. Then  $(E, \mathcal{I})$  is a matroid. Such a matroid is called **linear** or **representable** (over  $\mathbb{k}$ ). [Mac36, Whi35] There are at least three other ways of describing the same family of matroids, listed in 1b, 1c, and 1d below.
- 1b. (Point configurations) Let E be a set of points in  $\mathbb{k}^d$  and let  $\mathcal{I}$  be the set of affinely independent subsets of E; that is, the subsets  $\{\mathbf{a}_1, \ldots, \mathbf{a}_k\}$  whose affine span is k-dimensional. Then  $(E, \mathcal{I})$  is a matroid.
- 1c. (Hyperplane arrangements) Let E be a hyperplane arrangement in  $\mathbb{k}^d$  and let  $r(\{H_1, \ldots, H_k\}) = d \dim(H_1 \cap \cdots \cap H_k)$ . Then (E, r) is a matroid.
- 1d. (Subspaces) Let  $U \subseteq V$  be a subspace of a vector space  $V \cong \mathbb{k}^d$  with a chosen basis. Consider the coordinate subspaces  $V_S = \{\mathbf{v} \in V : v_s = 0 \text{ for } s \in S\}$ . Say  $B \subseteq [d]$  is a basis if  $U \cap V_B = \{0\}$ . Then  $(E, \mathcal{B})$  is a matroid.

Given a matrix over k, the matroid on the columns of A (in the sense of 1a) is the same as the matroid of the rowspace of A (in the sense of 1d).

- 2. (Graphs) Let E be the set of edges of a connected graph G and let C be the set of cycles of G (where each cycle is regarded as the list of its edges). Then (E, C) is a matroid. Such a matroid is called **graphical**. The bases are the spanning trees of G.
- 3. (Field extensions) Let  $\mathbb{F} \subseteq \mathbb{K}$  be two fields, and let E be a finite set of elements in the extension field  $\mathbb{K}$ . Say  $I = \{i_1, \ldots, i_k\} \subseteq E$  is independent if it is algebraically independent over  $\mathbb{F}$ ; that is, if there does not exist a non-trivial polynomial  $P(x_1, \ldots, x_k)$  with coefficients in  $\mathbb{F}$  such that  $P(i_1, \ldots, i_k) = 0$ . Then  $(E, \mathcal{I})$  is a matroid. Such a matroid is called **algebraic**. [vdW37]
- 4. (Matchings) Let  $G = (S \cup T, E)$  be a bipartite graph, so every edge in E joins a vertex of S and a vertex of T. Recall that a **partial matching** of G is a set of edges, no two of which

have a common vertex. Let  $\mathcal{I}$  be the collection of subsets  $I \subseteq T$  which can be matched to S; that is, those for which there exists a partial matching whose edges contain the vertices in I. Then  $(T, \mathcal{I})$  is a matroid. Such a matroid is called **transversal**. [EF65]

- 5. (Routings) Let G = (V, E) be a directed graph and  $B_0$  be an *r*-subset of the vertices. Let  $\mathcal{B}$  be the collection of *r*-subsets of *V* for which there exists a routing (a collection of *r* vertexdisjoint paths) starting at *B* and ending at  $B_0$ . Then  $(V, \mathcal{B})$  is a matroid. Such a matroid is called **cotransversal**. [Mas72]
- 6. (Dyck paths) For each Dyck path P of length n, let  $B \subseteq [2n]$  be the upstep set of P; that is, the set of integers i for which the ith step of P is an upstep. Let  $\mathcal{B}$  be the collection of upstep sets of Dyck paths of length n. Then  $([2n], \mathcal{B})$  is the set of basis of a matroid. This matroid is called a **Catalan matroid**  $\mathbf{C}_n$ . More generally, we may consider the lattice paths of 2n steps (1,1) and (1,-1) which stay between an upper and a lower border; their upstep sets form the bases of a **lattice path matroid**. [Ard03, BdMN03]
- 7. (Schubert matroids) Given  $n \in \mathbb{N}$  and a set of positive integers  $I = \{i_1 < \cdots < i_k\}$ , the sets  $\{j_1 < \cdots < j_k\} \subseteq [n]$  such that  $j_1 \geq i_1, \ldots, j_k \geq i_k$  are the bases of a matroid. These matroids are called **Schubert matroids** due to their connection to the Schubert cells in the Grassmannian  $\operatorname{Gr}(k, n)$ . This simple but important family of matroids has been rediscovered many times under names such as freedom matroids [CS05] and shifted matroids [Ard03, Kli07].
- 8. (Positroids) A **positroid** is a matroid on [n] which can be represented by the columns of a full rank  $d \times n$  matrix such that all its maximal minors are nonnegative. [Pos06] These matroids arise in the study of the totally nonnegative part of the Grassmannian, and have recently found applications in the physics of scattering amplitudes. Remarkably, these matroids can be described combinatorially, and they are in bijection with several interesting classes of objects with elegant enumerative properties. [ARW13b, Oh09, Pos06, Wil05]



Figure 7.1: Many manifestations of the same matroid.

**Example 7.3.1.** The list above gives eight different manifestations of the matroid on  $\{a, b, c, d, e, f\}$ with bases  $\mathcal{B} = \{abc, abd, abe, acd, ace\}$ . Figure 7.1 shows a vector configuration in  $\mathbb{R}^3$  (with  $f = \mathbf{0}$ ) which realizes M as a linear matroid, a graph which realizes it as a graphical matroid, a bipartite graph which realizes it as a transversal matroid, the five Dyck paths of length 3 which show that M is the Catalan matroid  $\mathbb{C}_3$ , and a directed graph with sinks  $B_0 = \{a, c, e\}$  which realize M as a cotransversal matroid. It can be realized as an algebraic matroid in the field extension  $\mathbb{R} \subset \mathbb{R}(x, y, z)$  with  $a = z^3$ , b = x + y, c = x - y, d = xy,  $e = x^2y^2$ , f = 1. It is also isomorphic to the Schubert matroid on [6] for  $I = \{2, 4, 6\}$ . Finally, it is a positroid, since in the realization shown, all five bases are positive: they all satisfy the "right-hand rule" for vectors in  $\mathbb{R}^3$ .

To give a small illustration of the power of matroids, let us revisit the almost trivial observation that all bases of a matroid have the same size. This result has many interesting consequences:

- All bases of a vector space have the same size, called the "dimension" of V.
- All spanning trees of a connected graph have the same number (v-1) of edges.
- All maximal partial matchings of a bipartite graph have the same size.
- All transcendence bases of a field extension have the same size, called the "transcendence degree" of the extension.

Naturally, once we work harder to obtain more interesting results about matroids, we will obtain more impressive results in all of these different fields.

# 7.4 Basic constructions.

We now discuss the notions of duality, minors, direct sums, and connected components in matroids; in Section 7.5 we explain how they generalize similar notions in linear algebra and graph theory.

**Proposition/Definition 7.4.1.** (Duality) Let  $M = (E, \mathcal{B})$  be a matroid, and let  $\mathcal{B}^* = \{E - B : B \in \mathcal{B}\}$ . Then  $M^* = (E, \mathcal{B}^*)$  is also a matroid, called the **dual matroid** of M.

Note that the loops of a matroid M are the coloops of  $M^*$ .

**Proposition/Definition 7.4.2.** (Deletion, contraction) Let  $M = (E, \mathcal{I})$  be a matroid and  $s \in E$ . Let

$$\mathcal{I}' = \{I \in \mathcal{I} : s \notin I\}, \qquad \qquad \mathcal{I}'' = \begin{cases} \{I \subseteq E - s : I \cup s \in I\}, & \text{if $s$ is not a loop} \\ \mathcal{I}, & \text{if $s$ is a loop} \end{cases}$$

Then  $M \setminus s = (E - s, \mathcal{I}')$  and  $M/s = (E - s, \mathcal{I}'')$  are also matroids. They are called the **deletion** of s from M (or the **restriction** of M to E - s) and the **contraction** of s from M.

**Proposition/Definition 7.4.3.** (Minors) Let M be a matroid on E.

1. Deletion and contraction commute; that is, for all  $s \neq t$  in E,

$$(M \setminus s) \setminus t = (M \setminus t) \setminus s, \quad (M/s)/t = (M/t)/s, \quad (M/s) \setminus t = (M \setminus t)/s.$$

2. Let the deletion  $M \setminus S$  (resp. contraction M/S) of a subset  $S \subseteq E$  from M be the successive deletion (resp. contraction) of the individual elements of S. Their rank functions are

$$r_{M\setminus S}(A) = r(A), \qquad \qquad r_{M/S}(A) = r(A \cup S) - r(S) \qquad (A \subseteq E - S)$$

Let a **minor** of M be a matroid of the form  $M/S \setminus T$  for disjoint  $S, T \subseteq E$ .

3. Deletion and contraction are dual operations; that is, for all  $S \subseteq E$ ,

$$(M\backslash S)^* = M^*/S$$

**Proposition/Definition 7.4.4.** (Direct sum, connected components)

- 1. If  $M = (E_1, \mathcal{I}_1)$  and  $M_2 = (E_2, \mathcal{I}_2)$  are matroids on disjoint ground sets, then  $\mathcal{I} = \{I_1 \cup I_2 : I_1 \in \mathcal{I}_1, I_2 \in \mathcal{I}_2\}$  is the collection of independent sets of a matroid  $M_1 \oplus M_2 = (E_1 \cup E_2, \mathcal{I})$ , called the **direct sum** of  $M_1$  and  $M_2$ .
- 2. Any matroid M decomposes uniquely as a direct sum  $M = M_1 \oplus \cdots \oplus M_c$ . The ground sets of  $M_1, \ldots, M_c$  are called the **connected components** of M. If c = 1, M is **connected**.

The matroid polytope of M has  $\dim(P_M) = n - c$  where n is the number of elements of M and c is the number of connected components.

### 7.5 A few structural results.

Although they will not be strictly necessary in our discussion of enumerative aspects of matroids, it is worthwhile to mention a few important structural results. In particular, in this section we explain the meaning of duality and minors for various families of matroids.

#### Duality.

• The dual of a linear matroid is linear. When M is the matroid of a subspace V of a vector space W,  $M^*$  is the matroid of its orthogonal complement  $V^{\perp}$ . For that reason,  $M^*$  is sometimes called the orthogonal matroid of M.

• The dual of a graphical matroid is not necessarily graphical; however, matroid duality is a generalization of graph duality. We say a graph G is **planar** if it can be drawn on the plane so that its edges intersect only at the endpoints. Such a drawing is called a **plane** graph. The dual graph  $G^*$  is obtained by putting a vertex  $v^*$  inside each region of G (including the "outside region"), and joining  $v^*$  and  $w^*$  by an edge labelled  $e^*$  if the corresponding regions of G are separated by an edge e in G. This construction is exemplified in Figure 7.2. The graph  $G^*$  depends on the drawing of G, but its matroid does not: if G is planar, we have  $M(G^*) = M(G)^*$ .



Figure 7.2: A plane graph G and its dual  $G^*$ .

• Embarrassingly, no one knows whether the dual of an algebraic matroid is always algebraic. This illustrates how poorly we currently understand this family of matroids.

- The cotransversal matroids are precisely the duals of the transversal matroids.
- The Catalan matroid is self-dual. The dual of a lattice path matroid is a lattice path matroid.

• Linear programming duality can be framed and generalized in the context of *oriented matroids*; see [BLVS<sup>+</sup>99] for details.

**Corollary 7.5.1.** If a connected planar graph has v vertices, e edges, and f faces, then v-e+f=2.

*Proof.* A basis of our graph G has v-1 elements and a basis of its dual graph  $G^*$  has f-1 elements; so if M is the matroid of G, we have  $(v-1) + (f-1) = r(M) + r(M^*) = e$ .

### Minors.

• The minors of a linear matroid are linear. If M is the matroid of a vector configuration  $E \subset V$ and  $\mathbf{v} \in E$ , then  $M \setminus \mathbf{v}$  is the matroid of  $E \setminus \mathbf{v}$ , and  $M / \mathbf{v}$  is the matroid of  $E \setminus \mathbf{v}$  modulo  $\mathbf{v}$ ; that is, the matroid of  $\pi(E \setminus \mathbf{v})$ , where  $\pi : V \to V/(\operatorname{span} \mathbf{v})$  is the canonical projection map.

• The minors of a graphical matroid are graphical. If M is the matroid of a graph G and e = uv is an edge, then  $M \setminus e$  is the matroid of the graph  $G \setminus e$  obtained by removing the edge e, and M/e is the matroid of the graph G/e obtained by removing the edge e and identifying vertices u and v.

• [Oxl11, Cor. 6.7.14] The minors of an algebraic matroid are algebraic.

• Transversal matroids are closed under contraction, but not deletion. Cotransversal matroids are closed under deletion, but not contraction. A **gammoid** is a deletion of a transversal matroid or, equivalently, a contraction of a cotransversal matroids. Gammoids are the smallest minor-closed class of matroids containing transversal (or cotransversal) matroids.

• [BdMN03] Catalan matroids are not closed under deletion or contraction. Lattice path matroids are closed under deletion and contraction.

#### Direct sums and connectivity.

• If  $M_1$  and  $M_2$  are the linear matroids of configurations  $E_1 \subset V_1$  and  $E_2 \subset V_2$  over the same field k, then  $M_1 \oplus M_2$  is the matroid of  $\{(\mathbf{v}_1, \mathbf{0}) : \mathbf{v}_1 \in E_1\} \cup \{(\mathbf{0}, \mathbf{v}_2) : \mathbf{v}_2 \in E_2\}$  in  $V_1 \times V_2$ .

• If  $M_1$  and  $M_2$  are the matroids of graphs  $G_1$  and  $G_2$ , then  $M_1 \oplus M_2$  is the matroid of the disjoint union of  $G_1$  and  $G_2$ . The matroid of a loopless graph G is connected if and only if G is 2-connected; that is, G is connected, and there does not exist a vertex of G whose removal disconnects it.

**Non-representability.** Although it is conjectured that almost all matroids are not linear and not algebraic, it takes some effort to construct examples of non-linear and non-algebraic matroids. We describe the simplest examples.

• The Fano matroid  $F_7$  is the linear matroid determined by the seven non-zero vectors in  $\mathbb{F}_2^3$ . This matroid is illustrated in the left panel of Figure 7.3, where an element is denoted by a vertex, and a circuit is denoted by a line or circle joining the points in question. The Fano matroid is representable over  $\Bbbk$  if and only if char( $\Bbbk$ ) = 2.

• The non-Fano matroid  $F_7^-$  is the linear matroid determined by the seven non-zero 0-1 vectors (0,0,1), (0,1,0), (1,0,0), (0,1,1), (1,0,1), (1,1,0), (1,1,1) in  $\mathbb{R}^3$ . It is representable over  $\mathbb{k}$  if and only if char( $\mathbb{k}$ )  $\neq 2$ .

• The direct sum  $F_7 \oplus F_7^-$  is not representable over any field.

• The Vámos matroid  $V_8$  is a matroid of rank 4 whose only non-trivial circuits are 1234, 1456, 1478, 2356, 2378, as illustrated in the right panel of Figure 7.3. This matroid is not representable



Figure 7.3: The Fano matroid, the non-Fano matroid  $F_7^-$ , and the Vamos matroid  $V_8$ .

over any field, because in any affine configuration of points with these coplanarities, 5678 would also be coplanar. The Vamos matroid is also not algebraic over any field.

**Relating various classes.** Most matroids that we encounter naturally in combinatorics are linear, although this is usually not clear from their definition.

- All the examples of Section 7.3 are linear, with the possible exception of algebraic matroids.
- Graphical, transversal, cotransversal matroids, and positroids are linear.
- Linear matroids over k are algebraic over k.
- Algebraic matroids over k of characteristic 0 are linear over a field extension of k.
- Catalan matroids are Schubert matroids.
- Schubert matroids are lattice path matroids.
- Lattice path matroids are transversal.
- Positroids are gammoids.

**Obstructions to representability.** A general question in structural matroid theory is to try to describe a family  $\mathcal{F}$  of matroids by understanding the "obstructions" to belong to  $\mathcal{F}$ . These results are inspired by Kuratowski's Theorem in graph theory, which asserts that a graph is planar if and only if it does not contain the complete graph  $K_5$  or the complete bipartite graph  $K_{3,3}$  as a minor. Similarly, one seeks a (hopefully finite) list of minors that a matroid must avoid to be in  $\mathcal{F}$ .

The question that has received the most attention has been that of characterizing the matroids that are representable over a particular field. Here are some results:

- M is representable over  $\mathbb{F}_2$  if and only if it has no minor isomorphic to  $U_{2,4}$ .
- *M* is representable over  $\mathbb{F}_3$  if and only if it has no minor isomorphic to  $U_{2,5}, U_{3,5}, F_7$ , or  $F_7^*$ .
- M is representable over every field if and only if it has no minor isomorphic to  $U_{2,4}, F_7$ , or  $F_7^*$ .
- M is graphical if and only if it has no minor isomorphic to  $U_{2,4}, F_7, F_7^*, M(K_5)^*, \text{ or } M(K_{3,3})^*$ .

• For any field of characteristic zero, there are infinitely many minor-minimal matroids that are not representable over  $\mathbb{F}$ .

• (Rota's Conjecture) For any finite field  $\mathbb{F}_q$ , there is a finite set  $\mathcal{S}_q$  of matroids such that M is representable over  $\mathbb{F}_q$  if and only if it contains no minor isomorphic to a matroid in  $\mathcal{S}_q$ .

In 2013, Geelen, Gerards, and Whittle announced a proof of Rota's conjecture. In [GGW14] they write: "We are now immersed in the lengthy task of writing up our results. Since that process will take a few years, we have written this article offering a high-level preview of the proof".

**Asymptotic enumeration.** The asymptotic growth of various classes of matroids is not well understood. Let P be a matroid property, and let p(n) be the number of matroids on [n] having property P. Let l(n) be the number of matroids on [n]. The asymptotic proportion of matroids having property P is  $\lim_{n\to\infty} p(n)/l(n)$ , if it exists. We list some conjectures. [MNWW11]

- Asymptotically almost every matroid is connected.
- Asymptotically almost every matroid M is **paving**; that is, all its circuits have size r or r+1.
- Asymptotically every matroid is asymmetric, that is, it has no non-trivial automorphism.
- Asymptotically every matroid is not linear.
- Asymptotically almost every matroid of n elements on rank r satisfies  $(n-1)/2 \le r \le (n+1)/2$ .

We also list some results:

• [BPvdP12, Knu74] The number  $m_n$  of matroids on [n] satisfies

$$n - \frac{3}{2}\log n + \frac{1}{2}\log \frac{2}{\pi} - o(1) \le \log\log m_n \le n - \frac{3}{2}\log n + \frac{1}{2}\log \frac{2}{\pi} + 1 + o(1)$$

• [MNWW11] Asymptotically almost every matroid is loopless and coloopless.

• [MNWW11] Asymptotically, the proportion of matroids that are connected is at least 1/2. (It is conjectured to equal 1.)

• [ARW13b] Asymptotically, the proportion of **positroids** that are connected is  $1/e^2$ .

# 7.6 The Tutte polynomial

Throughout this section, let R be an arbitrary commutative ring. A function f: Matroids  $\rightarrow R$  is a **matroid invariant** if f(M) = f(N) whenever  $M \cong N$ . Let L and C be the matroids consisting of a single loop and a single coloop, respectively.

#### 7.6.1 Explicit definition.

The invariant that appears most often in enumerative questions is the **Tutte polynomial** 

$$T_M(x,y) = \sum_{A \subseteq M} (x-1)^{r-r(A)} (y-1)^{|A|-r(A)}.$$
(25)

When we meet a new matroid invariant, the first question we usually ask is whether it is an evaluation of T(x, y).

#### 7.6.2 Recursive definition and universality property.

The ubiquity of the Tutte polynomial is not an accident: it is *universal* in a large, important family of matroid invariants. Let us make this precise. Let R be a ring. Say f: Matroids  $\rightarrow R$  is a **generalized Tutte-Grothendieck invariant** if for every matroid M and every element  $e \in M$ ,

$$f(M) = \begin{cases} af(M \setminus e) + bf(M/e) & \text{if } e \text{ is neither a loop nor a coloop} \\ f(M \setminus e)f(L) & \text{if } e \text{ is a loop} \\ f(M/e)f(C) & \text{if } e \text{ is a coloop} \end{cases}$$
(26)

for some non-zero constants  $a, b \in R$ . It is a **Tutte-Grothendieck invariant** if a = b = 1.

**Theorem 7.6.1.** The Tutte polynomial is a universal Tutte-Grothendieck invariant; namely:

1. For every matroid M and every element  $e \in M$ 

$$T_M(x,y) = \begin{cases} T_{M \setminus e}(x,y) + T_{M/e}(x,y) & \text{if } e \text{ is neither a loop nor a coloop} \\ y T_{M \setminus e}(x,y) & \text{if } e \text{ is a loop,} \\ x T_{M/e}(x,y) & \text{if } e \text{ is a coloop.} \end{cases}$$

2. Any generalized Tutte-Grothendieck invariant is a function of the Tutte polynomial. Explicitly, if f satisfies (26), then

$$f(M) = a^{n-r} b^r T_M\left(\frac{f(C)}{b}, \frac{f(L)}{a}\right).$$

where n is the number of elements and r is the rank of M.<sup>14</sup>

Sketch of Proof. This is a very powerful result with a very simple proof. The first part is a straightforward computation from the definitions. The second statement then follows easily by induction on the number of elements of M.

More generally, write  $T(x, y) = \sum_{i,j} t_{ij} x^i y^j$ . If f satisfies  $f(M) = f(M \setminus e) + f(M/e)$  when e is neither a loop nor a coloop, then  $f(M) = \sum_{i,j} t_{ij} f(C^i \oplus L^j)$ , where  $C^i \oplus L^j$  denotes the matroid consisting of i coloops and j loops.

**Proposition 7.6.2.** The Tutte polynomial behaves well with respect to duality and direct sums:

- 1. For any matroid M,  $T_{M^*}(x, y) = T_M(y, x)$ .
- 2. For any matroids M and N on disjoint ground sets,  $T_{M\oplus N}(x,y) = T_M(x,y)T_N(x,y)$ .

### 7.6.3 Activity interpretation.

An unexpected consequence of Theorem 7.6.1 is that the Tutte polynomial has non-negative coefficients; this is not at all apparent from the explicit formula (25). As usual, the natural question for a combinatorialist is: what do these coefficients count? The natural question for an algebraist is: what vector spaces have these coefficients as their dimensions? At the moment, the first question has a nice answer, while the second one does not.

Fix a linear order < on the elements of E. Say an element  $i \in B$  is **internally active** if there is no basis  $B - i \cup j$  with j < i. Say an element  $j \notin B$  is **externally active** if there is no basis  $B - i \cup j$  with i < j. These are dual notions: i is internally active with respect to basis B in M if and only if it is externally active with respect to basis E - B in  $M^*$ .

**Theorem 7.6.3.** [Cra69, Tut54] For any linear order < on the ground set of a matroid, let I(B) and E(B) be, respectively, the set of internally active and externally active elements with respect to B. Then

$$T_M(x,y) = \sum_{B \text{ basis}} x^{|I(B)|} y^{|E(B)|}.$$

<sup>&</sup>lt;sup>14</sup>We do not need to assume a and b are invertible; when we multiply by  $a^{n-r}b^r$ , we cancel all denominators.

Sketch of Proof. It is possible to give a standard deletion-contraction proof, by showing that the right hand side is a Tutte-Grothendieck invariant, and applying Theorem 7.6.1. However, there is a much more enlightening explanation. One may prove [Cra69] that the intervals  $[B \setminus I(B), B \cup E(B)]$  partition the Boolean lattice  $2^E$ . In other words, every subset of the ground set can be written uniquely in the form  $(B \setminus J) \cup F$  where B is a basis,  $J \subseteq I(B)$ , and  $F \subseteq E(B)$ . Furthermore, under these assumptions  $r((B \setminus J) \cup F) = r - |J|$ . Therefore

$$T_M(x,y) = \sum_{B \text{ basis }} \sum_{J \subseteq I(B)} \sum_{F \subseteq E(B)} (x-1)^{|J|} (y-1)^{|F|} = \sum_{B \text{ basis }} x^{|I(B)|} y^{|E(B)|}$$

as desired.

This combinatorial interpretation of the non-negative coefficients of the Tutte polynomial is very interesting combinatorially, but it would be even nicer to find an interpretation that does not depend on choosing a linear order on the ground set.

In a different direction, Procesi [DCP08a] asked the following question: Given a matroid M, is there a natural bigraded algebra whose bigraded Hilbert polynomial is  $T_M(x, y)$ ? This question is still open. It may be more tractable (and still very interesting) when M is representable; we will see some approximations in Section 7.7.

#### 7.6.4 Finite field interpretation.

The **coboundary polynomial**  $\overline{\chi}_M(X,Y)$  is the following simple transformation of the Tutte polynomial:

$$\overline{\chi}_M(X,Y) = (Y-1)^r T_M\left(\frac{X+Y-1}{Y-1},Y\right).$$

It is clear how to recover T(x, y) from  $\overline{\chi}(X, Y)$ . We have the following interpretation of the coboundary (and hence the Tutte) polynomial.

**Theorem 7.6.4.** (Finite Field Method) [Ard07, CR70, WW99] Let  $\mathcal{A}$  be a hyperplane arrangement of rank r in  $\mathbb{F}_q^d$ . For each point  $p \in \mathbb{F}_q^d$  let h(p) be the number of hyperplanes of  $\mathcal{A}$  containing p. Then

$$\sum_{p \in \mathbb{F}_q^d} t^{h(p)} = q^{n-r} \overline{\chi}(q, t).$$

*Proof.* For each  $p \in \mathbb{F}_q^n$ , let H(p) be the set of hyperplanes of  $\mathcal{A}$  that p lies on. Then

$$q^{n-r} \overline{\chi}_{\mathcal{A}}(q,t) = \sum_{\mathcal{B} \subseteq \mathcal{A}} q^{n-r(\mathcal{B})} (t-1)^{|\mathcal{B}|} = \sum_{\mathcal{B} \subseteq \mathcal{A}} q^{\dim \cap \mathcal{B}} (t-1)^{|\mathcal{B}|} = \sum_{\mathcal{B} \subseteq \mathcal{A}} |\cap \mathcal{B}| (t-1)^{|\mathcal{B}|}$$
$$= \sum_{\mathcal{B} \subseteq \mathcal{A}} \sum_{p \in \cap \mathcal{B}} (t-1)^{|\mathcal{B}|} = \sum_{p \in \mathbb{F}_q^n} \sum_{\mathcal{B} \subseteq H(p)} (t-1)^{|\mathcal{B}|} = \sum_{p \in \mathbb{F}_q^n} (1+(t-1))^{h(p)},$$

as desired.

Computing Tutte polynomials is extremely difficult (#P-complete [Wel93]) for general matroids, but it is still possible in some cases of interest. Theorem 7.6.4 is one of the most effective methods for computing Tutte polynomials of (a few) particular arrangements  $\mathcal{A}$  in  $\mathbb{k}^d$ , as follows.

If the hyperplanes of  $\mathcal{A}$  have integer coefficients (as most arrangements of interest do), we may use the same equations to define an arrangement  $\mathcal{A}_q$  over  $\mathbb{F}_q^d$ . If q is a power of a large enough prime, then  $\mathcal{A}$  and  $\mathcal{A}_q$  have isomorphic matroids, and hence have the same Tutte polynomial. Then Theorem 7.6.4 reduces the computation of  $T_{\mathcal{A}}(x, y)$  to an enumerative problem over  $\mathbb{F}_{q}^{d}$ , which can sometimes be solved. [Ard07]

#### 7.7Tutte polynomial evaluations

Many important invariants of a matroid are generalized Tutte-Grothendieck invariants, and hence are evaluations of the Tutte polynomial. In fact, many results outside of matroid theory fit naturally into this framework. In this section we collect, without proofs, results of this sort in many different areas of mathematics and applications.

One can probably prove every statement in this section by proving that the quantities in question satisfy a deletion-contraction recursion; many of the results also have more interesting and enlightening explanations. The wonderful surveys [BO92], [EMM11], and [Wel93] include most of the results mentioned here; we provide references for the other ones.

**General evaluations.** Let M be any matroid of rank r.

- The number of independent sets is T(2, 1).
- The number of spanning sets is T(1, 2).
- The number of bases is T(1,1).
- The number of elements of the ground set is  $\log_2 T(2,2)$ .
- The Möbius number is μ(M) = μ<sub>L<sub>M</sub></sub>(0, 1) = (-1)<sup>r</sup>T(1, 0).
  We have the formula χ<sub>M</sub>(q) = (-1)<sup>r(M)</sup>T(1 q, 0) for the characteristic polynomial:<sup>15</sup>

$$\chi_M(q) = \sum_{F \in L_M} \mu(\widehat{0}, F) q^{r-r(F)}$$

• The **beta invariant** of M is defined to be  $\beta(M) = [x^1y^0]T(x,y) = [x^0y^1]T(x,y)$ . It has some useful properties. We always have  $\beta(M) \ge 0$ . We have  $\beta(M) = 0$  if and only if M is disconnected. We have  $\beta(M) = 1$  if and only if M is the graphical matroid of a series-parallel graph; that is, a graph obtained from a single edge by repeatedly applying series extensions (convert an edge uvinto two edges uw and wv for a new vertex w) and parallel extensions (convert one edge uv into two edges joining u and v). There are similar characterizations of the matroids with  $\beta(M) \leq 4$ .

The independence complex of M is the simplicial complex consisting of the independent sets of M. Recall that the f-vector  $(f_0, \ldots, f_r)$  and h-vector  $(h_0, \ldots, h_r)$  are defined so that  $f_i$  is the number of sets of size i and  $\sum h_i x^{r-i} = \sum f_i (x-1)^{r-i}$ . Then:

- The reduced Euler characteristic of the independence complex is T(0, 1).
- The *f*-polynomial of the independence complex is ∑<sub>i</sub> f<sub>i</sub>x<sup>i</sup> = x<sup>r</sup>T(1 + 1/x, 1).
  The *h*-polynomial of the independence complex is ∑<sub>i</sub> h<sub>i</sub>x<sup>i</sup> = x<sup>r</sup>T(1/x, 1).

This *h*-polynomial is also known as the **shelling polynomial** of M.

There are (more than) two other interesting geometric/topological objects associated to a matroid. Björner [Bjö82] described the order complex of the (proper part of the) lattice of flats  $\Delta(L_M - \{\widehat{0}, \widehat{1}\})$ , showing it is a wedge of T(1, 0) spheres. The Bergman fan

<sup>&</sup>lt;sup>15</sup>Note that if  $\mathcal{A}$  is a hyperplane arrangement of rank r in  $\mathbb{k}^n$  and M is its matroid, then the characteristic polynomials defined in Section 6.2 is given by  $\chi_{\mathcal{A}}(q) = q^{d-r} \chi_M(q)$ .

Trop $(M) = \{w \in \mathbb{R}^E : \text{ for every circuit } C, \max_{c \in C} w_c \text{ is achieved at least twice}\}\$  is the tropical geometric analog of a linear space. The **Bergman complex** is its intersection with the unit sphere. Ardila and Klivans showed  $\mathcal{B}(M)$  and  $\Delta(L_M - \{\hat{0}, \hat{1}\})$  are homeomorphic. [AK06] Then:

- [Bjö82] The reduced Euler characteristic of the order complex  $\Delta(L_M \{\hat{0}, \hat{1}\})$  is T(1, 0).
- [AK06] The reduced Euler characteristic of the Bergman complex  $\mathcal{B}(M)$  is T(1,0).
- [Zha13] The Poincaré polynomial of the tropical cohomology of  $\operatorname{Trop}(M)$  is  $q^{r}T(1+1/q,0)$ .

One of the most intriguing conjectures in matroid theory is **Stanley's** h-vector conjecture [Sta77] which states that for any matroid M, there exists a set X of monomials such that:

- if m and m' are monomials such that  $m \in X$  and m'|m, then  $m' \in X$ ,
- all the maximal monomials in X have the same degree,
- there are exactly  $h_i$  monomials of degree i in X.

This conjecture has been proved, using rather different methods, for several families: duals of graphic matroids, [Mer01], lattice path matroids [Sch10] cotransversal matroids [Oh13], paving matroids [MNRIVF12], and matroids up to rank 4 or corank 2 [DLKK12, KS14]. The general case remains open.

**Graphs.** Let G = (V, E) be a graph with v vertices, e edges, and c connected components. Let M(G) be the matroid of G and T(x, y) be its Tutte polynomial.

(Colorings.) Recall that the chromatic polynomial  $\chi_G(q)$  of a graph, when evaluated at a positive integer q, counts the proper colorings of G with q colors.

• We have

$$\chi_G(q) = (-1)^{v-c} q^c T(1-q,0).$$

• More generally, for every q-coloring  $\chi$  of the vertices of G, let  $h(\chi)$  be the number of edges whose endpoints have the same color. Then

$$\sum_{\chi:V \to [q]} t^{h(\chi)} = (t-1)^{v-c} q^c T\left(\frac{q+t-1}{t-1}, t\right)$$
(27)

(Flows.) Fix an orientation of the edges of G and a finite Abelian group H of t elements, such as  $\mathbb{Z}/t\mathbb{Z}$ , for some  $t \in \mathbb{N}$ . An H-flow is an assignment  $f : E \to H$  of a "flow" (an element of H) to each edge such that at every vertex, the total inflow equals the outflow as elements of H. We say f is nowhere zero if  $f(e) \neq 0$  for all edges e.

• The number of nowhere zero *t*-flows is given by the **flow polynomial** 

$$\chi_G^*(t) = (-1)^{e-v+c} T(0, 1-t).$$

In particular, this number is independent of the orientation of G. It also does not depend on the particular group H, but only on its size.

• More generally, for every *H*-flow f on the edges of G, let h(v) be the number of edges having flow equal to 0. Then

$$\sum_{f:E \to H} t^{h(f)} = (t-1)^{e-v+c} T\left(t, \frac{q+t-1}{t-1}\right).$$

(Acyclic orientations.) An acyclic orientation of G is an orientation of the edges that produces no directed cycles.

• The number of acyclic orientations is T(2,0).

• A source (resp. a sink) of an orientation is a vertex with no incoming (resp. outgoing) edges. For any fixed vertex w, the number of acyclic orientations whose unique source is w is  $(-1)^{v-c}\mu(M(G)) = T(1,0)$ . In particular, it does not depend on the choice of w.

• For any edge e = uv, the number of acyclic orientations whose unique source is u and whose unique sink is v is the beta invariant  $\beta(M(G))$ . In particular, it is independent of the choice of e.

(Totally cyclic orientations.) A totally cyclic orientation of G is an orientation of the edges such that every edge is contained in a cycle. If G is connected, this is the same as requiring that there are directed paths in both directions between any two vertices.

• The number of totally cyclic orientations is T(0, 2).

• Given an orientation o of G, the outdegree of a vertex is the number of outgoing edges, and the outdegree sequence of o is  $(outdeg(v) : v \in V)$ . The number of distinct outdegree sequences among the orientations of G is T(2, 1)

• For any edge e, the number of totally cyclic orientations of G such that every directed cycle contains e is equal to  $2\beta(M(G))$ . In particular, it is independent of e.

(Eulerian orientations.) Recall that an orientation of G is Eulerian if each vertex has the same number of incoming and outgoing edges. Say G is 4-regular if every vertex has degree 4.

• A 4-regular graph G has  $(-1)^{v+c}T_G(0,-2)$  Eulerian orientations or ice configurations. (They are in easy bijection with the nowhere-zero  $\mathbb{Z}_3$ -flows.)

(Chip firing.) Let G = (V, E) be a graph and let  $q \in V$  be a vertex called the **bank**. A **chip** configuration is a map  $\theta: V \to \mathbb{Z}$  with  $\theta(v) \ge 0$  for  $v \ne q$  and  $\theta(q) = -\sum_{v \ne q} \theta(v)$ . We think of vertex v as having  $\theta(v)$  chips. We say  $v \ne q$  is ready if  $\theta(v) \ge \deg(v)$ , and q is ready if no other vertex is ready. In each step of the **chip firing game**, a ready vertex v gives away  $\deg(v)$  chips, giving e chips to w if there are e edges connecting v and w.

We say a configuration  $\theta$  is **stable** if the only vertex that is ready is q. We say  $\theta$  is **recurrent** if it is possible to fire a (non-empty) sequence of vertices subsequently, and return to  $\theta$ . We say  $\theta$  is **critical** if it is stable and recurrent. One can check that for any starting chip configuration, the chip firing game leads us to a unique critical configuration. The following result is the key to Merino's proof of Stanley's *h*-vector conjecture in the special case of cographic matroids:

• [Mer01] The generating function for critical configurations is

$$\sum_{\substack{\theta \text{ critical}}} y^{-\theta(q)} = y^{|E| - \deg(q)} T_G(1, y).$$

Chip-firing games have recently gained prominence as that they are intimately connected to the study of divisors on tropical algebraic curves. [BN07, GK08]

(Plane graphs and Eulerian partitions.) Let G be a connected plane graph. The medial graph H has a vertex on each edge of G, and two vertices of H are connected if the corresponding edges of G are neighbors in some face of G. This construction is shown in Figure 7.4.

An Eulerian partition  $\pi$  of H is a partition of the edges into closed paths. A closed path may revisit vertices, but it may not revisit edges; and different starting points and orientations are



Figure 7.4: A connected graph G and its medial graph H in bold.

considered to be the same path. Let  $\gamma(\pi)$  be the number of paths in  $\pi$ . If the edges of a vertex v of H are  $e_1, e_2, e_3, e_4$  in clockwise order, say  $\pi$  has a **crossing** at v if one of its paths uses edges  $e_1$  and  $e_3$  consecutively.

• [LV88] We have  $T_G(-1, -1) = (-1)^e (-2)^{\gamma(\pi_c)-1}$ , where  $\pi_c$  is the unique **fully crossing** Eulerian partition of H with a crossing at every vertex.

• [LV88] If  $\Pi_{nc}(H)$  is the set of Eulerian partitions of H with no crossings, then

$$T_G(x,x) = \sum_{P \in \Pi_{nc}(H)} (x-1)^{\gamma(P)-1}$$

In the example above, note that the fully crossing Eulerian partition of G consists of one Eulerian walk, in agreement with  $T_G(-1, -1) = 1 = (-1)^6 (-2)^{1-1}$ . We have  $T_G(x, x) = 3x^4 + 2x^3$ .

**Hyperplane arrangements.** Let  $\mathcal{A}$  be an arrangement of n hyperplanes in  $\mathbb{k}^d$  of rank r. Let  $V(\mathcal{A})$  be the complement of  $\mathcal{A}$  in  $\mathbb{k}^d$ . We restate some theorems from Section 6.2 and present new ones.

- If  $\mathbb{k} = \mathbb{R}$ , the number of regions of  $V(\mathcal{A})$  is T(2,0).
- If  $\mathbf{k} = \mathbb{C}$ , the Poincaré polynomial of  $V(\mathcal{A})$  is  $q^r T(1 + \frac{1}{q}, 0)$ .
- If  $\mathbf{k} = \mathbf{F}_q$ , the number of elements of  $V(\mathcal{A})$  is  $(-1)^r q^{n-r} T(1-q,0)$

• Suppose  $\mathbb{k} = \mathbb{R}$ , and consider an affine hyperplane H which is in general position with respect to  $\mathcal{A}$ . Then the number of regions of  $\mathcal{A}$  which have a bounded (and non-empty) intersection with H is  $(-1)^r |\mu(M)| = T(1,0)$ .

• Suppose  $\mathbb{k} = \mathbb{R}$ , and add to  $\mathcal{A}$  an affine hyperplane H' which is a translation of  $H \in \mathcal{A}$ . The number of bounded regions of  $\mathcal{A} \cup H'$  is  $\beta(M)$ .

Algebras from vector and hyperplane arrangements. There are several natural algebraic spaces related to the Tutte polynomial arising in commutative algebra, hyperplane arrangements, box splines, and index theory; we discuss a few. For each hyperplane H in a hyperplane arrangement  $\mathcal{A}$  in  $\mathbb{k}^d$  let  $l_H$  be a linear function such that H is given by the equation  $l_H(x) = 0$ .

• [Wag99] Let  $C_{\mathcal{A},0} = \operatorname{span}\{\prod_{H \in \mathcal{B}} l_H : \mathcal{B} \subseteq \mathcal{A}\}$ . This is a subspace of a polynomial ring in d variables, graded by degree. Its dimension is T(2, 1) and its Hilbert series is

$$\operatorname{Hilb}(C_{\mathcal{A},0};q) = \sum_{j\geq 0} \dim(C_{\mathcal{A},0})_j q^j = q^{n-r} T\left(1+q,\frac{1}{q}\right).$$

• [AP10, DM85, HR11, PS04, PSS99] More generally, let  $C_{\mathcal{A},k}$  be the vector space of polynomial functions such that the restriction of f to any line l has degree at most  $\rho_{\mathcal{A}}(h) + k$ , where  $\rho_{\mathcal{A}}(h)$  is

the number of hyperplanes of  $\mathcal{A}$  not containing h. It is not obvious, but true, that this definition of  $C_{\mathcal{A},0}$  matches the one above. Then

$$\operatorname{Hilb}(C_{\mathcal{A},-1};q) = q^{n-r}T\left(1,\frac{1}{q}\right), \qquad \operatorname{Hilb}(C_{\mathcal{A},-2};q) = q^{n-r}T\left(0,\frac{1}{q}\right)$$

and similar formulas hold for any  $k \geq -2$ .

• [BV99, PS06, Ter02] Let  $R(\mathcal{A})$  be the vector space of rational functions whose poles are in  $\mathcal{A}$ . It may be described as the k-algebra generated by the rational functions  $\{1/l_H : H \in \mathcal{A}\}$ ; we grade it so that deg $(1/l_H) = 1$ . Then

$$\operatorname{Hilb}(R(\mathcal{A});q) = \frac{q^d}{(1-q)^d} T\left(\frac{1}{q},0\right).$$

**Error-correcting codes.** Suppose we wish to transmit a message over a noisy channel. We might then encode our message in a redundant way, so that we can correct small errors introduced during transmission. An **error-correcting code** is a set  $C \subset A^n$  of **codewords** of length n over an alphabet A. The sender encodes each word into a redundant codeword, which they transmit. If the channel is not too noisy and the codewords in C are sufficiently different from each other, the recipient will succeed in recovering the original message. Of course it is useful to have many codewords available under these constraints.

A common kind of error-correcting code is a **linear code** C, which is a k-subspace of a vector space  $\mathbb{F}_q^n$ . The codewords have length n and alphabet  $\mathbb{F}_q$ . The **support** of a word is the set of non-zero entries. The **distance**  $d(\mathbf{u}, \mathbf{v})$  between two words  $\mathbf{u}$  and  $\mathbf{v}$  is the number of coordinates where  $\mathbf{u}$  and  $\mathbf{v}$  differ. Since  $\mathbf{u} - \mathbf{v} \in C$ , we have  $d(\mathbf{u}, \mathbf{v}) = |\operatorname{supp}(\mathbf{u} - \mathbf{v})|$ . The minimum **distance** d in the code is  $d = \min_{c \in C} |\operatorname{supp}(c)|$ . The code C is said to be of type [n, k, d].

• The weight enumerator of a linear code over  $\mathbb{F}_q$  is  $W_C(q, z) = \sum_{c \in C} z^{|\text{supp}(c)|}$ . Curtis Greene [Gre76] discovered that it can be expressed in terms of the Tutte polynomial; in fact, Theorem 7.6.4 is equivalent to the equation

$$W_C(q,z) = (1-z)^k z^{n-k} T_C\left(\frac{1+(q-1)z}{1-z}, \frac{1}{z}\right).$$
(28)

The **dual code** of *C* is  $C^{\perp} = \{\mathbf{x} \in \mathbb{F}_q^n : (\mathbf{x}, \mathbf{c}) = 0 \text{ for all } \mathbf{c} \in C\}$ . An important example is the **Hamming code** *H*. Let  $\mathcal{A}(2, n)$  be the rowspace of the  $n \times (2^n - 1)$  matrix whose columns are all the nonzero vectors in  $\mathbb{F}_2^n$ , and let  $H = \mathcal{A}(2, n)^{\perp}$ . This is a largest possible code (consisting of  $2^{2^n - n - 1}$  words) of length  $2^n - 1$  and distance 3.

• Florence MacWilliams [Mac63] gave the following remarkable identity relating the weight enumerators of C and  $C^{\perp}$ :

$$W_{C^{\perp}}(q,z) = \frac{(1+(q-1)z)^n}{q^k} W_C\left(q, \frac{1-z}{1+(q-1)z}\right)$$

In light of (28), this is equivalent to Tutte polynomial duality:  $T_{C^{\perp}}(x, y) = T_C(y, x)$ . It is easy to show that the weight enumerator of  $\mathcal{A}(2, n)$  is  $1 + (q^n - 1)z^{q^{n-1}}$ . [Gre76, Example 3.4] MacWilliams's Identity then gives us the weight enumerator of the Hamming code.

• Another nice result [Jae89] is that if C is a linear code in  $\mathbb{F}_2^n$  then  $T_C(-1, -1) = (-1)^n |C \cap C^{\perp}|$ .

**Probability and statistical mechanics.** The Tutte polynomial arises in several standard processes in probability and statistical physics. We consider three related models. [BO92, Sok05, Wel93]

• Suppose each edge of a connected graph G is white with probability p and black with probability 1-p (for fixed  $0 \le p \le 1$ ), independently of the other edges. The probability that the white graph is connected is given by the **reliability polynomial** 

$$R_G(p) = \sum_{\substack{A \subseteq E \\ A \text{ spanning}}} p^{|A|} (1-p)^{|E-A|} = (1-p)^{e-v+1} p^{v-1} T_G\left(1, \frac{1}{1-p}\right).$$

• The random cluster model depends on parameters  $0 \le p \le 1$  and q > 0. Now we choose a white set of edges at random, and the probability of choosing  $A \subset E$  is  $p^{|A|}(1-p)^{|E-A|}q^{c(A)}/Z$ , where c(A) = |V| - r(A) is the number of components of A, and Z is a scaling constant. To know the probability of a particular state, it is fundamental to know the scaling constant Z, which is the partition function

$$Z(p,q) = \sum_{A \subseteq E} p^{|A|} (1-p)^{|E-A|} q^{c(A)} = p^{v-c} (1-p)^{e-v+c} q^c T_G \left( 1 + \frac{q(1-p)}{p}, \frac{1}{1-p} \right)$$

• In the q-state Potts model, a graph G = (V, E) models a set V of "atoms" and a set E of "bonds" between them. (When q = 2, this is the Ising model.) Each atom can exist in one of q states or "spins". Each edge e = uv has an associated *interaction energy*  $J_e$  between u and v. The energy (or Hamiltonian) H of a configuration is the sum of  $-J_e$  over all edges e whose vertices have the same spin.<sup>16</sup> A configuration of energy H has Boltzmann weight  $e^{-\beta H}$  where  $\beta = 1/kT > 0$ , T is the temperature, and k is Boltzmann's constant.

The **partition function** is the sum of the Boltzmann weights of all configurations:

$$Z_G(q, \mathbf{w}) = \sum_{\sigma: V \to [q]} \prod_{\substack{e=ij \in E \\ \sigma(i) = \sigma(j)}} e^{\beta J_e} = \sum_{\sigma: V \to [q]} \prod_{\substack{e=ij \in E \\ \sigma(i) = \sigma(j)}} (1 + w_e)$$

where  $w_e = e^{\beta J_e} - 1$ . If all  $J_e$ s are equal to J and  $w = e^{\beta J} - 1$ , then (27) gives

$$Z_G(q,w) = w^{v-c}q^c T\left(\frac{q}{w}+1,w+1\right)$$

For general  $J_e$ , a simple computation shows that

$$q^{-v}Z_G(q, \mathbf{w}) = \sum_{A \subseteq E} q^{-r(A)} \prod_{e \in A} w_e = \widetilde{Z}_G(q, \mathbf{w})$$

where  $\widetilde{Z}_G(q, \mathbf{w})$  is the **multivariate Tutte polynomial** of Section 7.9.

#### Other applications.

• A **knot** is an embedding of a circle in  $\mathbb{R}^3$ . It is a difficult, important question to determine whether a knot can be deformed smoothly (without cutting or crossing segments) to obtain another

<sup>&</sup>lt;sup>16</sup>The case  $J_e \ge 0$  is called ferromagnetic, as it favors adjacent spins being equal. The case  $J_e \le 0$  is antiferromagnetic. Here we are assuming that there is no external magnetic field. If there were such a field, it would contribute an additional term to the Hamiltonian, and the direct connection with the Tutte polynomial is no longer valid.

knot; or even to determine whether a given knot is actually knotted or not. Let  $\mathbf{O}$  be the unknotted circle in  $\mathbb{R}^3$ .

One common approach is to construct a function f(K) of a knot K which does not change under smooth deformation. If  $f(K) \neq f(\mathbf{O})$ , then K is actually knotted. One such function is the **Jones polynomial** V(K).[Jon85] If K is an **alternating knot**, there is a graph G associated to it such that the Jones polynomial of K is an evaluation of the Tutte polynomial of G. For details, see [Wel93].

• [Rei99] There is a more symmetric finite field interpretation for the Tutte polynomial. Let M be an integer matrix, and let p and q be prime powers such that the matroid of M does not change when M is considered as a matrix over  $\mathbb{F}_p$  or over  $\mathbb{F}_q$ . Then

$$T_M(1-p, 1-q) = (-1)^{r(M)} \sum_{\substack{\mathbf{x} \in \operatorname{row}(M), \mathbf{y} \in \ker(M) \\ \operatorname{supp}(\mathbf{x}) \sqcup \operatorname{supp}(\mathbf{y}) = E}} (-1)^{|\operatorname{supp}(y)|},$$

where the rowspace row(M) is considered as a subspace of  $\mathbb{F}_p^n$  and the kernel ker(M) is considered as a subspace of  $\mathbb{F}_q^n$ , and  $\sqcup$  denotes a disjoint union.

• [Ard03, BdMN03] The Tutte polynomial of the Catalan matroid is

$$T_{\mathbf{C}_n}(x,y) = \sum_{P \text{ Dyck}} x^{a(P)} y^{b(P)},$$

where a(P) is the number of upsteps before the first downstep of P, and b(P) is the number of times that P returns to the x-axis. Since  $\mathbf{C}_n$  is self-dual, this polynomial is symmetric in x and y.

In fact, the coefficient of  $x^i y^j$  in  $T_{\mathbf{C}_n}(x, y)$  depends only on i + j. It would be interesting to find other families of matroids with this unusual property.

• [BdMN03] More generally, the Tutte polynomial of a lattice path matroid is the sum over the bases P of  $x^{a(P)}y^{b(P)}$ , where a(P) and b(P) are the number of returns to the upper and lower boundary paths, respectively.

• [KP04] A T-tetromino is a T-shape made of four unit squares. Figure 7.5 shows a T-tetromino tiling of an  $8 \times 8$  square. The number of **T-tetromino tilings** of a  $4m \times 4n$  rectangle equals  $2T_{L_{m,n}}(3,3)$ , where  $L_{m,n}$  is the  $m \times n$  grid graph. This result extends to T-tetromino tilings of many other shapes and, more generally, to coverings of many graphs with copies of the "claw" graph  $K_{1,3}$ .



Figure 7.5: A tiling of an  $8 \times 8$  rectangle by T-tetrominoes.

## 7.8 Computing the Tutte polynomial

As we already mentioned, computing the Tutte polynomial of an arbitrary graph or matroid is not a tractable problem. In the language of complexity theory, this is a #P-complete problem. [Wel93] However, the results of Section 7.6 allow us to compute the Tutte polynomial of **some** matroids of interest. We now survey some of the most interesting examples; see [MRIS12] for others. Some of these formulas are best expressed in terms of the **coboundary polynomial** 

$$\overline{\chi}_{\mathcal{A}}(X,Y) = (y-1)^{r(\mathcal{A})} T_{\mathcal{A}}(x,y), \quad \text{where } x = \frac{X+Y-1}{Y-1}, \quad y = Y.$$

Almost all of them are most easily proved using the Finite Field Method (Theorem 7.6.4) or its graph version (27).

- For the uniform matroid  $U_{k,n}$  we have  $T_{U_{k,n}}(x,y) = \sum_{i=1}^{r} \binom{n-i-1}{n-r-1} x^i + \sum_{j=1}^{n-r} \binom{n-j-1}{r-1} y^j$ .
- If  $M^{(k)}$  is the matroid obtained from M by replacing each element by k copies of itself, then

$$T_{M^{(k)}}(x,y) = (y^{k-1} + y^{k-2} + \dots + y + 1)^r T_M\left(\frac{y^{k-1} + y^{k-2} + \dots + y + x}{y^{k-1} + y^{k-2} + \dots + y + 1}, y^k\right)$$

This formula is straightforward in terms of coboundary polynomials:  $\overline{\chi}_{M^{(k)}}(X,Y) = \overline{\chi}_M(X,Y^K)$ . For an extensive generalization, see Section 7.9.

• [Ard07, Mph00] Root systems are arguably the most important vector configurations; these highly symmetric arrangements play a fundamental role in many branches of mathematics. For the general definition and properties, see for example [Hum90]; we focus on the four infinite families of classical root systems:

$$\begin{array}{rcl} A_{n-1} &=& \{e_i - e_j, : 1 \leq i < j \leq n\} \\ B_n &=& \{e_i - e_j, e_i + e_j \, : \, 1 \leq i < j \leq n\} \cup \{e_i \, : \, 1 \leq i \leq n\} \\ C_n &=& \{e_i - e_j, e_i + e_j \, : \, 1 \leq i < j \leq n\} \cup \{2e_i \, : \, 1 \leq i \leq n\} \\ D_n &=& \{e_i - e_j, e_i + e_j \, : \, 1 \leq i < j \leq n\} \end{array}$$

and a small number of exceptional root systems. Figure 7.6.



Figure 7.6: The root systems  $A_2, B_2, C_2$ , and  $D_2$ , respectively.

Let the **deformed exponential function** be  $F(\alpha, \beta) = \sum_{n \ge 0} \frac{\alpha^n \beta^{\binom{n}{2}}}{n!}$ . Then the **Tutte generating functions** of the infinite families A and  $\Phi = B, C, D$ :

$$T_A(X,Y,Z) = 1 + X \sum_{n \ge 1} \overline{\chi}_{A_{n-1}}(X,Y) \frac{Z^n}{n!}, \quad T_\Phi(X,Y,Z) = \sum_{n \ge 0} \overline{\chi}_{\Phi_n}(X,Y) \frac{Z^n}{n!}$$

are given by:

$$T_A = F(Z, Y)^X,$$
  

$$T_B = F(2Z, Y)^{(X-1)/2} F(YZ, Y^2),$$
  

$$T_C = F(2Z, Y)^{(X-1)/2} F(YZ, Y^2),$$
  

$$T_D = F(2Z, Y)^{(X-1)/2} F(Z, Y^2).$$

Aside from the four infinite families, there is a small number of exceptional root systems, which are also very interesting objects. Their Tutte polynomials are computed in [DCP08b].

• Since the matroid of the complete graph  $K_n$  is isomorphic to the matroid of the vector configuration  $A_{n-1}$ , the first formula above is a formula for the Tutte polynomials of the complete graphs, proved originally by Tutte [Tut67].

Similarly, the coboundary polynomials of the complete graphs  $K_{m,n}$  are given by

$$1 + X \sum_{\substack{m,n \ge 0 \\ (m,n) \neq (0,0)}} \overline{\chi}_{K_{m,n}}(X,Y) \frac{Z_1^m}{m!} \frac{Z_2^n}{n!} = \left(\sum_{m,n \ge 0} Y^{mn} \frac{Z_1^m}{m!} \frac{Z_2^n}{n!}\right)^X.$$

• [BR, Mph00]. The Tutte polynomial of the arrangement  $\mathcal{A}(p, n)$  of all linear hyperplanes in  $\mathbb{F}_p^n$  is best expressed in terms of a "*p*-exponential generating function":

$$\sum_{n \ge 0} \overline{\chi}_{\mathcal{A}(p,n)}(X,Y) \frac{u^n}{(p;p)_n} = \frac{(u;p)_{\infty}}{(Xu;p)_{\infty}} \sum_{n \ge 0} Y^{1+p+\dots+p^{n-1}} \frac{u^n}{(p;p)_n}$$

where  $(a; p)_{\infty} = (1 - a)(1 - pa)(1 - p^2 a) \cdots$  and  $(a; p)_n = (1 - a)(1 - pa) \cdots (1 - p^{n-1}a)$ .

• Equation (27) expresses the Tutte polynomial of a graph G as the number of q-colorings of G, weighted by the number of improperly colored edges. When G has a path-like or cycle-like structure, one may use the transfer-matrix method of Section 3.1.2 to carry out that enumeration. This was done for the grid graphs  $L_{m,n}$  for small fixed m [CMNN03] and for the wheel graphs  $W_n$  [CS01] shown in Figure 7.7. For example, the Tutte polynomial of the wheel graph is

$$T_{W_n}(x,y) = \frac{1}{2^n} \left( b + \sqrt{b^2 - 4a} \right)^n + \frac{1}{2^n} \left( b - \sqrt{b^2 - 4a} \right)^n + xy - x - y - 1$$

where b = 1 + x + y and a = xy.



Figure 7.7: The graphs  $L_{4,8}$  and  $W_8$ .

# 7.9 Generalizations of the Tutte polynomial

The multivariate Tutte polynomial. The multivariate Tutte polynomial of a matroid M is

$$\widetilde{Z}_M(q; \mathbf{w}) = \sum_{A \subseteq E} q^{-r(A)} \prod_{e \in A} w_e$$

where q and  $(w_e)_{e \in E}$  are indeterminates. When all  $w_e = w$ , we get  $Z(q, w) = w^{v-c}q^cT(\frac{q}{w}+1, w+1)$ .

Notice that  $\widetilde{Z}_M(q; \mathbf{w})$  determines E and r, and hence it determines M. Still, it is a useful encoding of the matroid.

• We saw in Section 7.7 that for a graph G and a positive integer q, the multivariate Tutte polynomial is equal to the partition function of the q-state Potts model on G.

• For a vector  $\mathbf{a} \in \mathbb{N}^n$ , let  $M(\mathbf{a})$  be the matroid M where each element e is replaced by  $a_e$  copies of e. It is natural to ask for the Tutte polynomials of the various matroids  $M(\mathbf{a})$ . For each  $\mathbf{a}$ ,

$$T_{M(\mathbf{a})}(x,y) = (x-1)^{r(\text{supp}(\mathbf{a}))} \widetilde{Z}_M\left((x-1)(y-1); y^{a_1-1}, \dots, y^{a_n-1}\right)$$

The generating function for the Tutte polynomials of **all** the matroids  $M(\mathbf{a})$  turns out to be equivalent to the multivariate Tutte polynomial, disguised under a change of variable:

$$\sum_{\mathbf{a}in\mathbb{N}^n} \frac{T_{M(\mathbf{a})}(x,y)}{(x-1)^{r(\mathrm{supp}(\mathbf{a}))}} w_1^{a_1} \cdots w_n^{a_n} = \frac{1}{\prod_{i=1}^n (1-w_i)} \widetilde{Z}_M\left((x-1)(y-1); \frac{(y-1)w_1}{1-yw_1}, \dots, \frac{(y-1)w_n}{1-yw_n}\right)$$

• For an algebraic interpretation of the multivariate Tutte polynomial, see [AP10].

The arithmetic Tutte polynomial When we have a collection  $A \subseteq \mathbb{Z}^n$  of integer vectors, there is a variant of the Tutte polynomial that is quite useful. The arithmetic Tutte polynomial is

$$M_A(x,y) = \sum_{B \subseteq A} m(B)(x-1)^{r(A)-r(B)}(y-1)^{|B|-r(B)}$$

where, for each  $B \subseteq A$ , the *multiplicity* m(B) is the index of  $\mathbb{Z}B$  as a sublattice of  $(\operatorname{span} B) \cap \mathbb{Z}^n$ . Using the vectors in B as the columns of a matrix, m(B) is the greatest common divisor of the minors of full rank. This polynomial is related to the zonotope of A [Sta91, DM12]:

- The volume of the zonotope Z(A) is  $M_A(1,1)$ .
- The zonotope Z(A) contains  $M_A(2,1)$  lattice points,  $M_A(0,1)$  of which are in its interior.
- The Ehrhart polynomial of the zonotope Z(A) is  $q^r M(1 + \frac{1}{a}, 1)$ .

Let  $T = \text{Hom}(\mathbb{Z}^n, G)$  be the group of homomorphisms from  $\mathbb{Z}^n$  to a multiplicative group G, such as the unit circle  $\mathbb{S}^1$  or  $\mathbb{F}^* = \mathbb{F} \setminus \{0\}$  for a field  $\mathbb{F}$ . Each element  $a \in A$  determines a (hyper)torus  $T_a = \{t \in T : t(a) = 1\}$  in T. For instance a = (2, -3, 5) gives the torus  $x^2y^{-3}z^5 = 1$ . Let

$$\mathcal{T}(A) = \{T_a : a \in A\}, \qquad R(A) = T \setminus \bigcup_{a \in (A)} T_a$$

be the *toric arrangement* of A and its complement, respectively.

- [ERS09, Moc12] If  $G = \mathbb{S}^1$ , the number of regions of  $R(\mathcal{A})$  in the torus  $(\mathbb{S}^1)^r$  is  $M_A(1,0)$ .
- [DCP05, Moc12] If  $G = \mathbb{C}^*$ , the Poincaré polynomial of  $R(\mathcal{A})$  is  $q^r M(2 + \frac{1}{q}, 0)$ .

• [BM12, ACH14] If  $G = \mathbb{F}_q^*$ , the number of elements of  $R(\mathcal{A})$  is  $(-1)^r q^{n-r} M(1-q,0)$ . Furthermore,

$$\sum_{p \in T} t^{h(p)} = (t-1)^r q^{n-r} M_A\left(\frac{q+t-1}{t-1}, t\right)$$

where h(p) is the number of tori of  $\mathcal{T}(A)$  that p lies on.

As with ordinary Tutte polynomials, this last result may be used as a **finite field method** to compute arithmetic Tutte polynomials for some vector configurations A. However, at the moment there are very few results along these lines. One exception is the case of root systems; the study of their geometric properties motivates much of the theory of arithmetic Tutte polynomials. Explicit formulas for the arithmetic Tutte polynomials of the classical root systems  $A_n, B_n, C_n$ , and  $D_n$  are given in [ACH14].

#### 7.10 Matroid subdivisions, valuations, and the Derksen-Fink invariant.

A very interesting recent developments in matroid theory has been the study of matroid subdivisions and valuative matroid invariants. We offer a brief account of some key results and some pointers to the relevant bibliography.

A matroid subdivision is a polyhedral subdivision  $\mathcal{P}$  of a matroid polytope  $P_M$  where every polytope  $P \in \mathcal{P}$  is itself a matroid polytope. Equivalently, it is a subdivision of  $P_M$  whose only edges are the edges of  $P_M$ . In the most important case, M is the uniform matroid  $U_{d,n}$  and  $P_M$  is the hypersimplex  $\Delta(d, n)$ .



Figure 7.8: A matroid subdivision of  $\Delta(2, 4)$ .

Matroid subdivisions arose in algebraic geometry [HKT06, Kap93, Laf03], in the theory of valuated matroids [DW92, Mur96], and in tropical geometry [Spe08]. For instance, Lafforgue showed that if a matroid polytope  $P_M$  has no nontrivial matroid subdivisions, then the matroid M has (up to trivial transformations) only finitely many realizations over a fixed field  $\mathbb{F}$ . This is one of very few results about realizability of matroids over arbitrary fields.

The connection with tropical geometry is a rich source of examples. Let K be the field of Puiseux series  $a_0 x^{m/N} + a_1 x^{(m+1)/N} + \cdots$  (where  $a_0, a_1, \ldots \in \mathbb{C}$ ,  $m, N \in \mathbb{Z}$  and N > 0). Roughly speaking, every subspace L of  $K^n$  may be *tropicalized*, and there is a canonical way of decomposing the resulting tropical linear space into Bergman fans of various matroids. These matroids give a matroid subdivision of  $\Delta(d, n)$ . For some nice choices of L, the corresponding matroid subdivisions can be described explicitly. For details, see for example [AK06, Rin13, Spe08].

There is also a useful connection with the subdivisions of the product of simplices  $\Delta_{d-1} \times \Delta_{n-d-1}$ , which are much better understood, as we discussed in Section 5.4. Note that the vertex figure of

any vertex of  $\Delta(d, n)$  is  $\Delta_{d-1} \times \Delta_{n-d-1}$ . Every matroid subdivision of  $\Delta(d, n)$  then induces a "local" polyhedral subdivision of  $\Delta_{d-1} \times \Delta_{n-d-1}$  at each one of its vertices. Conversely, we can "cone" any subdivision S of  $\Delta_{d-1} \times \Delta_{n-d-1}$  to get a matroid subdivision that looks like S at a given vertex. [HJS11, Rin13]

**Enumerative aspects of matroid subdivisions.** Say a face P of a subdivision  $\mathcal{P}$  of  $P_M$  is *internal* if it is not on the boundary of  $P_M$ . Let  $\mathcal{P}^{\text{int}}$  be the set of internal faces of  $\mathcal{P}$ . Currently, the most interesting enumerative question on matroid subdivisions is the following.

**Conjecture 7.10.1.** (Speyer's f-vector conjecture [Spe08]) A matroid subdivision of the hypersimplex  $\Delta(d, n)$  has at most  $\frac{(n-c-1)!}{(d-c)!(n-d-c)!(c-1)!}$  interior faces of dimension n-c, with equality if and only if all facets correspond to series-parallel matroids.

Speyer constructed a subdivision simultaneously achieving the conjectural maximum number of interior faces for all c. In attempting to prove this conjecture, he pioneered the study of valuative matroid invariants. We say that a matroid invariant f is **valuative** if for any matroid subdivision  $\mathcal{P}$  of any matroid polytope  $P_M$  we have

$$f(M) = \sum_{P_{M_i} \in \mathcal{P}^{\text{int}}} (-1)^{\dim P_M - \dim P_{M_i}} f(M_i).$$

There are obvious matroid valuations, such as the volume or (thanks to Ehrhart reciprocity) the Ehrhart polynomial of  $P_M$ . Much more remarkably, we have the following result.

**Theorem 7.10.1.** [Spe08] The Tutte polynomial  $T_M(x, y)$  is a matroid valuation.

**Corollary 7.10.2.** The f-vector conjecture is true for c = 1.

Proof. By Theorem 7.10.1 the beta invariant  $\beta(M) = [x^1 y^0] T_M(x, y)$  is also a matroid valuation. Recall that  $\beta(N) = 0$  if and only if N is not connected (or, equivalently, if  $P_N$  is not a facet of  $\mathcal{P}$ ),  $\beta(N) = 1$  if and only if N is series-parallel, and  $\beta(N) \ge 2$  otherwise. Also  $\beta(U_{d,n}) = \binom{n-2}{d-1}$ . Therefore, in a matroid subdivision of  $\Delta(d, n)$  we have

$$\binom{n-2}{d-1} = \beta(U_{d,n}) = \sum_{\substack{P_N \in \mathcal{P} \\ P \text{ facet}}} \beta(N) \ge (\text{number of facets of } \mathcal{P}).$$

with equality if and only if every facet is a series-parallel matroid.

For matroids M realizable over some field of characteristic 0, Speyer [Spe09] used the K-theory of the Grassmannian to construct another polynomial invariant  $g_M(t)$ , which he used to prove the f-vector conjecture for matroid subdivisions whose matroids are realizable in characteristic 0.

**Theorem 7.10.3.** [Spe09] The f-vector conjecture is true for matroid subdivisions consisting of matroids which are representable over a field of characteristic 0.

Speyer's proof of this result relies on the nonnegativity of  $g_M(t)$ , which is proved geometrically for matroids realizable in characteristic 0. The nonnegativity of  $g_M(t)$  for all M, which would prove Conjecture 7.10.1 in full generality, remains open.

**The Derksen-Fink invariant.** Many other natural matroid functions were later discovered to be valuative. [AFR10, BJR09, Der09]. An example of a very general valuation on matroid polytopes from [AFR10] is the formal sum  $R(M) = \sum_{A \subseteq E} R_{A,r(A)}$  of symbols of the form  $R_{S,k}$  where S is a subset of E and k is an integer. Many other nice valuations can be obtained from this one. In fact, the matroid M can clearly be recovered from R(M).<sup>17</sup>

Eventually, Derksen [Der09] constructed a valuative matroid invariant, which he and Fink proved to be universal. [DF10] We call it the Derksen–Fink invariant. They gave several versions; we present one which is particularly simple to define.

Let  $Matroids_{n,r}$  be the set of matroids on [n] of rank r. Let  $C_n$  be the set of n! complete chains  $S: \emptyset = S_0 \subsetneq S_1 \subsetneq \cdots \subsetneq S_{n-1} \subsetneq S_n = [n]$ . For a complete chain S and a matroid M = ([n], r), let  $r[S_{i-1}, S_i] = r(S_i) - r(S_{i-1})$  for  $1 \le i \le n$ ; this is always 0 or 1. The **Derksen–Fink invariant** of M is

$$\mathcal{G}(M) = \sum_{S \in \mathcal{C}_n} U_{(r[S_0, S_1], r[S_1, S_2], \dots, r[S_{n-1}, S_n])}$$

in the vector space **U** spanned by the  $\binom{n}{r}$  formal symbols  $U_{w_1\cdots w_n}$  where  $w_1, \ldots, w_n \in \{0, 1\}$  and  $w_1 + \cdots + w_n = r$ .

**Theorem 7.10.4.** [DF10] The Derksen–Fink invariant  $\mathcal{G}$ : Matroids<sub>n,r</sub>  $\rightarrow \mathbf{U}$  is a universal valuative matroid invariant; that is, for any valuative matroid invariant f: Matroids<sub>n,r</sub>  $\rightarrow V$  there is a linear map  $\pi : \mathbf{U} \rightarrow V$  such that  $f = \pi \circ \mathcal{G}$ .

**Acknowledgments.** I am extremely grateful to my combinatorics teachers, Richard Stanley, Gian-Carlo Rota, Sergey Fomin, Sara Billey, and Bernd Sturmfels (in chronological order); their influence on the way that I understand combinatorics is apparent in these pages. I've also learned greatly from my wonderful collaborators and students.

Several superb surveys have been instrumental in the preparation of this manuscript; they are mentioned throughout the text, but it is worth acknowledging a few that I found especially help-ful: Aigner's A Course in Enumeration, Beck and Robins' Computing the continuous discretely, Brylawski and Oxley's The Tutte polynomial and its applications, Flajolet and Sedgewick's Analytic Combinatorics, Krattenthaler's Advanced Determinantal Calculus, and Stanley's Enumerative Combinatorics and Lectures on hyperplane arrangements.

I am very grateful to Miklós Bóna for the invitation to contribute to this project. Writing this survey has not been an easy task, but I have enjoyed it immensely, and have learned an enormous amount. I also thank Alex Fink, David Speyer, and the anonymous referee for their helpful comments. Finally, and most importantly, I am so very thankful to May-Li for her patience, help, and support during my months of hard work on this project.

<sup>&</sup>lt;sup>17</sup>However, note that R is not a matroid isomorphism invariant.

# References

- [AB07] Federico Ardila and Sara Billey, Flag arrangements and triangulations of products of simplices, Adv. Math. 214 (2007), no. 2, 495–524. MR 2349710 (2008k:32080)
- [ABD10] Federico Ardila, Carolina Benedetti, and Jeffrey Doker, *Matroid polytopes and their volumes*, Discrete Comput. Geom. **43** (2010), no. 4, 841–854. MR 2610473 (2012b:52026)
- [ABH<sup>+</sup>11] Federico Ardila, Matthias Beck, Serkan Hosten, Julian Pfeifle, and Kim Seashore, Root polytopes and growth series of root lattices, SIAM J. Discrete Math. 25 (2011), no. 1, 360–378. MR 2801233 (2012e:52043)
- [ACH14] Federico Ardila, Federico Castillo, and Michael Henley, *The arithmetic tutte polynomials of the classical root systems*, 2014.
- [AD09] Federico Ardila and Mike Develin, Tropical hyperplane arrangements and oriented matroids, Math. Z.
   262 (2009), no. 4, 795–816. MR 2511751 (2010i:52031)
- [Adi96] Ron M. Adin, A new cubical h-vector, Proceedings of the 6th Conference on Formal Power Series and Algebraic Combinatorics (New Brunswick, NJ, 1994), vol. 157, 1996, pp. 3–14. MR 1417283 (97i:52010)
- [AFR10] Federico Ardila, Alex Fink, and Felipe Rincón, Valuations for matroid polytope subdivisions, Canad.
   J. Math. 62 (2010), no. 6, 1228–1245. MR 2760656 (2012c:05075)
- [Aig07] Martin Aigner, A course in enumeration, Graduate Texts in Mathematics, vol. 238, Springer, Berlin, 2007. MR 2339282 (2008f:05001)
- [AK06] Federico Ardila and Caroline J. Klivans, *The Bergman complex of a matroid and phylogenetic trees*, J. Combin. Theory Ser. B **96** (2006), no. 1, 38–49. MR 2185977 (2006i:05034)
- [AP10] Federico Ardila and Alexander Postnikov, Combinatorics and geometry of power ideals, Trans. Amer. Math. Soc. 362 (2010), no. 8, 4357–4384. MR 2608410 (2011g:05322)
- [AR12] Drew Armstrong and Brendon Rhoades, The Shi arrangement and the Ish arrangement, Trans. Amer. Math. Soc. 364 (2012), no. 3, 1509–1528. MR 2869184
- [Ard03] Federico Ardila, *The Catalan matroid*, J. Combin. Theory Ser. A **104** (2003), no. 1, 49–62. MR 2018420 (2004k:05052)
- [Ard04] \_\_\_\_\_, The coefficients of a Fibonacci power series, Fibonacci Quart. **42** (2004), no. 3, 202–204. MR 2093874
- [Ard07] \_\_\_\_\_, Computing the Tutte polynomial of a hyperplane arrangement, Pacific J. Math. 230 (2007), no. 1, 1–26. MR 2318445 (2008g:52034)
- [Arm13] Drew Armstrong, Hyperplane arrangements and diagonal harmonics, J. Comb. 4 (2013), no. 2, 157–190. MR 3096132
- [ARW13a] Federico Ardila, Felipe Rincón, and Lauren Williams, Positively oriented matroids are realizable, 2013.
- [ARW13b] \_\_\_\_\_, Positroids and non-crossing partitions, 2013.
- [AS02] M. Azaola and F. Santos, The number of triangulations of the cyclic polytope C(n, n 4), Discrete Comput. Geom. **27** (2002), no. 1, 29–48, Geometric combinatorics (San Francisco, CA/Davis, CA, 2000). MR 1871687 (2002):52007)
- [Ath96] Christos A. Athanasiadis, Characteristic polynomials of subspace arrangements and finite fields, Adv. Math. 122 (1996), no. 2, 193–233. MR 1409420 (97k:52012)
- [Ath98] \_\_\_\_\_, On free deformations of the braid arrangement, European J. Combin. **19** (1998), no. 1, 7–18. MR 1600259 (99d:52008)
- [Ath99] \_\_\_\_\_, Extended Linial hyperplane arrangements for root systems and a conjecture of Postnikov and Stanley, J. Algebraic Combin. **10** (1999), no. 3, 207–225. MR 1723184 (2000i:52039)
- [AZ01] Tewodros Amdeberhan and Doron Zeilberger, Determinants through the looking glass, Adv. in Appl. Math. 27 (2001), no. 2-3, 225–230, Special issue in honor of Dominique Foata's 65th birthday (Philadelphia, PA, 2000). MR 1868963 (2002h:15013)
- [BB85] Margaret M. Bayer and Louis J. Billera, Generalized Dehn-Sommerville relations for polytopes, spheres and Eulerian partially ordered sets, Invent. Math. 79 (1985), no. 1, 143–157. MR 774533 (86f:52010b)

- [BB97] Louis J. Billera and Anders Björner, Face numbers of polytopes and complexes, Handbook of discrete and computational geometry, CRC Press Ser. Discrete Math. Appl., CRC, Boca Raton, FL, 1997, pp. 291–310. MR 1730171
- [BB05] Anders Björner and Francesco Brenti, *Combinatorics of Coxeter Groups*, Graduate Texts in Mathematics, vol. 231, Springer, April 2005.
- [BdMN03] Joseph Bonin, Anna de Mier, and Marc Noy, Lattice path matroids: enumerative aspects and Tutte polynomials, J. Combin. Theory Ser. A 104 (2003), no. 1, 63–94. MR 2018421 (2004j:05035)
- [BER97] Louis J. Billera, Richard Ehrenborg, and Margaret Readdy, The c-2d-index of oriented matroids, J. Combin. Theory Ser. A 80 (1997), no. 1, 79–105. MR 1472106 (98h:05051)
- [BGW03] Alexandre V. Borovik, I. M. Gelfand, and Neil White, *Coxeter matroids*, Progress in Mathematics, vol. 216, Birkhäuser Boston, Inc., Boston, MA, 2003. MR 1989953 (2004i:05028)
- [Bjö82] Anders Björner, On the homology of geometric lattices, Algebra Universalis 14 (1982), no. 1, 107–128. MR 634422 (83d:05029)
- [BJR09] Louis J. Billera, Ning Jia, and Victor Reiner, A quasisymmetric function for matroids, European J. Combin. 30 (2009), no. 8, 1727–1757. MR 2552658 (2010m:05058)
- [BK91] Margaret M. Bayer and Andrew Klapper, A new index for polytopes, Discrete Comput. Geom. 6 (1991), no. 1, 33–47. MR 1073071 (91k:52024)
- [BL81] Louis J. Billera and Carl W. Lee, A proof of the sufficiency of McMullen's conditions for f-vectors of simplicial convex polytopes, J. Combin. Theory Ser. A 31 (1981), no. 3, 237–255. MR 635368 (82m:52006)
- [BLVS<sup>+</sup>99] Anders Björner, Michel Las Vergnas, Bernd Sturmfels, Neil White, and Günter M. Ziegler, Oriented matroids, second ed., Encyclopedia of Mathematics and its Applications, vol. 46, Cambridge University Press, Cambridge, 1999. MR 1744046 (2000j:52016)
- [BM12] Petter Brndn and Luca Moci, *The multivariate arithmetic tutte polynomial*, 2012.
- [BMM<sup>+</sup>12] L. J. Billera, J. Tatch Moore, C. Dufort Moraites, Y. Wang, and K. Williams, Maximal unbalanced families, 2012.
- [BN07] Matthew Baker and Serguei Norine, *Riemann-Roch and Abel-Jacobi theory on a finite graph*, Adv. Math. **215** (2007), no. 2, 766–788. MR 2355607 (2008m:05167)
- [BO92] Thomas Brylawski and James Oxley, The Tutte polynomial and its applications, Matroid applications, Encyclopedia Math. Appl., vol. 40, Cambridge Univ. Press, Cambridge, 1992, pp. 123–225. MR 1165543 (93k:05060)
- [BP93] J. Bétréma and J.-G. Penaud, Animaux et arbres guingois, Theoret. Comput. Sci. 117 (1993), no. 1-2, 67–89, Conference on Formal Power Series and Algebraic Combinatorics (Bordeaux, 1991). MR 1235169 (94e:05079)
- [BP99] David Bressoud and James Propp, *How the alternating sign matrix conjecture was solved*, Notices Amer. Math. Soc. **46** (1999), no. 6, 637–646. MR 1691562 (2000k:15002)
- [BPvdP12] N. Bansal, R. A. Pendavingh, and J. G. van der Pol, On the number of matroids, 2012.
- [BR] Michael Barany and Victor Reiner, *The tutte polynomial of a finite projective space*.
- [BR07] Matthias Beck and Sinai Robins, *Computing the continuous discretely*, Undergraduate Texts in Mathematics, Springer, New York, 2007, Integer-point enumeration in polyhedra. MR 2271992 (2007h:11119)
- [Bry86] Thomas Brylawski, Appendix of matroid cryptomorphisms, Theory of matroids, Encyclopedia Math. Appl., vol. 26, Cambridge Univ. Press, Cambridge, 1986, pp. 298–316. MR 849400
- [BV99] Michel Brion and Michèle Vergne, Arrangement of hyperplanes. I. Rational functions and Jeffrey-Kirwan residue, Ann. Sci. École Norm. Sup. (4) 32 (1999), no. 5, 715–741. MR 1710758 (2001e:32039)
- [BV08] Welleda Baldoni and Michèle Vergne, Kostant partitions functions and flow polytopes, Transform. Groups 13 (2008), no. 3-4, 447–469. MR 2452600 (2009k:52029)
- [BW82] Anders Björner and Michelle Wachs, Bruhat order of Coxeter groups and shellability, Adv. in Math. 43 (1982), no. 1, 87–100. MR 644668 (83i:20043)

- [CDS95] Dragos M. Cvetković, Michael Doob, and Horst Sachs, *Spectra of graphs*, third ed., Johann Ambrosius Barth, Heidelberg, 1995, Theory and applications. MR 1324340 (96b:05108)
- [CLS14] Cesar Ceballos, Jean-Philippe Labbé, and Christian Stump, Subword complexes, cluster complexes, and generalized multi-associahedra, J. Algebraic Combin. 39 (2014), no. 1, 17–51. MR 3144391
- [CMNN03] N. Calkin, C. Merino, S. Noble, and M. Noy, Improved bounds for the number of forests and acyclic orientations in the square lattice, Electron. J. Combin. 10 (2003), Research Paper 4, 18 pp. (electronic). MR 1975754 (2004d:05093)
- [CR70] Henry H. Crapo and Gian-Carlo Rota, On the foundations of combinatorial theory: Combinatorial geometries, preliminary ed., The M.I.T. Press, Cambridge, Mass.-London, 1970. MR 0290980 (45 #74)
- [Cra69] Henry H. Crapo, The Tutte polynomial, Aequationes Math. 3 (1969), 211–229. MR 0262095 (41 #6705)
- [CS01] S.-C. Chang and R. Shrock, Zeros of Jones polynomials for families of knots and links, Phys. A 301 (2001), no. 1-4, 196–218. MR 1874776 (2002k:57006)
- [CS05] Henry Crapo and William Schmitt, A free subalgebra of the algebra of matroids, European J. Combin.
   26 (2005), no. 7, 1066–1085. MR 2155892 (2006e:05035)
- [CSZ11] Cesar Ceballos, Francisco Santos, and Gnter M. Ziegler, Many non-equivalent realizations of the associahedron, 2011.
- [dB46] N. G. de Bruijn, A combinatorial problem, Nederl. Akad. Wetensch., Proc. 49 (1946), 758-764 =Indagationes Math. 8, 461–467 (1946). MR 0018142 (8,247d)
- [DCP05] C. De Concini and C. Procesi, On the geometry of toric arrangements, Transform. Groups 10 (2005), no. 3-4, 387–422. MR 2183118 (2006m:32027)
- [DCP08a] \_\_\_\_\_, Hyperplane arrangements and box splines, Michigan Math. J. 57 (2008), 201–225, With an appendix by A. Björner, Special volume in honor of Melvin Hochster. MR 2492449 (2010f:52037)
- [DCP08b] \_\_\_\_\_, The zonotope of a root system, Transform. Groups **13** (2008), no. 3-4, 507–526. MR 2452603 (2009i:17013)
- [Der09] Harm Derksen, Symmetric and quasi-symmetric functions associated to polymatroids, J. Algebraic Combin. 30 (2009), no. 1, 43–86. MR 2519849 (2010e:05052)
- [DF10] Harm Derksen and Alex Fink, Valuative invariants for polymatroids, Adv. Math. **225** (2010), no. 4, 1840–1892. MR 2680193 (2011k:52016)
- [Dia09] Hansheng Diao, A poset structure on quasifibonacci partitions, J. Combin. Theory Ser. A 116 (2009), no. 8, 1361–1373. MR 2568804 (2010m:05026)
- [Dil50] R. P. Dilworth, A decomposition theorem for partially ordered sets, Ann. of Math. (2) 51 (1950), 161–166. MR 0032578 (11,309f)
- [DKM02] A. Dress, J. H. Koolen, and V. Moulton, On line arrangements in the hyperbolic plane, European J. Combin. 23 (2002), no. 5, 549–557. MR 1931939 (2003h:52024)
- [DLKK12] Jesús A. De Loera, Yvonne Kemper, and Steven Klee, h-vectors of small matroid complexes, Electron.
   J. Combin. 19 (2012), no. 1, Paper 14, 11. MR 2880645
- [DLRS10] Jesús A. De Loera, Jörg Rambau, and Francisco Santos, *Triangulations*, Algorithms and Computation in Mathematics, vol. 25, Springer-Verlag, Berlin, 2010, Structures for algorithms and applications. MR 2743368 (2011j:52037)
- [DM85] Wolfgang Dahmen and Charles A. Micchelli, On the local linear independence of translates of a box spline, Studia Math. 82 (1985), no. 3, 243–263. MR 825481 (87k:41008)
- [DM12] Michele D'Adderio and Luca Moci, Ehrhart polynomial and arithmetic Tutte polynomial, European J. Combin. 33 (2012), no. 7, 1479–1483. MR 2923464
- [DS04] Mike Develin and Bernd Sturmfels, *Tropical convexity*, Doc. Math. **9** (2004), 1–27 (electronic). MR 2054977 (2005i:52010)
- [DW92] Andreas W. M. Dress and Walter Wenzel, Valuated matroids, Adv. Math. 93 (1992), no. 2, 214–250. MR 1164708 (93h:05045)
- [EF65] Jack Edmonds and D. R. Fulkerson, Transversals and matroid partition, J. Res. Nat. Bur. Standards Sect. B 69B (1965), 147–153. MR 0188090 (32 #5531)

- [EF05] Sen-Peng Eu and Tung-Shan Fu, A simple proof of the Aztec diamond theorem, Electron. J. Combin.
   12 (2005), Research Paper 18, 8 pp. (electronic). MR 2134181 (2006e:05037)
- [EMM11] Joanna A. Ellis-Monaghan and Criel Merino, Graph polynomials and their applications I: The Tutte polynomial, Structural analysis of complex networks, Birkhäuser/Springer, New York, 2011, pp. 219– 255. MR 2777919 (2012e:05004)
- [ERS09] Richard Ehrenborg, Margaret Readdy, and Michael Slone, *Affine and toric hyperplane arrangements*, Discrete Comput. Geom. **41** (2009), no. 4, 481–512. MR 2496314 (2010a:52035)
- [FGS06] Philippe Flajolet, Stefan Gerhold, and Bruno Salvy, On the non-holonomic character of logarithms, powers, and the nth prime function, Electron. J. Combin. 11 (2004/06), no. 2, Article 2, 16. MR 2195433 (2006h:05013)
- [FH91] William Fulton and Joe Harris, Representation theory, Graduate Texts in Mathematics, vol. 129, Springer-Verlag, New York, 1991, A first course, Readings in Mathematics. MR 1153249 (93a:20069)
- [FR07] Sergey Fomin and Nathan Reading, Root systems and generalized associahedra, Geometric combinatorics, IAS/Park City Math. Ser., vol. 13, Amer. Math. Soc., Providence, RI, 2007, pp. 63–131. MR 2383126
- [FS05] Eva Maria Feichtner and Bernd Sturmfels, Matroid polytopes, nested sets and Bergman fans, Port. Math. (N.S.) 62 (2005), no. 4, 437–468. MR 2191630 (2006j:05036)
- [FS09] Philippe Flajolet and Robert Sedgewick, Analytic combinatorics, Cambridge University Press, Cambridge, 2009. MR 2483235 (2010h:05005)
- [FSM94] C. Flye Sainte-Marie, Solution to question nr. 48, L'intermédiaire des Mathématiciens (1894), 107–110.
- [Fuj05] Satoru Fujishige, Submodular functions and optimization, second ed., Annals of Discrete Mathematics, vol. 58, Elsevier B. V., Amsterdam, 2005. MR 2171629 (2006d:90098)
- [GBV88] D. Gouyou-Beauchamps and G. Viennot, Equivalence of the two-dimensional directed animal problem to a one-dimensional path problem, Adv. in Appl. Math. 9 (1988), no. 3, 334–357. MR 956559 (90c:05009)
- [GGMS87] I. M. Gel'fand, R. M. Goresky, R. D. MacPherson, and V. V. Serganova, Combinatorial geometries, convex polyhedra, and Schubert cells, Adv. in Math. 63 (1987), no. 3, 301–316. MR 877789 (88f:14045)
- [GGW14] Jim Geelen, Bert Gerards, and Geoff Whittle, *Solving rota's conjecture*, Notices of the Amer. Math. Soc. **61** (2014), no. 7, 736–743.
- [GK04] Sudhir R. Ghorpade and Christian Krattenthaler, The Hilbert series of Pfaffian rings, Algebra, arithmetic and geometry with applications (West Lafayette, IN, 2000), Springer, Berlin, 2004, pp. 337–356. MR 2037100 (2005c:13010)
- [GK08] Andreas Gathmann and Michael Kerber, A Riemann-Roch theorem in tropical geometry, Math. Z. 259 (2008), no. 1, 217–230. MR 2377750 (2009a:14014)
- [GKZ08] I. M. Gelfand, M. M. Kapranov, and A. V. Zelevinsky, Discriminants, resultants and multidimensional determinants, Modern Birkhäuser Classics, Birkhäuser Boston, Inc., Boston, MA, 2008, Reprint of the 1994 edition. MR 2394437 (2009a:14065)
- [GM88] Mark Goresky and Robert MacPherson, Stratified Morse theory, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) [Results in Mathematics and Related Areas (3)], vol. 14, Springer-Verlag, Berlin, 1988. MR 932724 (90d:57039)
- [Gre76] Curtis Greene, Weight enumeration and the geometry of linear codes, Studies in Appl. Math. 55 (1976), no. 2, 119–128. MR 0447020 (56 #5335)
- [Grü03] Branko Grünbaum, Convex polytopes, second ed., Graduate Texts in Mathematics, vol. 221, Springer-Verlag, New York, 2003, Prepared and with a preface by Volker Kaibel, Victor Klee and Günter M. Ziegler. MR 1976856 (2004b:52001)
- [GV85] Ira Gessel and Gérard Viennot, Binomial determinants, paths, and hook length formulae, Adv. in Math.
   58 (1985), no. 3, 300–321. MR 815360 (87e:05008)
- [Hen91] Peter Henrici, Applied and computational complex analysis. Vol. 2, Wiley Classics Library, John Wiley & Sons, Inc., New York, 1991, Special functions—integral transforms—asymptotics—continued fractions, Reprint of the 1977 original, A Wiley-Interscience Publication. MR 1164865 (93b:30001)

[HJS11] Sven Herrmann, Michael Joswig, and David Speyer, Dressians, tropical grassmannians, and their rays, 2011. [HK81] Frank Harary and Jerald A. Kabell, Counting balanced signed graphs using marked graphs, Proc. Edinburgh Math. Soc. (2) 24 (1981), no. 2, 99-104. MR 625282 (83a:05072) [HK12] June Huh and Eric Katz, Log-concavity of characteristic polynomials and the Bergman fan of matroids, Math. Ann. 354 (2012), no. 3, 1103-1116. MR 2983081 [HKT06] Paul Hacking, Sean Keel, and Jenia Tevelev, Compactification of the moduli space of hyperplane arrangements, J. Algebraic Geom. 15 (2006), no. 4, 657–680. MR 2237265 (2007j:14016) [HL12] Takayuki Hibi and Nan Li, Unimodular equivalence of order and chain polytopes, 2012. [HLM08] Silvia Heubach, Nelson Y. Li, and Toufik Mansour, Staircase tilings and k-Catalan structures, Discrete Math. 308 (2008), no. 24, 5954–5964. MR 2464886 (2009k:05026) [HLO13] Takayuki Hibi, Nan Li, and Hidefumi Ohsugi, The face vector of a half-open hypersimplex, 2013. [HR11] Olga Holtz and Amos Ron, Zonotopal algebra, Adv. Math. 227 (2011), no. 2, 847-894. MR 2793025 (2012e:52065)[HS85] William A. Harris, Jr. and Yasutaka Sibuya, The reciprocals of solutions of linear ordinary differential equations, Adv. in Math. 58 (1985), no. 2, 119–132. MR 814747 (87j:34012) Jürgen Herzog and Ngô Viêt Trung, Gröbner bases and multiplicity of determinantal and Pfaffian [HT92] ideals, Adv. Math. 96 (1992), no. 1, 1–37. MR 1185786 (94a:13012) [Huh12] June Huh, Milnor numbers of projective hypersurfaces and the chromatic polynomial of graphs, J. Amer. Math. Soc. 25 (2012), no. 3, 907-927. MR 2904577 [Hum90] James E. Humphreys, *Reflection groups and Coxeter groups*, Cambridge Studies in Advanced Mathematics, vol. 29, Cambridge University Press, Cambridge, 1990. MR 1066460 (92h:20002) [Jae89] Francois Jaeger, On Tutte polynomials of matroids representable over GF(q), European J. Combin. 10 (1989), no. 3, 247–255. MR 1029171 (91a:05022) [Jon85] Vaughan F. R. Jones, A polynomial invariant for knots via von Neumann algebras, Bull. Amer. Math. Soc. (N.S.) 12 (1985), no. 1, 103–111. MR 766964 (86e:57006) [Jon03] Jakob Jonsson, Generalized triangulations of the n-gon, 2003. [Jon05] , Generalized triangulations and diagonal-free subsets of stack polyominoes, J. Combin. Theory Ser. A 112 (2005), no. 1, 117-142. MR 2167478 (2006d:05011) [Jun31] R. Jungen, Sur les séries de Taylor n'ayant que des singularités algébrico-logarithmiques sur leur cercle de convergence, Comment. Math. Helv. 3 (1931), no. 1, 266-306. MR 1509439 [Kak00] Subhash Kak, Yamātārājabhānasalagām: an interesting combinatoric sūtra, Indian J. Hist. Sci. 35 (2000), no. 2, 123–127. MR 1820110 (2001m:01013) [Kap93] M. M. Kapranov, Chow quotients of Grassmannians. I, I. M. Gel'fand Seminar, Adv. Soviet Math., vol. 16, Amer. Math. Soc., Providence, RI, 1993, pp. 29–110. MR 1237834 (95g:14053) [Kas63] P. W. Kasteleyn, Dimer statistics and phase transitions, J. Mathematical Phys. 4 (1963), 287–293. MR 0153427 (27 #3394) [KL79] David Kazhdan and George Lusztig, Representations of Coxeter groups and Hecke algebras, Invent. Math. 53 (1979), no. 2, 165-184. MR 560412 (81j:20066) [Kli07] Caroline J. Klivans, Threshold graphs, shifted complexes, and graphical complexes, Discrete Math. 307 (2007), no. 21, 2591-2597. MR 2359603 (2008j:05373) [KM59] Samuel Karlin and James McGregor, Coincidence probabilities, Pacific J. Math. 9 (1959), 1141–1164. MR 0114248 (22 #5072) [KN86] Joseph P. S. Kung and Hien Q. Nguyen, Weak maps, Theory of matroids, Encyclopedia Math. Appl., vol. 26, Cambridge Univ. Press, Cambridge, 1986, pp. 254–271. MR 849398 [Knu74] Donald E. Knuth, The asymptotic number of geometries, J. Combinatorial Theory Ser. A 16 (1974), 398-400. MR 0335312 (49 #94)

- [KP04] Michael Korn and Igor Pak, Tilings of rectangles with T-tetrominoes, Theoret. Comput. Sci. 319 (2004), no. 1-3, 3–27. MR 2074945 (2005d:52041)
- [KP13] Matjaž Konvalinka and Igor Pak, Triangulations of Cayley and Tutte polytopes, Adv. Math. 245 (2013), 1–33. MR 3084421
- [Kra99] C. Krattenthaler, Advanced determinant calculus, Sém. Lothar. Combin. **42** (1999), Art. B42q, 67 pp. (electronic), The Andrews Festschrift (Maratea, 1998). MR 1701596 (2002i:05013)
- [Kra05] \_\_\_\_\_, Advanced determinant calculus: a complement, Linear Algebra Appl. **411** (2005), 68–166. MR 2178686 (2006g:05022)
- [Kre72] G. Kreweras, Sur les partitions non croisées d'un cycle, Discrete Math. 1 (1972), no. 4, 333–350. MR 0309747 (46 #8852)
- [KS14] Steven Klee and Jose Alejandro Samper, Lexicographic shellability, matroids and pure order ideals, 2014.
- [Laf03] L. Lafforgue, Chirurgie des grassmanniennes, CRM Monograph Series, vol. 19, American Mathematical Society, Providence, RI, 2003. MR 1976905 (2004k:14085)
- [Las78] Alain Lascoux, Classes de Chern d'un produit tensoriel, C. R. Acad. Sci. Paris Sér. A-B 286 (1978), no. 8, A385–A387. MR 0476745 (57 #16302)
- [Liu05] Fu Liu, Ehrhart polynomials of cyclic polytopes, J. Combin. Theory Ser. A 111 (2005), no. 1, 111–127. MR 2144858 (2006a:05012)
- [LP07] Thomas Lam and Alexander Postnikov, Alcoved polytopes. I, Discrete Comput. Geom. 38 (2007), no. 3, 453–478. MR 2352704 (2008m:52031)
- [LP09] László Lovász and Michael D. Plummer, *Matching theory*, AMS Chelsea Publishing, Providence, RI, 2009, Corrected reprint of the 1986 original [MR0859549]. MR 2536865
- [LV88] Michel Las Vergnas, On the evaluation at (3,3) of the Tutte polynomial of a graph, J. Combin. Theory Ser. B 45 (1988), no. 3, 367–372. MR 969255 (89i:05105)
- [Mac36] Saunders MacLane, Some Interpretations of Abstract Linear Dependence in Terms of Projective Geometry, Amer. J. Math. 58 (1936), no. 1, 236–240. MR 1507146
- [Mac63] Jessie MacWilliams, A theorem on the distribution of weights in a systematic code, Bell System Tech.
   J. 42 (1963), 79–94. MR 0149978 (26 #7462)
- [Mas72] J. H. Mason, On a class of matroids arising from paths in graphs, Proc. London Math. Soc. (3) 25 (1972), 55–74. MR 0311496 (47 #58)
- [McN03] Peter McNamara, EL-labelings, supersolvability and 0-Hecke algebra actions on posets, J. Combin. Theory Ser. A 101 (2003), no. 1, 69–89. MR 1953281 (2004c:05215)
- [Mer01] Criel Merino, The chip firing game and matroid complexes, Discrete models: combinatorics, computation, and geometry (Paris, 2001), Discrete Math. Theor. Comput. Sci. Proc., AA, Maison Inform. Math. Discret. (MIMD), Paris, 2001, pp. 245–255 (electronic). MR 1888777 (2002k:91048)
- [MM12] Karola Meszaros and Alejandro H. Morales, Flow polytopes of signed graphs and the kostant partition function, 2012.
- [MN13] Satoshi Murai and Eran Nevo, On the generalized lower bound conjecture for polytopes and spheres, Acta Math. **210** (2013), no. 1, 185–202. MR 3037614
- [MNRIVF12] Criel Merino, Steven D. Noble, Marcelino Ramírez-Ibáñez, and Rafael Villarroel-Flores, On the structure of the h-vector of a paving matroid, European J. Combin. 33 (2012), no. 8, 1787–1799. MR 2950481
- [MNWW11] Dillon Mayhew, Mike Newman, Dominic Welsh, and Geoff Whittle, On the asymptotic proportion of connected matroids, European J. Combin. **32** (2011), no. 6, 882–890. MR 2821559 (2012g:05039)
- [Moc12] Luca Moci, A Tutte polynomial for toric arrangements, Trans. Amer. Math. Soc. **364** (2012), no. 2, 1067–1088. MR 2846363
- [MP95] N. V. R. Mahadev and U. N. Peled, Threshold graphs and related topics, Annals of Discrete Mathematics, vol. 56, North-Holland Publishing Co., Amsterdam, 1995. MR 1417258 (97h:05001)

- [Mph00] Eunice Gogo Mphako, Tutte polynomials of perfect matroid designs, Combin. Probab. Comput. 9 (2000), no. 4, 363–367. MR 1786925 (2002f:05049)
- [MRIS12] Criel Merino, Marcelino Ramrez-Ibez, and Guadalupe Rodrguez Sanchez, *The tutte polynomial of some matroids*, 2012.
- [MS71] P. McMullen and G. C. Shephard, Convex polytopes and the upper bound conjecture, Cambridge University Press, London-New York, 1971, Prepared in collaboration with J. E. Reeve and A. A. Ball, London Mathematical Society Lecture Note Series, 3. MR 0301635 (46 #791)
- [Mur96] Kazuo Murota, Convexity and Steinitz's exchange property, Adv. Math. 124 (1996), no. 2, 272–311.
   MR 1424312 (97m:90070)
- [Nak00] Tomoki Nakamigawa, A generalization of diagonal flips in a convex polygon, Theoret. Comput. Sci. 235 (2000), no. 2, 271–282, Combinatorics and optimization (Okinawa, 1996). MR 1756126 (2001d:52035)
- [Niv69] Ivan Niven, Formal power series, Amer. Math. Monthly **76** (1969), 871–889. MR 0252386 (40 #5606)
- [Oh09] Suho Oh, Combinatorics of positroids, 21st International Conference on Formal Power Series and Algebraic Combinatorics (FPSAC 2009), Discrete Math. Theor. Comput. Sci. Proc., AK, Assoc. Discrete Math. Theor. Comput. Sci., Nancy, 2009, pp. 721–732. MR 2721556 (2011m:05084)
- [Oh13] \_\_\_\_\_, Generalized permutohedra, h-vectors of cotransversal matroids and pure O-sequences, Electron. J. Combin. **20** (2013), no. 3, Paper 14, 14. MR 3104512
- [OS80] Peter Orlik and Louis Solomon, Combinatorics and topology of complements of hyperplanes, Invent. Math. 56 (1980), no. 2, 167–189. MR 558866 (81e:32015)
- [OT92] Peter Orlik and Hiroaki Terao, Arrangements of hyperplanes, Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], vol. 300, Springer-Verlag, Berlin, 1992. MR 1217488 (94e:52014)
- [Ox111] James Oxley, *Matroid theory*, second ed., Oxford Graduate Texts in Mathematics, vol. 21, Oxford University Press, Oxford, 2011. MR 2849819 (2012k:05002)
- [Pos06] Alexander Postnikov, Total positivity, grassmannians, and networks, 2006.
- [Pos09] Alexander Postnikov, Permutohedra, associahedra, and beyond, Int. Math. Res. Not. IMRN (2009), no. 6, 1026–1106. MR 2487491 (2010g:05399)
- [PRW08] Alex Postnikov, Victor Reiner, and Lauren Williams, Faces of generalized permutohedra, Doc. Math. 13 (2008), 207–273. MR 2520477 (2010j:05425)
- [PS98] George Pólya and Gabor Szegő, Problems and theorems in analysis. II, Classics in Mathematics, Springer-Verlag, Berlin, 1998, Theory of functions, zeros, polynomials, determinants, number theory, geometry, Translated from the German by C. E. Billigheimer, Reprint of the 1976 English translation. MR 1492448
- [PS00] Alexander Postnikov and Richard P. Stanley, Deformations of Coxeter hyperplane arrangements, J. Combin. Theory Ser. A 91 (2000), no. 1-2, 544–597, In memory of Gian-Carlo Rota. MR 1780038 (2002g:52032)
- [PS04] Alexander Postnikov and Boris Shapiro, Trees, parking functions, syzygies, and deformations of monomial ideals, Trans. Amer. Math. Soc. 356 (2004), no. 8, 3109–3142 (electronic). MR 2052943 (2005a:05066)
- [PS06] Nicholas Proudfoot and David Speyer, A broken circuit ring, Beiträge Algebra Geom. 47 (2006), no. 1, 161–166. MR 2246531 (2007c:13029)
- [PSS99] Alexander Postnikov, Boris Shapiro, and Mikhail Shapiro, Algebras of curvature forms on homogeneous manifolds, Differential topology, infinite-dimensional Lie algebras, and applications, Amer. Math. Soc. Transl. Ser. 2, vol. 194, Amer. Math. Soc., Providence, RI, 1999, pp. 227–235. MR 1729365 (2001a:53079)
- [Ram97] Jörg Rambau, Triangulations of cyclic polytopes and higher Bruhat orders, Mathematika 44 (1997), no. 1, 162–194. MR 1464385 (98i:52019)
- [Rei99] Victor Reiner, An interpretation for the Tutte polynomial, European J. Combin. 20 (1999), no. 2, 149–161. MR 1676189 (2000b:05037)

[Ret13]	Vladimir Retakh, Israel moiseevich gelfand, part i, Notices of the Amer. Math. Soc <b>60</b> (2013), no. 1, 24–49.
[Rin13]	Felipe Rincón, <i>Local tropical linear spaces</i> , Discrete Comput. Geom. <b>50</b> (2013), no. 3, 700–713. MR 3102586
[Rob91]	David P. Robbins, <i>The story of</i> $1, 2, 7, 42, 429, 7436, \cdots$ , Math. Intelligencer <b>13</b> (1991), no. 2, 12–19. MR 1098216 (92d:05010)
[Rob96]	Neville Robbins, <i>Fibonacci partitions</i> , Fibonacci Quart. <b>34</b> (1996), no. 4, 306–313. MR 1394758 (97i:11105)
[Rot64]	Gian-Carlo Rota, On the foundations of combinatorial theory. I. Theory of Möbius functions, Z. Wahrscheinlichkeitstheorie und Verw. Gebiete <b>2</b> (1964), 340–368 (1964). MR 0174487 (30 #4688)
[Rot71]	, Combinatorial theory, old and new, Actes du Congrès International des Mathématiciens (Nice, 1970), Tome 3, Gauthier-Villars, Paris, 1971, pp. 229–233. MR 0505646 (58 #21703)
[Rot97]	, <i>Indiscrete thoughts</i> , Birkhäuser Boston, Inc., Boston, MA, 1997, With forewords by Reuben Hersh and Robert Sokolowski, Edited with notes and an epilogue by Fabrizio Palombi. MR 1419503 (97i:01026)
[Sag99]	Bruce E. Sagan, Why the characteristic polynomial factors, Bull. Amer. Math. Soc. (N.S.) <b>36</b> (1999), no. 2, 113–133. MR 1659875 (2000a:06021)
[Sag01]	, <i>The symmetric group</i> , second ed., Graduate Texts in Mathematics, vol. 203, Springer-Verlag, New York, 2001, Representations, combinatorial algorithms, and symmetric functions. MR 1824028 (2001m:05261)
[Sai80]	Kyoji Saito, Theory of logarithmic differential forms and logarithmic vector fields, J. Fac. Sci. Univ. Tokyo Sect. IA Math. <b>27</b> (1980), no. 2, 265–291. MR 586450 (83h:32023)
[San05]	Francisco Santos, <i>The Cayley trick and triangulations of products of simplices</i> , Integer points in polyhedra—geometry, number theory, algebra, optimization, Contemp. Math., vol. 374, Amer. Math. Soc., Providence, RI, 2005, pp. 151–177. MR 2134766 (2005k:52025)
[Sch10]	Jay Schweig, On the h-vector of a lattice path matroid, Electron. J. Combin. 17 (2010), no. 1, Note 3, 6. MR 2578897 (2010k:05053)
[Sok05]	Alan D. Sokal, The multivariate Tutte polynomial (alias Potts model) for graphs and matroids, Surveys in combinatorics 2005, London Math. Soc. Lecture Note Ser., vol. 327, Cambridge Univ. Press, Cambridge, 2005, pp. 173–226. MR 2187739 (2006k:05052)
[Sok12]	$-\!\!\!\!$ , The leading root of the partial theta function, Adv. Math. <b>229</b> (2012), no. 5, 2603–2621. MR 2889137
[Spe08]	David E. Speyer, <i>Tropical linear spaces</i> , SIAM J. Discrete Math. <b>22</b> (2008), no. 4, 1527–1558. MR 2448909 (2009m:52023)
[Spe09]	, A matroid invariant via the K-theory of the Grassmannian, Adv. Math. <b>221</b> (2009), no. 3, 882–913. MR 2511042 (2010i:14114)
[SS12]	Luis Serrano and Christian Stump, Maximal fillings of moon polyominoes, simplicial complexes, and Schubert polynomials, Electron. J. Combin. <b>19</b> (2012), no. 1, Paper 16, 18. MR 2880647
[Sta]	Richard P. Stanley, Catalan addendum.
[Sta72]	R. P. Stanley, Supersolvable lattices, Algebra Universalis 2 (1972), 197–217. MR 0309815 (46 #8920)
[Sta77]	Richard P. Stanley, <i>Cohen-Macaulay complexes</i> , Higher combinatorics (Proc. NATO Advanced Study Inst., Berlin, 1976), Reidel, Dordrecht, 1977, pp. 51–62. NATO Adv. Study Inst. Ser., Ser. C: Math. and Phys. Sci., 31. MR 0572989 (58 #28010)
[Sta80a]	R. P. Stanley, <i>Differentiably finite power series</i> , European J. Combin. <b>1</b> (1980), no. 2, 175–188. MR 587530 (81m:05012)
[Sta80b]	Richard P. Stanley, <i>Decompositions of rational convex polytopes</i> , Ann. Discrete Math. <b>6</b> (1980), 333–342.
[Sta80c]	Richard P. Stanley, <i>The number of faces of a simplicial convex polytope</i> , Adv. in Math. <b>35</b> (1980), no. 3, 236–238. MR 563925 (81f:52014)

[Sta86]	, Two poset polytopes, Discrete Comput. Geom. 1 (1986), no. 1, 9–23. MR 824105 (87e:52012)
[Sta87]	Richard Stanley, Generalized H-vectors, intersection cohomology of toric varieties, and related results, Commutative algebra and combinatorics (Kyoto, 1985), Adv. Stud. Pure Math., vol. 11, North-Holland, Amsterdam, 1987, pp. 187–213. MR 951205 (89f:52016)
[Sta91]	Richard P. Stanley, A zonotope associated with graphical degree sequences, Applied geometry and discrete mathematics, DIMACS Ser. Discrete Math. Theoret. Comput. Sci., vol. 4, Amer. Math. Soc., Providence, RI, 1991, pp. 555–570. MR 1116376 (92k:52020)
[Sta94]	, Flag f-vectors and the cd-index, Math. Z. 216 (1994), no. 3, 483–499. MR 1283084 (96b:06006)
[Sta99]	, <i>Enumerative combinatorics. Vol. 2</i> , Cambridge Studies in Advanced Mathematics, vol. 62, Cambridge University Press, Cambridge, 1999, With a foreword by Gian-Carlo Rota and appendix 1 by Sergey Fomin. MR 1676282 (2000k:05026)
[Sta07]	, An introduction to hyperplane arrangements, Geometric combinatorics, IAS/Park City Math. Ser., vol. 13, Amer. Math. Soc., Providence, RI, 2007, pp. 389–496. MR 2383131
[Sta12]	, <i>Enumerative combinatorics. Volume 1</i> , second ed., Cambridge Studies in Advanced Mathe- matics, vol. 49, Cambridge University Press, Cambridge, 2012. MR 2868112
[Ste07]	John R. Stembridge, A short derivation of the Möbius function for the Bruhat order, J. Algebraic Combin. 25 (2007), no. 2, 141–148. MR 2310418 (2008b:05185)
[Stu96]	Bernd Sturmfels, <i>Gröbner bases and convex polytopes</i> , University Lecture Series, vol. 8, American Mathematical Society, Providence, RI, 1996. MR 1363949 (97b:13034)
[Tak90]	Lajos Takács, On the number of distinct forests, SIAM J. Discrete Math. <b>3</b> (1990), no. 4, 574–581. MR 1069116 (91i:05046)
[Ter80]	Hiroaki Terao, Arrangements of hyperplanes and their freeness. I, J. Fac. Sci. Univ. Tokyo Sect. IA Math. 27 (1980), no. 2, 293–312. MR 586451 (84i:32016a)
[Ter02]	, Algebras generated by reciprocals of linear forms, J. Algebra <b>250</b> (2002), no. 2, 549–558. MR 1899865 (2003c:16052)
[TF61]	H. N. V. Temperley and Michael E. Fisher, <i>Dimer problem in statistical mechanics—an exact result</i> , Philos. Mag. (8) <b>6</b> (1961), 1061–1063. MR 0136398 (24 #B2436)
[Tut54]	W. T. Tutte, A contribution to the theory of chromatic polynomials, Canadian J. Math. 6 (1954), 80–91. MR 0061366 (15,814c)
[Tut67]	, On dichromatic polynomials, J. Combinatorial Theory <b>2</b> (1967), 301–320. MR 0223272 (36 $\#6320$ )
[Tut91]	, On the Birkhoff-Lewis equations, Discrete Math. <b>92</b> (1991), no. 1-3, 417–425. MR 1140602 (92k:05052)
[vdW37]	B. L. van der Waerden, Zur algebraischen Geometrie, Math. Ann. <b>113</b> (1937), no. 1, 36–39. MR 1513077
[Ver71]	Daya-Nand Verma, Möbius inversion for the Bruhat ordering on a Weyl group, Ann. Sci. École Norm. Sup. (4) 4 (1971), 393–398. MR 0291045 (45 #139)
[Wac07]	Michelle L. Wachs, <i>Poset topology: tools and applications</i> , Geometric combinatorics, IAS/Park City Math. Ser., vol. 13, Amer. Math. Soc., Providence, RI, 2007, pp. 497–615. MR 2383132
[Wag99]	David G. Wagner, Algebras related to matroids represented in characteristic zero, European J. Combin. <b>20</b> (1999), no. 7, 701–711. MR 1721927 (2001j:13002)
[Wal48]	H. S. Wall, Analytic Theory of Continued Fractions, D. Van Nostrand Company, Inc., New York, N. Y., 1948. MR 0025596 (10,32d)
[Wei35]	Louis Weisner, Some properties of prime-power groups, Trans. Amer. Math. Soc. <b>38</b> (1935), no. 3, 485–492. MR 1501823
[Wel76]	D. J. A. Welsh, <i>Matroid theory</i> , Academic Press [Harcourt Brace Jovanovich, Publishers], London-New York, 1976, L. M. S. Monographs, No. 8. MR 0427112 (55 #148)
[Wel93]	Complexity: knots colourings and counting London Mathematical Society Lecture Note Se-

[Wel93] \_\_\_\_\_, Complexity: knots, colourings and counting, London Mathematical Society Lecture Note Series, vol. 186, Cambridge University Press, Cambridge, 1993. MR 1245272 (94m:57027)

[Whi35] Hassler Whitney, On the Abstract Properties of Linear Dependence, Amer. J. Math. 57 (1935), no. 3, 509-533. MR 1507091 Neil White (ed.), Theory of matroids, Encyclopedia of Mathematics and its Applications, vol. 26, [Whi86] Cambridge University Press, Cambridge, 1986. MR 849389 (87k:05054) [Whi87] Neil White (ed.), Combinatorial geometries, Encyclopedia of Mathematics and its Applications, vol. 29, Cambridge University Press, Cambridge, 1987. MR 921064 (88g:05048) [Whi92] Neil White (ed.), Matroid applications, Encyclopedia of Mathematics and its Applications, vol. 40, Cambridge University Press, Cambridge, 1992. MR 1165537 (92m:05004) [Wil05] Lauren K. Williams, Enumeration of totally positive Grassmann cells, Adv. Math. 190 (2005), no. 2, 319–342. MR 2102660 (2005i:05197) [Wil06] Herbert S. Wilf, generatingfunctionology, third ed., A K Peters, Ltd., Wellesley, MA, 2006. MR 2172781 (2006i:05014)[WW86] Michelle L. Wachs and James W. Walker, On geometric semilattices, Order 2 (1986), no. 4, 367–385. MR 838021 (87f:06004) [WW99] Dominic J. A. Welsh and Geoffrey P. Whittle, Arrangements, channel assignments, and associated polynomials, Adv. in Appl. Math. 23 (1999), no. 4, 375–406. MR 1733134 (2000m:05055) [Zas75]Thomas Zaslavsky, Facing up to arrangements: face-count formulas for partitions of space by hyperplanes, Mem. Amer. Math. Soc. 1 (1975), no. issue 1, 154, vii+102. MR 0357135 (50 #9603) Doron Zeilberger, The amazing  $3^n$  theorem and its even more amazing proof. [Zei] [Zei96] , Proof of the refined alternating sign matrix conjecture, New York J. Math. 2 (1996), 59–68, electronic. MR 1383799 (97c:05010) [Zei99] , Proof of a conjecture of Chan, Robbins, and Yuen, Electron. Trans. Numer. Anal. 9 (1999), 147–148 (electronic), Orthogonal polynomials: numerical and symbolic algorithms (Leganés, 1998). MR 1749805 (2001b:05015) [Zha13] Ilia Zharkov, The Orlik-Solomon algebra and the Bergman fan of a matroid, J. Gökova Geom. Topol. GGT 7 (2013), 25-31. MR 3153919 [Zie95] Günter M. Ziegler, Lectures on polytopes, Graduate Texts in Mathematics, vol. 152, Springer-Verlag, New York, 1995. MR 1311028 (96a:52011) [Zie02] , Face numbers of 4-polytopes and 3-spheres, Proceedings of the International Congress of Mathematicians (Beijing, 2002), vol. III, Higher Ed. Press, Beijing, 2002, pp. 625–634. MR 1957565 (2004a:52019)